

Leveraging shared mobility data for urban development and policymaking: A study of Dutch municipalities' data supply, demand, and challenges & opportunities

Master's Thesis - master Sustainable Business and Innovation

Author: F.J.H. (Floris) van den Dool, BSc

Student number: 1339605

Contact: f.j.h.vandendool@students.uu.nl

Supervisor 1: I. Susha, dr

Supervisor 2: B. Pel, dr

Institution and department: Utrecht University, Geosciences

Program: Sustainable Business & Innovation

Date: 11-7-2024

Wordcount: 29.963



ABSTRACT

This thesis examines the demand and supply of shared mobility data for urban development and policy-making in Dutch municipalities, additionally focusing on challenges and opportunities in acquiring and using data. It examines how shared mobility providers position themselves in the context of data sharing and how their data is demanded by municipalities for evidence-based decision-making to address policy issues, such as urban densification and sustainability and shared mobility nuisances, working towards the development of smart and sustainable cities.

Using a modified version of Susha et al.'s (2017) taxonomy of data collaborations, which focuses on characteristics of data supply and demand and Diran & van Veenstra's (2020) barriers to data collection and use, this study investigates the research problem centered around the mismatch between the data supply of shared mobility providers and the demand of Dutch municipalities. Utilizing an interpretive research philosophy, this thesis follows an abductive qualitative approach with a multiple case study design, focusing on four major Dutch municipalities: Amsterdam, Rotterdam, The Hague, and Eindhoven. The data was collected based on semi-structured interviews with municipalities, researchers/experts, and shared mobility providers, supplemented with secondary data from policy documents and a literature review. Through thematic analysis, unique insights are discovered regarding shared mobility data sharing. It becomes clear that despite all the benefits that arise when shared mobility data is shared, this data sharing is hindered by challenges such as privacy, legal restrictions, and resistance from private companies.

The findings indicate municipal demand for detailed and preferably dynamic data that includes trip start and end locations, travel times, and usage patterns to improve urban development and policymaking and reduce reliance on private cars. Whereas providers of shared two-wheelers meet this data demand by making it mandatory in the operating permit, providers of shared cars offer more resistance. This research highlights the importance of a clear and thorough data demand in which transparency in data processing and use follow directly from a policy problem/use case.

Furthermore, the thesis emphasizes overcoming challenges such as silo thinking and lack of knowledge and capabilities of Dutch municipalities. A promising initiative that can both facilitate municipalities on their flaws and also facilitate shared mobility providers in providing transparency and a clear goal of municipal data demand is the CDS-M. Further research can delve into this and explore what makes a well-supported data demand.

 Keywords: Shared Mobility, Data Sharing, Urban Development, Policymaking, Data Supply & Demand, Dutch Municipalities, Smart & Sustainable Cities, New Mobility Services (NMS), Data-driven Decision Making, Business-to-Government (B2G) Data Sharing, Public-Private Partnership, Innovation, Collaboration



SUMMARY

European cities classified as smart (should) use data in decision-making and urban planning for municipal problem-solving. Nonetheless, data supply and demand often do not align and thus complications may arise in the decision-making process. Therefore, this research aims to understand how Dutch municipalities leverage shared mobility data to inform urban development and policymaking. Specifically, it seeks to explore the supply and demand for shared mobility data within Dutch smart cities, and to identify the challenges and opportunities municipalities face in accessing and utilizing this data, posing the following research question: "What is the supply and demand for shared mobility data within Dutch municipalities to inform (smart) urban development and policymaking, and which access and utilization challenges and opportunities arise in the data sharing process?"

To do so, this research uses an interpretivist philosophy to understand the subjective perspective and experiences of municipalities, researchers/experts, and shared mobility providers. Through an exploratory multiple case study design, this research is enabled to gather in-depth information from the stakeholders involved. Through primary data acquired through semi-structured interviews with 4 different municipalities, 5 different researchers/experts, and 4 industry representatives, this study provides a comprehensive understanding of shared mobility data sharing and its use within municipalities for urban planning and policymaking. Based on a combination of criterion and purposive sampling, 4 relevant Dutch municipalities were selected. The municipalities of Amsterdam, Rotterdam, Eindhoven, and The Hague all qualified having more than 150,000 residents which has a direct relationship with the number of shared mobility providers within the city and the degree of data usage. The acquired data was subjected to thematic coding analysis, which identified predetermined but also new themes. The pre-determined themes were extracted from the modified taxonomy of Susha et al., (2017), which was constructed to represent the supply and demand of data within collaborations. Furthermore, the challenges identified by Diran & van Veenstra, (2020) are taken as a basis and further tested for urban development challenges in the data regime, in this case, shared mobility.

Municipalities see the benefits of shared mobility for addressing urban issues like densification and urbanization, as they aim to reduce private vehicle ownership to free up public space and achieve sustainability goals. However, they are also afraid of 'a second Uber' which might disrupt the status quo in the mobility regime within Duch municipalities, asking for regulation. For this, municipalities want to obtain shared mobility data to solve short-term policy problems, but shared mobility providers indicate that municipalities should not do this and should look at the longer term, which directly corresponds to more concrete policy problems and use cases for data. Shared mobility providers, especially car-sharing providers, perceive the data requests of municipalities often as unfounded, asking for excessive data with suboptimal purposes, leading to reluctance from providers to share data. CDS-M and its corresponding program of Natuurlijk Deelmobiliteit are still developing in collaboration with shared mobility providers and municipalities, aiming to address these issues and provide clarity regarding data demand and supply, as well as the methods of data sharing. Essentially this program aims to tackle most of the challenges identified in this research, specifically focusing on matching shared mobility data supply and demand through the development of standardized data sharing and use cases.

Shared mobility data sharing can be done in three ways: mandatory, voluntary, and a combination of both. Two-wheel providers are mandated for data sharing as it is part of their permit to operate within the city. Car-sharing providers, however, are essentially only sharing data voluntarily as they do create



far fewer nuisances and disruption compared to two-wheelers so there is no immediate reason for municipalities to request data. Therefore dynamic data is often shared by two-wheel providers and more static data is shared by car-sharing companies. However, it should be noted that this data sharing is only done after concrete municipal data demand, with clear use cases for the data. Nevertheless, municipalities are increasingly pushing towards mandatory data sharing for all mobility providers. One of the interviewed car-sharing providers even opted for a mandatory data-sharing basis, where additional data can be shared with municipalities voluntarily. Also, some shared mobility providers proactively share data as they see benefits in improving services and infrastructure, enabling mutual benefits. An alternative to data sharing has also arisen: engaging in dialogue with shared mobility providers and municipalities enables mutual learning, meaning that both parties can benefit from collaboration and communication.

However, the data supply is compromised by several factors. Privacy sensitivity remains at the core of concerns in data sharing. Data should not be retractable to individuals or impede the GDPR, therefore data is anonymized and aggregated as much as possible. Additionally, data is valuable and considered intellectual property that requires protection to maintain a competitive advantage. Mobility providers are cautious about sharing data, to avoid compromising their business interests. Also, there is a lack of coherence and centralization in data sharing, which indicates an immediate demand for centralization and handling data sharing ethically. Additionally, the use of shared mobility data is also restricted. Especially, the barriers such as lack of expertise and skills, determining the value and purpose of data, legal limitation/ GDPR, and difficulties in linking, analyzing, and visualizing data are crucial for Dutch municipalities. Although municipalities seem to know the value of data, they lack the expertise and skills to deal with them and often do not address specific purposes of data use. Municipalities are too siloed in their way of thinking and operations. As municipalities often lack the knowledge and technical capabilities to access and analyze data, they often rely on third-party platforms, such as Vianova and CROW. These platforms help integrate and visualize data, though challenges remain, especially with car-sharing data. However dependence on these parties may compromise internal knowledge accumulation.

Ideally, municipalities want to receive as much and as direct data as possible from each mobility provider. Data supply and demand and the various aspects therein are case-specific and therefore difficult to generalize. However, shared two-wheel providers are substantially supplying more data compared to car-sharing providers. But before municipalities can effectively use data at all they should come to clear and transparent data demands and processing and overcome multiple internal challenges.



TABLE OF CONTENTS

Abstract					
Summary					
List of Fig	gures7				
List of tab	les				
List of abl	previations7				
List of det	Finitions				
1. Intro	duction				
1.1.	Problem statement and research questions				
1.2.	Aim of research				
1.3.	Research relevance				
2. back	ground				
2.1.	Climate-neutral and smart cities in Europe				
2.2.	Status quo and resistance to change				
2.3.	Urban technological innovation				
2.4.	Collaboration				
2.5.	Policy making				
2.6.	New Mobility Services (NMS)				
2.7.	Digitalization				
3. Theo	retical framework				
4. Rese	Research methodology				
4.1.	Research philosophy				
4.2.	Research type & design				
4.3.	Sampling strategy				
4.4.	Data collection				
4.5.	Data analysis				
4.6.	Operationalization				
4.7.	Research quality indicators				
4.8.	Data ethics and privacy				
5. Results					
5.1.	Dutch municipal data demand				
5.2.	.2. Supply of shared mobility data (and its challenges)				
5.3.	5.3. Digitalization and data sharing developments and opportunities in the Netherlands				



	5.4. Challenges and opportunities in utilization of and leveraging data in urban planning and			
	policy	making		
	5.5.	Municipality and shared mobility provider comparison	53	
6.	Dis	cussion	56	
	6.1.	Empirical findings and implications	56	
	6.2.	Limitations	60	
	6.3.	Contributions and future directions	61	
7.	Cor	nclusion	63	
8.	Rec	commendations for practitioners	66	
9.	Ack	cnowledgements	67	
10	. 1	Bibliography	68	
11	. /	Appendix A – (Data and digitalization strategy)	80	
12	. /	Appendix B – (Law and regulations in data-driven decision-making)	83	
13	. /	Appendix C – (Maas findings)	85	
14	. /	Appendix D – (Initial literature review)	88	
15	• 1	Appendix E – (Cross-case analysis foundation)		
16	. /	Appendix F – (Consent form and information)		
17	. /	Appendix G – (Description ethical issues)		
18	. /	Appendix H – (Interview guide municipalities)		
19	. /	Appendix I – (Interview guide researchers/Experts)	109	
20	. /	Appendix J – (Interview guide industry representatives)		



LIST OF FIGURES

Figure 1: Overview of various New Mobiliy Services (TNO, 2020).

Figure 2: Dimensions of the taxonomy of data collaboratives based on the supply-demand relationship (Susha et al., 2017).

Figure 3: Research model (including data collection, analyses, theoretical framework, and answering research questions)

Figure 4: Display of API function (Gemeente Amsterdam, 2021a)

Figure 5: Levels of MaaS (TNO, 2020)

LIST OF TABLES

Table 1: Barriers and challenges in data collection and data use by policymakers

Table 2: Inteview respondents

Table 3: Initial codebook (categories derived from theoretical framework

Table 4: Comparison matrix

Table 5: MaaS challenges

 Table 6: MaaS use purposes

Table 7: Search keywords and results

LIST OF ABBREVIATIONS

API	Application Programming Interface
AVG	Algemene Verordening Gegevensbescherming
B2G	Business to Government
CBS	Centraal Bureau voor de Statistiek
CDS-M	City Data Specificatie Mobiliteit
CG	Common Ground
CoE-DSC	Centre of Excellence in Data Sharing and Collaboration
CROW	Centrum voor Regelgeving en Onderzoek in de Grond-,
	Water- en Wegenbouw en Verkeerstechniek
ER	Expert Respondent
ERTRAC	European Road Transport Research Advisory Council
EU	European Uninion
GBFS	General Bikeshare Feed Specification
GDPR	General Data Protection Regulation
GIS	Geographic Information System
GPS	Global Positioning System
IBDS	Inter Bestuurlijke Data Strategie
ICT	Information and Communication Technology
IR	Industry Respondent
ISO	International Organization for Standardization
I&W	Infrastructuur en Waterstaat
MaaS	Mobility as a Service
MDS	Mobility Data Specification
MPC	Multi-party Computation
MR	Municipal Respondent
NAP	National Access Point
NMS	New Mobility Service



NS	Nederlandse Spoorwegen
NTM	Nationaal Toegangspunt Mobiliteitsdata
OMF	Open Mobility Foundation
OUP	Open Urban Platform
SDG	Sustainable Development Goal
STOMP	Stappen, Trappen, Openbaar vervoer, Mobility service, Prive
	vervoer
TNO	Nederlandse Organisatie voor Toegepast
	Natuurwetenschappelijk Onderzoek
TOMP-API	Transport Operator to Mobility/MaaS Provider Application
	Programming Interface
UITP	International Association of Public Transport
UN	United Nations
UNECE	United Nations Economic Commission for Europe
UNEP	United Nations Environment Programme
US	United States
VNG	Vereniging van Nederlandse Gemeenten

LIST OF DEFINITIONS

Trip data: information about origins, destinations, travel times, driver behavior, and other statistics, while securing user privacy (Bezerra et al., 2019; Aryandoust et al., 2019; Roy et al., 2020; Ludwig et al., 2023; Neun et al., 2023; Cohen, 2018; Hawken et al., 2020).

NMS: It is recognized that it refers to new actors, new types of ownership and business models, new vehicle types, and new uses of public space (ERTRAC, 2021). Large group access and shared vehicles form the essence according to research from TNO, (2020).

Platform companies: "Platform companies are marketplaces" – Martijn Arets (ER3)

Shared mobility providers: *"Shared mobility providers are rental companies with a completely different approach, often with a technology layer"* – Martijn Arets (ER3)

Smart and sustainable city: An innovative city that uses ICT to improve quality of life, efficiency, and competitiveness while meeting the needs of current and future generations in economic, social, environmental, and cultural aspects (UNECE, 2023). In addition, the UN Sustainable Development Goal (SDG) 11, which is widely used across European cities, aligns with smart and sustainable cities, focusing on inclusive and sustainable urbanization, safe and affordable transportation, reducing environmental impacts, and improving disaster resilience (UNEP, 2017).

MaaS: Currently, there is no uniform definition of MaaS. However, it is generally seen as: *"the integration of various forms of transport services into a single mobility service accessible on demand"* (MaaS Alliance, 2017).

Data-driven work: Data-driven work is the practice of operating based on data that arises from actual events in society. This data is analyzed and processed into information, which, when combined with domain knowledge, can be interpreted correctly. This can lead to better-informed decisions with greater value for society (Nijman & Alberts-de Gier, 2023).



1. INTRODUCTION

Between 2014 and 2050 the global population growth is expected to be 32%, with urban environments experiencing a 63% increase (Estevez et al., 2016). As of April 2023, 56% percent of the world's population live in cities. These 4.4 billion people living in cities are expected to double and reach a point where nearly 70% of the world's population lives in cities by 2050 (The World Bank, 2023). This growth is closely linked to globalization (Al-Rodhan & Gérard Stoudmann Director, 2006), which impacts urban planning and development (Dunarintu & Dociu, 2012). The increased population in cities leads to increased commuting and traffic (Grace Dobush, 2019), resulting in social and environmental problems such as increased greenhouse gas emissions, traffic fatalities, transportation and fuel consumption, and air and noise pollution (Kalhor & Mahdisoltani, 2015). For urban development, it is therefore important to have direct insights into the changing behavior of civil society (Friedmann, 2005).

The increasing densification of cities has led to a growing demand for alternative transportation options, such as shared mobility and on-demand services (Butler et al., 2020). The rapid upsurge of platform companies offering products and services that can be shared leads to higher car-driving efficiency and therefore fewer cars in public spaces (Maselli et al., 2016). Additionally, the market of New Mobility Services opens up and is seen as the way forward to counteract urban (development) problems due to increased vehicle use and mobility efficiency (Kamargianni et al., 2016a). Municipalities show that they need a different perspective on mobility to keep cities accessible livable and traffic-safe (Gemeente Den Haag, 2021), but also to meet climate requirements (Gemeente Den Haag, 2020b). A direct consequence of this modal shift takes place in terms of urban design (Hawken et al., 2020).

To address the changing transport and transportation relationships in cities, a long-term vision in urban planning and development is crucial (Bibri & Krogstie, 2017). Smart, sustainable urban planning addresses environmental and social issues and aims for sustainable development (Bibri & Krogstie, 2017). Creating smart and sustainable cities requires a data-informed decision-making approach (Hawken et al., 2020; Paskaleva et al., 2021). As indicated by the United Nations Economic Commission for Europe (UNECE), the use of Information and Communication Technologies (ICTs) is essential in urban innovation and developing smart and sustainable cities (UNECE, 2023). ICTs provide a comprehensive way to collect, analyze, and integrate urban data, such as human mobility, spatiotemporal, traffic flow, environmental, energy, transport, and socio-economic data, into development plans. Europe leads the way in smart city initiatives (Estevez et al., 2016) and Hawken et al. (2020) and Geropanta et al. (2021) acknowledge that European cities are using various technologies and data to develop smart and sustainable cities. However, data demand by urban planners and policymakers is not met with supply (Martin et al., 2018; S. G. Verhulst et al., 2016).

Although local governments are increasingly using trip data (Cohen, 2018), commitment to this method requires data-sharing partnerships (Hawken et al., 2020). However, challenges like data privacy, misuse of data, and different interests arise. Strong governance structures and regulations are needed for effective and efficient data sharing (Benli-Trichet & Kübler, 2022; Zhang, 2019). In this, balancing interests and frameworks to ensure data security, privacy, and usage is essential (Susha et al., 2017; Ruijer, 2021). The central problem, however, is that this (big) data is often owned by private companies that want to keep the data private for innovation and differentiation. This situation is evolving as more externally gathered data opens up the potential for positive societal solutions (Verhulst, 2021). For example, a specific group of private companies collect huge amounts of trip data such as trip origins, destinations, and travel time from its users (Hawken et al., 2020; Ludwig et al.,



2023). Sharing this data with governments can be helpful for urban planning and policymaking. These private companies can provide valuable information on traffic flow and energy usage, enabling initiatives to reduce greenhouse gas emissions and improve energy efficiency (Hudović Kljuno & Krivošić Dizdarević, 2021). This data can also help municipalities make informed and evidence-based decisions for infrastructure and transportation systems development (Mills et al., 2022; Cohen, 2018; Hawken et al., 2020). Although data for public good is still in its early stages (Verhulst, 2021), this data can help urban planners and policymakers incentivize smart and sustainable transportation (Grace Dobush, 2019).

1.1. PROBLEM STATEMENT AND RESEARCH QUESTIONS

Based on the literature, European smart cities (should) use data in decision-making and urban planning (Hawken et al., 2020; Paskaleva et al., 2021). Nonetheless, the literature also shows that data supply and demand often do not align and thus complications may arise in the decision-making process (Verhulst, 2021). Therefore, this research aims to understand how Dutch municipalities (will) leverage shared mobility data to inform urban development and policymaking. Specifically, it seeks to explore the supply and demand for shared mobility data within Dutch smart cities, and to identify the challenges and opportunities municipalities face in accessing and utilizing this data, posing the following research questions:

"What is the supply and demand for shared mobility data within Dutch municipalities to inform (smart) urban development and policymaking, and which access and utilization challenges and opportunities arise in the data sharing process?"

Sub-questions:

- 1. "How is the Dutch municipalities' shared mobility data demand formed and what data do they need for urban planning and policymaking purposes?"
- 2. "What is the current state of data-sharing initiatives between shared mobility providers and Dutch municipalities, and how is this data sourced?"
- 3. "What are the technical and organizational challenges and opportunities experienced by Dutch municipalities in accessing and using data from shared mobility providers active in the Netherlands?"

1.2. AIM OF RESEARCH

The central goal of this research is to investigate the supply and demand of private shared mobility (trip) data within Dutch municipalities. This research aims to understand the supply, demand, and use of such data by Dutch municipalities and how this might affect strategies and policies for developing smarter and more sustainable urban systems in response to urban innovations and changing urban dynamics.

To this end, the three sub-questions are constructed to work toward answering the main question. The first sub-question focuses on Dutch municipal data needs and how these are expressed. This contributes to the understanding of the motivations and objectives in using such data for urban planning and policymaking. The second sub-question explores what data is available to Dutch municipalities and how this data is offered. This opens up the data supply for municipalities and knowing what data they could use in urban planning and policymaking. The third sub-question examines the challenge in practical utilization and acquirement of data within Dutch municipalities to



understand the complications that arise in data sharing as well as urban planning and policy making (based on data).

1.3. RESEARCH RELEVANCE

1.3.1. SCIENTIFIC RELEVANCE

This research addresses the gap between the supply and demand of shared mobility (trip) data in urban planning and policymaking processes within Dutch cities, by investigating the supply and demand of shared mobility data. By providing empirical insights into how Dutch municipalities form their shared mobility data demand and aim to leverage this data, this study contributes to the understanding of data-driven solutions for urban development and policymaking. Specifically, this research provides further focus on the development of smart and sustainable cities and urban mobility, such as research by (Bibri & Krogstie, (2017) and Hawken et al., (2020), which demonstrates the importance of data in urban development and policymaking.

In addition, this research helps identify developments in the field of shared mobility data sharing facilitation and the need for it. By investigating this, it shines light on potential developments and possible mutual accommodations and thus successful data collaborations. This contribution is important for the Sustainable Business & Innovation field, as it provides useful knowledge on how municipalities want to benefit from integrating shared mobility data into their strategic frameworks, which can be inspiring for other municipalities. Moreover, this research contributes to the understanding of how the supply and demand of shared mobility data can be brought closer together.

This research also builds on the theoretical supply and demand underpinnings of Susha et al., (2017), as well as the distinctions between voluntary, compulsory, and intermediate data sharing according to the theories of (Rukanova et al., 2020; Susha, Rukanova, et al., 2019; Klievink, Van Der Voort, et al., 2018; Vigorito, 2022)). Through practical discoveries within shared mobility data sharing, this research demonstrates whether and how these theories are practically relevant within the context of urban development and policymaking. Additionally, this research tests the generalizability of the data acquisition and utilization challenges described by Diran & van Veenstra, (2020) within the urban development and policymaking context.

1.3.2. SOCIETAL RELEVANCE

This research contributes to improving urban development in Dutch municipalities by discovering how shared mobility data can be used in decision-making processes. The insights gained from this study inform urban planners and policymakers about the potential of shared mobility data to facilitate integrated collaboration and evidence-based decision-making. Additionally, this research identifies practical challenges and opportunities in shared mobility data acquisition and utilization by Dutch municipalities, offering lessons learned from large Dutch municipalities.

Shared mobility itself is a form of smart and sustainable mobility, which directly positively influences (urban) climates. By shared mobility providers sharing data, this mobility sector can be improved (due to infrastructural developments or direct stimulation) which increasingly has positive societal effects.

Shared mobility providers, however, can also benefit from the identification of municipal data demand as they could (pro)actively engage with municipalities and share data to increase and improve infrastructures and mobility regulations which enables them to further grow and operate within the built environment.



By specifically focusing on Dutch municipalities this research underscores the importance of shared mobility in the Netherlands and allows findings to be directly relevant and applicable to the Dutch context. Dutch municipalities could use this study to objectively study the supply and demand of shared mobility data and the needed developments to further facilitate and integrate (data sharing) collaboration in urban development and policymaking (in the field of shared mobility).

As this research is highly explorative this research allows researchers and students to further investigate the area of shared mobility and its data sharing specifically in the Netherlands or other countries. This research identifies a large number of developments and challenges which ask for further research.

THESIS OUTLINE

Chapter 2 conducts an extensive state-of-the-art review to establish the current state of knowledge on shared mobility data sharing and urban development and policymaking in the Dutch municipal context. Chapter 3 presents the theoretical framework guiding the study, integrating key theories and concepts to provide a conceptual lens for analysis and for understanding the supply and demand dynamics of shared mobility data. Chapter 4 outlines the research design and explains the rationale behind the chosen methodology and how it aligns with the research objectives. Chapter 5 presents the results derived from the analysis of case studies and other data, offering a detailed presentation and interpretation of empirical findings in relation to the research questions and theoretical framework. This chapter provides insights into the shared mobility data supply, demand, acquisition, and utilization challenges faced by Dutch municipalities. Chapter 6 critically examines the findings, by providing a comprehensive discussion that highlights consistencies and deviations with existing literature. It explores the implications of the findings, addresses the study's limitations, and provides recommendations for future research. Chapter 7 concludes the study by answering the research questions. It reflects on the empirical findings and underscores the current state of shared mobility data sharing among Dutch municipalities and its significance in urban development. Chapter 8 presents a short and concrete overview of recommendations for practitioners, including actionable steps to innovate.



2. BACKGROUND

This chapter delves into the state-of-the-art knowledge necessary to understand how Dutch municipalities are currently positioned toward (smart) urban development and policymaking, digitalization, innovation, data, and data usage. It first looks at the development and status quo (change resistance) in Dutch mobility systems. Then the developments and vision of collaborations are addressed to identify the willingness and recognition of (external) expertise and capabilities. The focus of this chapter is on the development of NMS and its influence on Dutch urban development. After which MaaS and digitalization within urban development are also briefly discussed.

2.1. CLIMATE-NEUTRAL AND SMART CITIES IN EUROPE

The European Commission aims to develop 100 climate-neutral and smart cities by 2030, with Amsterdam, Eindhoven & Helmond, Groningen, Rotterdam, The Hague, and Utrecht selected as development cities in the Netherlands. The Climate City Contracts outline the cities' commitments, involving residents, research institutions, and the private sector (European Commission, 2022). The Dutch Ministry of Infrastructure and Water Management's (I&W) Growth Fund focuses on building a digital infrastructure for data sharing and utilization, facilitating data acquisition, and improving the quantity (Van Schijndel-de Nooij, 2022). The European Green Deal aims to move towards a sustainable future through digitization, with funding provided for digitization projects in urban environments through the Digital Europe program (European Commission, 2023a). Municipalities are translating these goals into concrete mobility master plans, developing strategies to make shared transport a serious alternative to car ownership (Gemeente Eindhoven, 2023).

2.2. STATUS QUO AND RESISTANCE TO CHANGE

Urban development often overemphasizes cars, a dominant non-human entity that shapes people's spatial environment (MR5). However, car' externalities are often overlooked in urban development. Reconsidering this could lead to radically different mobility system principles. To address negative externalities like emissions and traffic congestion, the government is seeking a fully-fledged alternative that seamlessly replaces cars in urban development (ER3). However, changing this approach and mode of transport is considered challenging. Many people still choose to drive, even though other means of transportation might be faster for getting from point A to point B (MR2). Additionally, Jeanette van Eijk from Greenwheels (IR3) is convinced that car usage is specifically high in the Netherlands, saying: *"I believe there is no country in Europe where so many people own so many cars per capita as in the Netherlands. It's quite a sacred cow."*

Municipalities aim to actively address this issue to break the residents' habits and routines, as they often cannot do it themselves (MR2; MR6). To support this, municipalities want to work according to the STOMP (Walking, Biking, Public transport, Mobility service, Private transport) principle. This principle prioritizes modes of transport in a way that minimizes private car use within a city. It starts with walking, followed by cycling, public transport, mobility services, and lastly, private cars (ER1). Municipal respondents suggest adjusting policies and regulations to eliminate the current focus on private cars to successfully reverse the mobility pyramid (MR2; MR6).

2.2.1. RESISTANCE TO CHANGE

In practice, it is difficult to initiate change in the field of urban mobility. Municipalities observe that many people continue to use their cars within the city despite the compact urban design. Consequently, even with the availability of trams and various transport options that might be even



faster than car transport, a significant number of residents prefer driving themselves (MR2). This situation has prompted a focus on encouraging residents to consider faster and more efficient alternatives such as bicycles or other modes of transport, as this leads to less individual dependence on private vehicle use (Gemeente Den Haag, 2020b). The challenge lies in changing established habits and making residents aware of these viable options. In practice, however, shared mobility is primarily considered appealing to those who do not own a car yet or are looking to replace a second car (Gemeente Den Haag, 2020b). This habit is also influenced by governance and politics (MR6; ER2). When a conservative party favoring car use is in power, the development of infrastructure is based on this preference (MR6).

This can be explained using the theory of resistance to change. People are often afraid of the unknown. They know that driving currently works for them despite experiencing problems. However, they do not know what will happen if they take another mode of transportation. They have not yet been able to experience confidence with the alternative and therefore prefer to stay tied to their routines because they are confident (Pardo Del Val & Martínez Fuentes, 2003). The municipality can respond to this by implementing policies that are consistent with promoting facilitation or communicating transportation alternatives such as shared mobility (ER2). As Rik Braams from TNO and the Ministry of I&W (ER2) notes, *"The simplest definition of policymaking is behavior change; you want to effect a change in people's behavior."*

2.3. URBAN TECHNOLOGICAL INNOVATION

Urban innovation is the process of improving urban environments through the development and adoption of new ideas, primarily technological and socio-technical. It involves interactions between stakeholders and involves the use of new technologies to alter collaboration structures and routines in the public domain. By examining the interactions between technological and social changes, it is possible to discover the innovation's associated technologies and social effects (Williams, 1997). In recent years, the driving force and location of (social) innovation in the public sector have shifted from being government-centric to a cooperative approach between government and other actors (Hartley et al., 2013; De Vries et al., 2014). Governments should consider actors capable of implementing public innovation, fostering interactions between companies, governments, knowledge institutions, and citizens to enhance the innovation process and create value for all stakeholders (Hartley et al., 2013). Developers of new technologies need to take into account the adoption capabilities of (government) organizations and citizens (Meijer & Thaens, 2018). Both municipalities and industry representatives indicate that, implementing things such as data standards and working with them costs a lot of money, especially if you are already stuck to routines (MR3; IR3). The integration of new technologies like APIs into existing systems and processes is challenging, necessitating careful consideration of adoption capabilities and financial implications (IR3). In addition, many smart mobility innovations arise within the private sector. By giving space to and facilitating these companies, innovations can continue to develop within the public domain and have a positive effect on the urban area when data and knowledge from these companies is shared with municipalities (Gemeente Den Haag, 2020b).

2.4. COLLABORATION

Policy decisions are initially determined and developed solely by the municipalities, often excluding market parties from the process. From the shared mobility provider's perspective this prevents optimal results (ER5; IR4). Dani Sprecher From MyWheels (IR4) emphasizes this, stating, *"Six months earlier, we should have been at the table commenting on all those turns so that they could decide together which one was the right one."* The goal is to find an overlapping common interest between



mobility users, the government, and mobility providers, as noted by Ferdinand Burgersdijk from the European Commission (ER5). Municipalities also recognize this in their new goals and development strategies. The ultimate target group should be at the heart of development (Maltha et al., 2021). Through collaboration, the municipalities are bringing technological opportunities together with where the need is for practical solutions in public space (Gemeente Amsterdam, 2023).

Within the development of smart cities, collaboration spanning local, regional, and international levels is essential (Liu et al., 2024; Maltha et al., 2021; Gunterman, 2020). Municipalities often acknowledge societal issues and seek solutions, however, the national government fails to take a concrete role in this collaboration and development. The national government should take on the role of coordination and leadership to facilitate innovation and the provision of new mobility options (MR1; MR6; MR7), but should not go too far in this (ER1). As more private sector companies arrive within the public space, municipalities should seek collaboration opportunities with them (ERTRAC, 2021). Governments often overlook the importance of information in decision-making and development plans, resulting in missed collaboration with local governments and stakeholders. More frequent exploration is needed to substantiate choices and include stakeholders in addressing societal issues (Boeije, 2019).

The development of smart cities holds the greatest potential in metropolitan regions that collaborate closely. However, there are many differences between municipalities, which impede development and collaboration. Martijn Arets from Professional Outsider Consultancy (ER3) points out that *"Some municipalities have a vision, while others do not. Some have the resources to think about these issues, while others do not."* The integration of smart city governance into traditional systems may lead to challenges that require sustainable and multi-stakeholder participation strategies (Nesti, 2020). In this development, the use of data is an opportunity for smart and sustainable development (Yigitcanlar et al., 2019). Similarly, D. E. Mills et al. (2021) recognize the importance of authentic collaboration between urban governments, private organizations, and citizens for achieving urban development goals.

By collaborating with and using data of private companies, governments can create win-win scenarios. The joint investment can result in financial gain for private companies and improved urban development for governments, meeting the needs of residents (Ecorys, 2021). Digitalization is crucial for solving societal issues, and collaboration between local governments and stakeholders is essential for sharing data and knowledge. The data strategy within municipalities therefore provides direction and coherence in creating data-driven initiatives (VNG, 2024). Despite the promised benefits of these collaboration initiatives, the associated risks must also be analyzed and mitigated (ERTRAC, 2021).

2.5. POLICY MAKING

Data is needed for short-term policy cycles for traffic regulation and policy adjustments, emphasizing the need for more competent urban planners and policymakers to interpret data and the importance of data for promoting smart and sustainable mobility (Liu & Dijk 2022). Specifically, to achieve traffic adaptation, there is a greater need for real-time traffic and mobility-related data in short-term policy cycles (X. Liu & Dijk, 2022). Leveraging data and integrating it with smart technologies can innovate urban living and management (França et al., 2021; Wang et al., 2021). However, understanding challenges such as data isolation and limited knowledge derivation from data is essential (A. Wang et al., 2021; Sarwat, 2015). Sarwat (2015) and Silva et al. (2018) underscore the importance of data management techniques like IoT to develop effective smart cities that efficiently manage and process large amounts of urban data, leading to a better understanding of urban mobility patterns and more informed urban planning and management. By leveraging the power of big data, cities can make more



informed, timely, and effective decisions, leading to smarter and more sustainable urban environments (Sarwat, 2015; Silva et al., 2018). Oregi et al. (2015) found that the use of ICT tools can significantly improve sustainability performance and facilitate stakeholder communication. However, conflicts and synergies exist between different policy levels, and better policy alignment for sustainable urban mobility still needs to be addressed (Oregi et al., 2015). Ferdinand Burgersdijk from the European Commission (ER5) highlights the complexity of this multi-level regulation: *"If we have regulated it in Europe, then we have to regulate it again nationally and then we have to regulate it again locally, whether provincial or municipal."*

2.6. NEW MOBILITY SERVICES (NMS)

Developments in technology have led to the emergence of new forms of mobility within cities, referred to as New Mobility Services (NMS) (UITP, 2020). According to the European Road Transport Research Advisory Council (ERTRAC), the definition of New Mobility Services (NMS) is not vet complete. However, it is recognized that it refers to new actors, new types of ownership and business models, new vehicle types, and new uses of public space (see Figure 1). This necessitates new regulations to shape the demand, impact, and regulation of these developments (ERTRAC, 2021). Additionally, researchers from TNO state: 'New transport services are defined as car, bike, moped, or scooter sharing, where a fleet of vehicles can be accessed by a large group of users; ridesharing (carpooling or vanpooling), in which rides in a vehicle are shared; and on-demand ridesourcing (pooled or individual), where a driver offers an on-demand ride to a passenger. "(TNO, 2020). With the changing mobility demand and supply, there is a transition from traditional transportation, often associated with a fixed schedule, to NMS, which are frequently aligned with people's on-demand mobility needs (ER1). These NMS are also characterized by being able to be shared, which facilitates the transition from private (car) ownership to shared mobility (ER1). NMS have the potential to drastically change cities, making it essential to assess the advantages and disadvantages of this innovation (ERTRAC, 2021). NMS is also part of the MaaS development in which municipalities see opportunities to innovate urban mobility. For further information about MaaS see Appendix C.



Figure 1 Overview of various New Mobiliy Services (TNO, 2020).

2.6.1. IMPACT OF INNOVATION

This transition aligns with global initiatives such as the United Nations' (UN) Sustainable Development Goals (SDGs). The added value of NMS and particularly shared mobility, is actually that you have a more efficient use of resources (ER5). However, these technological advancements



also lead to unforeseen impacts that are not yet fully understood. Despite the positive promises, there are also concerns and mistrust about the impact of innovation (UITP, 2020). Companies should establish accountability and research is needed to measure the positive impact of their services and products before implementation. This includes making the service accessible to everyone and researching the type of transportation their products replace to gain a broader context of their contribution to sustainable mobility (ER3).

2.6.2. DISRUPTION AND CHANGE

Urban mobility systems are at a turning point, poised for disruption and change (UITP, 2020). Cities want to understand the impact of NMS on travel behavior and calculate where and how they can design mobility in the city and prevent disruption. (Bidasca & de la Quintana, 2020). Rik Braams from TNO and the Ministry of I&W (ER2) notes that within the Ministry of I&W, there is significant concern about disruptive innovations like Uber, which extensively disrupted the taxi domain. "There is an underlying narrative, a kind of fear of a second Uber. Essentially, there is a risk in not having innovation on the radar in time. This is a major reason for officials at the Ministry of I&W to engage with innovation and maintain contact with the market. It's basically out of a fear that the field will be disrupted, and you won't notice it in time, causing it to take you by surprise," explains Rik Braams from TNO and the Ministry of I&W (ER2). Jeanette van Eijk from Greenwheels (IR4) indicates that shared two-wheel providers rushed the urban areas and offered significantly more vehicles than carsharing providers, suggesting the importance of regulating two-wheel providers, but criticizing the need of car-sharing regulation as it is not disruptive: "Our entire fleet consists of 2800 cars. (...) So, we never come to cities with an 'I'll just drop 10 cars here' approach. We do have that gradually in some cities, but we are expanding very step-by-step." In this sense, disruptive innovation refers to innovations where a new market is created by offering different values than the current market. Often this is achieved by being simpler, easier, or cheaper than the competition. Market capitalization is a phenomenon where disruptive innovations start in a niche market and can eventually dominate and replace competition, often criticized by incumbent businesses and stakeholders as undesirable and harmful compared to current practices. Nevertheless, these innovations are often able to evolve and create a new reality in which the industry and user behavior are influenced (O'Reilly III & Tushman, 2021). Ferdinand Burgersdijk from the European Commission (ER5) emphasizes the importance of adaptability in the face of such changes, quoting Charles Darwin: "It's not the strongest nor the most intelligent of people that survive, but the ones most adaptable to change. We need to adapt to change."

2.6.3. INFRASTRUCTURAL CHANGES

The availability of NMS depends on the characteristics of the city (MR3; ER3; IR2; IR3). A city with good biking lane infrastructure is better suited for initiatives that offer two-wheelers. Other infrastructural factors also play a role, such as the number of parking spaces, city congestion, and the availability of mobility hubs (MR2; MR3). To integrate NMS into the city, urban planning, and policymaking for the facilitation of NMS in these areas are essential (MR2). Ferdinand Burgersdijk from the European Commission (ER5) underscores the dependency of NMS on public infrastructure: *"The bike rental company can only make money because they can use public amenities."*

Experts suggest that standardizing shared mobility in urban development can improve societal benefits and address the central spatial development issue of urban space utilization (ER3; ER5). This approach increases efficiency by using fewer resources when shared mobility is used more frequently than private cars (ER3), and optimizes urban space (ER5). By offering shared mobility within the city, fewer private cars and parking spaces will be needed, as Jeanette van Eijk from Greenwheels (IR4)



indicates a single shared car can save up to 14 parking spaces. This aligns with municipal goals regarding green spaces and mitigating urban heat stress (IR3).

Over recent years, there has been a shift from private ownership of transportation means to shared transportation. This shift is causing a restructuring of infrastructure within the built environment. One approach gaining traction is the STOMP-procedure, a principle within the mobility sector that is now increasingly being used as a standard for new developments within municipalities (MR6; ER1).

The International Association of Public Transport (UITP) envisions a future where public transport is the backbone of urban mobility systems, with NMS integrated to enhance sustainability and reduce reliance on private cars (UITP, 2020;Kamargianni et al., (2016). As cities grow larger and become more densely populated, it becomes increasingly difficult to further facilitate and expand (public) transportation. Therefore, good cooperation with these transport providers, as well as with users, is needed to identify the right needs. NMS provide solutions by offering transportation within the city as a first/last mile transport provider (ER2; IR1), transportation from peri-urban to inter-urban areas (Bidasca & de la Quintana, 2020). Niels Wiersma from municipality Eindhoven (MR1) explains, "*It is also in our policy of shared mobility and in the larger mobility plans of the municipality that we like to see shared mobility as a very good addition to existing forms of public transport, shaping the solution for the last mile."* A shared two-wheeler industry representative (IR1) adds, "*Our goal is to play in partnerships with public transport, to address the first and last mile.*"

2.6.4. MUNICIPAL ROLE

To adapt to digitalization and NMS municipalities need to change their role to manage the urban mobility system effectively (ERTRAC, 2021). Municipalities must all ask themselves what role it has in the city's changing (digital) infrastructure (MR5). A policy officer from municipality Rotterdam (MR6) highlights the need for municipalities to take on a director's role to ensure that the end user benefits from the mobility transition: "The municipality must take the director's role to ensure that the end user benefits from the mobility transition and that the new forms of mobility are offered in the right way and the right quantity. This is being done on individual subjects, but a coherent whole is not yet being realized." Despite this, municipalities cannot yet confidently assure residents that they don't need their own cars because suitable alternatives for getting from A to B are available, particularly in urban areas. "The potential is there, only we don't see the separate innovations converging within the municipality." adds a policy officer from municipality Rotterdam (MR6).

2.7. DIGITALIZATION

The rapid digitization of infrastructure is transforming traditional cities into socio-physical-digital ones, integrating digital dimensions with social and physical aspects of urban life. This new paradigm creates a triangular relationship in urban functioning, but the central point between these aspects is becoming larger and more integrated in physical reality, potentially leading to different consequences (MR5). Digitization offers governments many opportunities to collaborate with NMS parties, but in practice, it is still insufficiently utilized. This also applies to the use of open data and open information (Boeije, 2019). Collaborating on these challenges is operational, meaning that parties learn about each other's processes and effectively utilize each other's expertise and data for development. For this to happen, governments need to map out the data landscape and ensure that employees understand data and develop digital skills (Keur & Gunterman, 2021).

Data is often seen as a private good, even though it can have a significant impact on societal problems. When data is kept private, its public value cannot be fully realized and socially undesired data



monopolization by (large) private companies will happen (El-Dardiry & Overvest, 2019). The rise in digitality in daily life and urban development makes cities more vulnerable to private companies, raising concerns about government dependency and control, as governments may become captive customers, hindering independent urban infrastructure management (MR5), indicating demand for stringent market regulations (Rijksoverheid, 2021c).



3. THEORETICAL FRAMEWORK

This chapter presents the theoretical framework used to guide the research and give meaning to the results found. The first part of the framework is built upon the framework of Susha et al., (2017). Additionally, the chapter addresses the challenges identified by Diran & van Veenstra, (2020).

To successfully investigate how Dutch municipalities (try to) leverage ride-hailing trip data and what the current supply and demand of this data is within smart cities in the Netherlands, this research needs a suitable theoretical framework. Within the literature, there are several frameworks on smart and sustainable urban planning such as the smart city initiative framework by Ooms et al., (2020), which addresses the different phases and related aspects of smart city development. But also collaboration/co-creation frameworks are often found in the literature. Such as the Quadruple Helix framework by Paskaleva et al., (2021a), which emphasizes the importance of collaboration between multiple stakeholders (government, industry, academia, and citizens) for the success of smart city projects. Or the big data utilization frameworks such as those used by Silva et al., (2018), J. Wang et al., (2017) or Türk et al., (2021). However, there are relatively few practical theories and frameworks on data supply and demand.

Nonetheless, as a starting point for building a theoretical framework for this research, the taxonomy of Susha et al., (2017) is used. This concerns a taxonomy of data collaboratives in which data gets leveraged to address societal challenges. The taxonomy is constructed to represent the supply and demand of data within collaboratives (see Figure 2). However, rapid technological developments may affect the underlying variables of the taxonomy and further research in supply and demand matching infrastructures may discover new issues (Susha et al., 2017).

This research will specifically addresses the



Figure 2 Dimensions of the taxonomy of data collaboratives based on the supply-demand relationship (Susha et al., 2017).

inner circle, the data layer, from the diagram. This includes dimensions such as **data type**, **data content** and **data level** for the data supply side and **policy problem**, **use purpose**, **expected outcome**, and **continuity** for the demand side. Herein, according to the authors, **data type** is understood as, for example, consumer or user-generated data. The **content of the data** includes, for example, words, locations, behaviors, transactions, and nature. The **level of data** refers to the specificity at administrative levels, and the diversity of data providers. For the demand side, the framework classifies the **policy problem** as unspecified or specified. This refers directly to whether the municipality wants to implement the data purposefully or wants to have the data first and then see what it can be applied for. The **purpose of use** is divided into three groups, being; *primary use*, in which the data is used directly for what it was collected, *secondary use*, in which the data is used for something similar, *tertiary use*, in which the data is used for something else that was initially thought, or *end use*, in which data is processed and the result is used by others. The **expected outcome** of the data collaborative concerns the determination to use the data for policy intervention, data science, or data-driven innovation. Last, **continuity** of collaboration is named in which the influx of data becomes apparent. This can be on-demand, event-based, or continuous (Susha et al., 2017). To match data



supply and demand, the concepts of **facilitation** and **degree of access** are incorporated into the theoretical framework. Here, **facilitation** refers to the support for data collaboration, distinguishing between *direct contact, intermediary contact, or coordinating contact through an intermediary.* **Degree of access** concerns the openness of data, with distinctions between *real-time direct access to raw data, a copy of data, modified/enriched data, outcomes of processed data, or open data.*

However, what is missing within this framework is the way of data sharing, **data sharing mode**, on the supply side. To investigate this, a distinction is made, based on the literature, between *voluntary*, *compulsory*, *and intermediate* data sharing (Rukanova et al., 2020; Susha, Rukanova, et al., 2019; Klievink, Van Der Voort, et al., 2018; Vigorito, 2022).

While this adapted framework of Susha et al., (2017) can be used effectively in identifying supply and demand for shared mobility data, it does not directly provide a way to explore the challenges that Dutch smart cities experience in accessing and using data from ride management companies. Despite well-known frameworks by Rukanova et al., (2020) which identify barriers, drivers, and enablers of voluntary B2G data sharing and the governance processes that foster it, and research of Klievink, Van Der Voort, et al., (2018) which adds components such as trust and institutionalization offer some input in this regard, but do not cover the exact challenges. However, Diran & van Veenstra, (2020) do so. They discovered several challenges/barriers to both data collection and data use. Although Diran & van Veenstra, (2020) dedicate their research specifically to heat transition policymaking, these barriers are not expected to differ significantly for policymaking based on shared mobility data, which makes it a suitable starting point for this research. These barriers and challenges are specifically addressed to gathering data from other parties than the municipality itself. See Table 1 below for the identified barriers and challenges.

Data collection	Data use
GDPR and privacy restrictions	Cautious to make decisions based on insights from data
High investments in time and costs	Data lacking quality and consistency
Lack of expertise and skills	Incomplete and missing data
Scattered distribution of data	Data is not always validated
Poor data findability and access	Data lacking detail level
Lacking access rights	Lack of supporting tools
Many formats and standards	Difficult to determine the value and purpose of data
Lack of awareness, trust, and openness in data sharing	Data preparation is resource-intensive
	Difficult to link, analyze, and visualize data
	Resources at the limits to process the quantity of data
	Legal limitations to data use

Table 1 Barriers and challenges in data access and data use by policymakers

By combining and adapting these theories, the foundation has been formed for identifying supply, demand, and challenges for the study of how Dutch municipalities (try to) leverage ride-hailing trip data and what the current supply and demand of this data is within smart cities in the Netherlands.



4. RESEARCH METHODOLOGY

This chapter provides an overview of the research methodology used in this study aimed at understanding demand, supply, and use of shared mobility data by Dutch municipalities. It delves into the research philosophy, type, design, sampling strategy, data collection methods, and data analysis procedures. Furthermore, the chapter addresses the research quality indicators, limitations, and ethical considerations, ensuring a robust and transparent methodological approach.

4.1. RESEARCH PHILOSOPHY

This research aims to understand the supply and demand of shared mobility data within Dutch municipalities and how they aim to use this data for urban development and policymaking in response to changing urban dynamics in terms of urbanization and changing work patterns. The interpretivism philosophy is used to understand different subjective perspectives and experiences of stakeholders in urban development and shared mobility data sharing.

The iterative nature of the philosophy allows for exploring the nuanced and context-specific nature of data sharing in the mobility sector. This philosophy enables a holistic approach to understanding qualitative insights on stakeholders' experiences and perspectives on urban development and data data-sharing collaborations and data use practices (Clarck et al., 2021).

4.2. RESEARCH TYPE & DESIGN

This research follows an abductive, qualitative research approach, aligning with the interpretivism philosophy (Clarck et al., 2021). This qualitative approach provides the right opportunities to discover new aspects and explore complex phenomena in shared mobility data sharing through qualitative data collection as well as confirming theoretical aspects. This approach facilitates an in-depth understanding of various stakeholder perspectives, challenges, and opportunities in data sharing, acquisition and utilization for evidence-based decision-making in urban development and policymaking. It allows for the emergence of patterns and themes, revealing the complexity of the subject. The flexibility of the research approach also allowed small adjustments in research strategy and direction based on the acquired data (Clarck et al., 2021). More specifically, this research follows an exploratory multiple case study design, which is a flexible method that allows for in-depth data collection and analysis, which leads to a deeper understanding of the phenomenon and adjusting research questions and directions (Clarck et al., 2021). Conducting multiple case studies has provided opportunities for a comprehensive exploration and understanding of the shared mobility ecosystem and its data sharing with municipalities, effectively identifying patterns and comparing different examples which reinforces the validity of the findings (Eisenhardt, 1989).

4.3. SAMPLING STRATEGY

This research focuses on qualitative data collection through participants who meet specific criteria. For this purpose, the sampling strategy is deployed by a combination of criterion and purposive sampling (Palinkas et al., 2015). This ensures selecting cases that are likely to replicate or extend theory and provide opportunities to provide rich insights and deepen understanding of the shared mobility data sharing phenomenon (Eisenhardt, 1989).

First, criteria were established to identify and select appropriate Dutch cities and participants. In this, the main criterion is the selection of cities with more than 150,000 residents according to the Central Bureau of Statistics (CBS). This condition is tied to the presence of shared mobility companies, which primarily operate in large cities, and the volume of trips made within the city, which strengthens the



quantity of data available (Tao et al., 2019). Additionally, larger cities, as they are more innovative, are more likely to engage more extensively and deeply with data usage, thereby making a more significant contribution to this research (Carlino et al., 2007).

The selection of other stakeholders goes by relevance, availability, and applicability in the selected cities. This purposeful sampling is based on maximizing variation in the experiences and perspectives of key stakeholders in urban development in Dutch municipalities. These key stakeholders are:

- 1. Urban planners and policymakers who are involved in shared mobility and (sustainable) urban development. They offer valuable insights into decision-making processes and the role of (trip) data in (smart and sustainable) urban development and policymaking. This information can help identify challenges and opportunities for data access and utilization, understanding data needs and supply, and regulations for data collaboration and sharing.
- 2. Researchers and (Data) experts in the field of (smart) urban development and data collaboration/ utilization. They shape (trip) data collection, supply, and utilization, and provide insights on successful practices related to (smart) urban development, and data collaborations, forming connections between theory and practice. Incorporating their critical perspectives uncovers innovation possibilities and potential solutions for urban development and policymaking based on shared mobility data.
- 3. Industry representatives of shared mobility, holding (trip) data within the selected municipalities. They offer insights into data supply and the implications of platform-, and shared mobility companies in urban planning and policymaking. Understanding these factors helps identify data sources and collaboration opportunities.

4.4. DATA COLLECTION

This study uses primary and secondary data to understand shared mobility data demand, supply, and use, in smart and sustainable urban development. Primary data has been obtained through in-depth, semi-structured, one-on-one interviews with policymakers/urban planners, researchers/experts, and industry representatives in the field of shared mobility data in Dutch municipalities. The qualitative research technique allows for both anticipated answers and spontaneous discoveries, ensuring participants are not influenced by others or the researcher (Clarck et al., 2021). The conduct of interviews is location-independent for the researcher, using both physical and digital methods, such as Microsoft Teams or telephone. Audio-recorded and transcribed interviews, in agreement with interviewees, are used for thematic analysis in NVivo (Clarck et al., 2021). See Table 2 for the precise list of interview respondents.

Number	Name	Code	Role	Organization	Duratio
		name			n
		Municipalities			
1	Niels Wiersma	MR1	Data and Platform Strategist	Municipality of Eindhoven	33:42
2	Emma de Wijs	MR2	Policy Advisor Urban Planning	Municipality of The Hague	51:28
3	Undisclosed	MR3	Policy officer	Municipality of The Hague	52:38

Table 2 Interview respondents



4	Antoine Gribnau	MR4	Advisor Geo Information and open data	Municipality of The Hague	21:19
5	Roland van der Heijden	MR5	Program Manager Digital City	Municipality of Rotterdam	38:37
6	Undisclosed	MR6	Policy Officer	Municipality of Rotterdam	42:14
7	Gemma Schepers	MR7	Project manager Smart Mobility	Municipality of Amsterdam	54:23
			Researchers	& Experts	
8	Marlous Hovestad & Rosanne Klerx	ER1	Mobility data expert, Community Manager MaaS	Nationaal Toegangspunt Mobiliteitsdata	56:21
9	Rik Braams	ER2	Senior Scientist for Transformative Government/ Innovation Officer Innovation in Mobility	TNO/ Ministry of I&W	36:32
10	Martijn Arets	ER3	International Platform Expert	Professional Outsider Consultancy	44:00
11	Edwin van den Belt	ER4	Software Architect	Dat.mobility & TOMP-API & CDS-M working groups	52:30
12	Ferdinand Burgersdijk	ER5	Data governance, digitalization and mobility expert	European Commission/ FRCB B.V/ UITP	1:17:39
		Industry Representatives			
13	Undisclosed	IR1	Undisclosed	Shared two-wheeler provider	21:26
14	Undisclosed	IR2	Undisclosed	Shared two-wheeler provider	35:11
15	Jeanette van Eijk	IR3	Public Affairs Manager	Greenwheels	1:00:13
16	Dani Sprecher	IR4	Manager Public and Government Affairs/ Co-founder coalitie van Deelauto- aanbieders	MyWheels	42:14

This enumeration lists the four case study cities, each requiring interviews with the municipality. In addition, the research includes interviews with industry representatives (shared mobility providers) and researchers & experts. In this, the municipalities are considered as the data demanding parties as the industry representatives are considered the data-supplying parties. The researchers & experts act as an objective control and explaining group that connects supply and demand and gives context to the



data-sharing phenomenon. The interview guide and questions for each stakeholder can be found in **Appendix H-J**.

To complement primary data, this study uses desk research to gather and analyze secondary data. The study comprehensively reviews and synthesizes existing literature and reports on smart city and data sharing, analyses of data and digitalization strategies, and data sharing initiatives from shared mobility providers. In this way, context is provided, allowing a comprehensive understanding of the topic to be formed. The literature study allows for an understanding of innovations and developments within Dutch municipalities. Starting with secondary data analysis, a data foundation has been formed and research questions and interview questions have been sharpened. Additionally, desk research is used to find policy documents to provide input for municipal case study (thematic) analysis.

The inductive data collection method provides qualitative data while triangulating data collection methods strengthens research validity and reliability (Clarck et al., 2021). The research is iterative, moving between theory and data collection to steer the research but also aims to test the existing framework based on new findings and focus on problem and solution direction, additionally aiming to revise the theoretical framework.

4.5. DATA ANALYSIS

Thematic analysis

To investigate the municipalities as cases (within-case analysis) as well as investigating interviews with industry representatives and researchers/experts, thematic analysis is used. Thematic analysis is a qualitative research method for identifying, analyzing, and reporting themes that emerge within data. It involves reading and re-reading the data to gain a comprehensive understanding and make initial notes to capture context and nuances (Khokhar et al., 2020). The data is then systematically coded using NVivo software. Within the initial coding process, the theoretical framework based on Susha et al., (2017) and Diran & van Veenstra, (2020) is central as this provides the initial codebook (see Table 3). This codebook includes predefined categories facilitating a structured approach for this coding process. This guarantees not only the reliability of the data based on thematic analysis but also contributes to the existing body of knowledge through theory testing (Clarck et al., 2021). This deductive approach to coding ensures alignment with the theoretical constructs and sets the foundation for understanding the data in terms of the established theoretical framework. Additionally, open coding is used to identify new themes within the data. Adding this inductive approach ensures that while the analysis is guided by the theoretical expectations, it remains open to capturing novel insights and deviations from the theory, allowing new themes to be revealed (Clarck et al., 2021). Both deductive and open coding have been done through a combination of in vivo and descriptive coding techniques.

<u>Categories:</u>	Description:
Supply:	
Data type	Nature of data collection
Data content	Specific information the data contains
Data level	Specificity or granularity of the data

Table 3 Initial codebook (categories derived from theoretical framework



Data sharing mode	The method through which data is disseminated or made accessible
Supply & demand matching:	
Facilitation	Support mechanisms to aid data collaboration
Degree of access	The extent of data accessibility
Demand:	
Policy problem	Predetermined use case for data
Use purpose	How directly the date serve the initial purpose
Expected outcome	Anticipated result of data use
Continuity	The frequency and regularity of data supply
Data acquisition challenges	Problems faced in collecting data
Data utilization challenges	Difficulties in using collected data effectively

Next, axial coding is applied to discover the relationships between codes and how they refer back to the theoretical constructs in the framework. The next step is selective coding. Here the focus is on refining and selecting core themes that are most relevant to the theoretical framework. To do this, data is synthesized into broader themes to discover the essence of the theoretical constructs and the relationships between them. For each theme, it is reasoned how and why it belongs to the theory or why not (Clarck et al., 2021).

To limit theoretical biases and assumptions regarding the coding process, there is continuous reflection on the process. This iterative process does not follow a chronological approach, allowing for a richer and more detailed representation of complex data (Clarck et al., 2021).

Cross-case analysis

This study shortly engages in cross-case analysis to compare shared mobility data supply, demand, access, and utilization across four Dutch municipalities and shared mobility providers (which are distinguished based on offering cars or offering two-wheelers). By systematically studying themes, derived from thematic analysis based on interviews and documents, within each city, and interviews among shared mobility providers, the analysis facilitates the discovery of patterns, variations, and case-specific insights, enhancing the broader understanding and interpretation of the thematic findings (Clarck et al., 2021; Eisenhardt, 1989). This method not only allows for contrasting and validating themes across different cases but also broadens the perspective by highlighting the complexity and diversity of implementing shared mobility data sharing in various urban contexts.

During the cross-case analysis, themes and content from each city or shared mobility provider are meticulously compared. This comparison leverages the theoretical framework as a tool to examine various aspects of the cities or shared mobility providers but remains flexible to incorporate newly emerging themes and concepts that arise during the analysis. This dynamic approach ensures that the theoretical framework is not only applied but also evaluated for its effectiveness in capturing the nuances of each case.

This research is profoundly explorative, which results in findings (especially from cross-case analysis, as they are based on single interviews, and a limited number of documents) that are preliminary, but suggest directions for further research which includes more extensive data collection.



4.6. OPERATIONALIZATION

Research model

Figure 3 provides a clear visualization of the data collection contribution and process flow. However, the scheme excludes specific literature review topics and documents to maintain clarity. It should be noted that this research model serves as a guideline within the study and is not exclusively adhered to. This allows for deviation from the direct contribution to a topic when a respondent unexpectedly shares relevant information. Additionally, within this model, the direct relationship between thematic analysis and its contribution to the theoretical framework and research questions is omitted to maintain clarity. The thematic analysis for each respondent group was conducted in the same manner, resulting in an integrated view of the supply, demand, and challenges of shared mobility data sharing in Dutch municipalities.



Figure 3 Research design (including data collection, analyses, theoretical framework, and answering research questions)

4.7. RESEARCH QUALITY INDICATORS

Internal reliability

To maintain consistency in data collection, a semi-structured interview guide was created for each stakeholder group (municipalities, researchers & experts, and industry representatives). Before conducting interviews with the respondents, these guides were tested through pilot interviews. Based on the feedback from these pilot interviews, the interview questions were adjusted and refined for clarity and relevance. Additionally, the study was designed and conducted by a single researcher, minimizing differences in interpretation within the study.

External reliability

To ensure trustworthiness and making sure the same results can be obtained by another researcher, the research process includes a transparent methodology, data collection, analysis, and processing strategy, and is built on theoretical frameworks. Although this research is explorative and interpretive it provides clear and transparent findings based on external research, and individual findings and experiences. Additionally, this research clearly states its limitations in the process and results.



Internal validity

To ensure the integrity of the research design, causal relationships are not biased or influenced by external factors. Triangulating data allows findings to be tested and validated against each other. In this study, this is done through interviews from multiple perspectives, integration of policy documents and reports, as well as existing literature and studies. Additionally, feedback was sought from third parties throughout the research process, and the content of the research was discussed with interviewees and the research supervisor from Utrecht University to identify and address potential biases or inconsistencies.

External validity

This research focuses on the development of smart cities in the Netherlands, selecting four municipalities based on sampling criteria. This specification allows for detailed research specifically tailored to the Dutch urban development and policymaking sector. However, the generalizability may be limited due to the specific focus on the Netherlands and the different development visions and cultures of other countries and cities/municipalities.

4.8. DATA ETHICS AND PRIVACY

This research prioritizes data ethics and privacy in data collection, handling, and storage. Informed consent is requested before data acquisition, and participants' processing and use are transparent. Data has been handled accurately and confidentially, with privacy and security guaranteed by storing it on encrypted servers from the University Utrecht and processing it anonymously if desired. See **Appendix E and F** for the informed consent form and information about privacy and confidentiality/ ethical issues.



5. RESULTS

This chapter provides an in-depth examination of the thematic analysis outcomes derived from the interviews with municipalities, researchers & experts, and industry representatives. In addition, these results are supplemented and confirmed by the thematic analysis of policy documents, reports, and articles.

First, the data demand is elaborated according to the terminology provided by Susha et al., (2017). This directly addresses the first sub-question of this research: "How is the Dutch municipalities' shared mobility data demand formed and what data do they need for urban planning and policy-making purposes?" Then the supply of shared mobility data as well as the data acquisition challenges are elaborated according to the continuation of the framework provided by Susha et al., (2017). This also addresses the second sub-question of this research: "What is the current state of data-sharing initiatives between shared mobility providers and Dutch municipalities, and how is this data sourced?" Following the framework of Susha et al., (2017), data facilitation and opportunities are also shown. These are the developments supporting shared mobility data sharing. The data utilization challenges are presented accordingly. Combined, this contributes to answering the third sub-question: "What are the technical and organizational challenges and opportunities experienced by Dutch municipalities in accessing and using data from shared mobility providers active in the Netherlands?"

In the last section, a cross-case analysis is presented. In this section commonalities, differences and exceptions between shared two-wheel providers and car-sharing providers as well as among Dutch municipalities are presented to understand the complexity of the topic.

5.1. DUTCH MUNICIPAL DATA DEMAND

Data is increasingly seen as a valuable resource in urban planning and management (ER5). The specific data demands of municipalities vary significantly based on several factors, including the market share of the provider in the city, the frequency of use, and existing data (MR3). A shared two-wheeler industry representative (IR2) further explains: "*The data requested varies by municipality*. *Generally, you have standard KPIs like the number of trips made, the duration of those trips, and the average distance of a trip. There are also more detailed questions, such as the start and end points of each trip, where most trips begin, and in which area.*" Additionally, the data volume significantly influenced the municipality's data demand, as a policy officer from municipality The Hague (MR3) highlights: "Look, the amount of data a municipality receives does matter. It also depends on the volume of vehicles in the city. If only 80 drivers are using Bolt cars among a large number of people who have regular cars, the impact is low. Consequently, the need for extensive data and close monitoring is also much lower."

The approach to data sharing also differs between shared two-wheelers and car-sharing companies. Two-wheeler providers are often required to share data through agreements and licensing, whereas car-sharing providers share less data and are more reserved. A policy officer from municipality The Hague (MR3) notes, "*The shared scooter providers share quite a lot, and the car-sharing providers also share some information. However, this is more on a static basis and less in real-time. They are indeed a bit more reserved.*" Municipalities now often include specific data requirements in their permits. A shared two-wheeler industry representative (IR2) states, "*Municipalities now include a fixed list of data points they want to receive from us in the permits.*" Gemma Schepers from municipality Amsterdam (MR7) explains the challenges faced by car-sharing providers, particularly those associated with large car brands: "*The car-sharing companies find it very difficult. We've been in discussions with them for about a year and a half or two years now, and they are very resistant. They*



find it quite daunting. This is particularly because some car-sharing companies are part of large car brands, which find data sharing more intimidating."

However, changes are underway. For example, the municipality of Rotterdam is now requiring carsharing- and shared two-wheeler providers to share specific data as part of their permit conditions (MR6). A policy officer from municipality Rotterdam (MR6) outlines this new approach: "In the municipality, we have regulations in place, so upfront in the permit conditions, it is stated that providers must share data. This applies to both car-sharing and shared two-wheelers. This is regulated differently. Shared two-wheelers are given specific permits to operate. They may be allowed a certain number of vehicles per vehicle or as a provider. However, this system is being revised and optimized. The policy regulation specifies that this optimization applies to car-sharing. The implementation decision outlines the conditions they must meet. This is how it is organized."

According to shared mobility providers, municipalities must articulate a clear purpose for the requested data and ensure they operate within legal boundaries to motivate private companies to share their data (IR3; IR4). Dani Sprecher From MyWheels (IR4) highlights the importance of understanding these frameworks: "The government, cities, and data requesters—such as urban officials, ministries, but primarily cities—are all different. It may sound a bit generalizing, but they must clearly understand the frameworks within which they are actually allowed to request information."

When municipalities request data without a clear societal value or legal grounding, it can lead to tensions. Especially car-sharing providers do not appreciate unclear or even, from their perspective, illegal data demands. Dani Sprecher From MyWheels (IR4) further explains, "At some point, you encounter things where you can say, 'Okay, but this no longer relates to the concept of whether it has societal added value or not.' It's really about whether we run our business one way or another. These are business economic choices that municipalities should not be interfering with at all." He adds that municipalities should avoid requesting data for short-term issues: "At its core, data sharing should be about long-term matters. So, you shouldn't request data for things that are not related to the short term." Jeanette van Eijk from Greenwheels (IR3) thinks that municipalities often overreach in their data demands: "We have come to realize that, in my opinion, much more is being requested than is actually allowed, at least from us. Let me put it this way: we do not comply with that." Dani Sprecher From MyWheels (IR4) confirms by mentioning a municipal data request requesting useless things such as car color, car action radius, etc. This suggests a need for municipalities to better align their data requests with legal and practical constraints. However, municipalities find it challenging to obtain a legal basis for requesting this data. As Niels Wiersma from municipality Eindhoven (MR1) points out: "You really need a legal foundation to obtain personal information. For such an application, it is indeed quite complicated unless you have a very specific purpose."

Industry representatives suggest municipalities sometimes exhibit a shortsighted approach when imposing data demands on shared mobility providers. Dani Sprecher From MyWheels (IR4) recounts an instance of this mindset where a policy officer said: "Your cars are in public spaces, so you must share all your data.' That was his way of thinking. He genuinely believed that." Says Dani Sprecher From MyWheels (IR4). Jeanette van Eijk from Greenwheels (IR3) also observed a tendency for rushed data demands without sufficient justification: "What we noticed was a tendency to quickly assume, 'Just arrange this,' or 'Just share that.' In fact, we were the first among the shared mobility providers to ask: 'Why do you want this information, and what is the logic behind it?' It's not that we don't want to share; we've actually shared quite a bit of information by now."



5.1.1. POLICY PROBLEM

Municipalities often face difficulties regarding densification and urbanization, prompting them to seek innovative solutions within the city (MR1; MR2; MR7). Shared mobility emerges as a key component in addressing these challenges. According to a policy officer from municipality The Hague (MR3), shared mobility is viewed as a promising solution in the mobility transition, especially as cities grapple with overcrowded conditions and limited space. "*Cities are overcrowded with cars, and there's insufficient space while more people are moving in. We need a solution. This solution involves encouraging more people to walk, cycle, use public transport, and also utilize shared mobility,"* the policy officer from municipality The Hague (MR3) explains.

In rapidly growing cities, densification is particularly pressing due to the influx of new residents and limited land availability for new construction. Niels Wiersma from municipality Eindhoven (MR1) points out that Eindhoven's rapid growth, fueled partly by companies like ASML, necessitates significant densification, especially in the city center. This increasing urban population results in less public space, demanding careful decisions on its usage. Gemma Schepers from municipality Amsterdam (MR7) emphasizes, "More and more people are moving into the city, resulting in less public space. Therefore, we must make very careful decisions about how we use the public space we have." This sentiment is echoed by a policy officer from municipality Rotterdam (MR6), who notes that the integration of various urban ambitions converges in the public space, presenting a complex challenge. To address the lack of urban space, municipalities recognize the potential and necessity of shared mobility, particularly shared two-wheelers. They anticipate these innovations will take up a significant share of mobility, facilitating continued urban development. As Gemma Schepers from municipality Amsterdam (MR7) predicts, "Bicycles, scooters, and small vehicles will take up a much larger share of public space in the city, and this trend will continue."

Municipalities have set goals often framed as policy problems. These policy problems are the practical use cases behind the data demand, such as **spatial planning, nuisance prevention, and sustainability** (MR1; MR2; MR3; MR4; MR7). Shared mobility offers an alternative to private vehicle ownership, contributing to municipal spatial development plans and sustainability goals. A policy officer from municipality The Hague (MR3) explains, "As a municipality, you need to have insight into the mobility within your city to manage it effectively, whether something goes wrong, you want to enhance something, or you aim to expand certain aspects." Gemma Schepers from municipality Amsterdam (MR7) adds that gaining insight into how people use shared mobility allows for better management and public space design. "For example, if an area sees high usage, you might need to allocate more space for parking," she says. Niels Wiersma from municipality Eindhoven (MR1) explains that this is possible using data regarding the start and ending points of shared mobility. This approach helps municipalities optimize the physical public space effectively (MR7).

Reducing private vehicle ownership also leads to fewer vehicles in public spaces and reduced greenhouse gas emissions. Rik Braams from TNO and the Ministry of I&W (ER2) highlights the policy shift towards shared transport to achieve these environmental goals from the Ministry of I&W perspective, stating, *"They say that personal transport is not always the best solution everywhere, so we need to look at shared transport. This shared transport should provide the same convenience and seamlessness, which essentially requires Mobility as a Service (MaaS)."* Every municipality aims to transition to sustainable and smart mobility, reducing private car use in favor of public transportation and shared mobility alternatives. A policy officer from municipality Rotterdam (MR6) advocates for this direction, suggesting that funding for shared transport should be considered alongside public transport funding to supplement the overall mobility system.



However, municipalities also face challenges related to the nuisances caused by shared mobility. For example, Martijn Arets from Professional Outsider Consultancy (ER3) notes that in The Hague, shared scooters often cause disturbances when large numbers of them move to popular destinations like Scheveningen. To address parking issues, Gemma Schepers from municipality Amsterdam (MR7) explains that digital zones within apps now restrict scooter parking to designated areas, reducing disorganized parking, and indicating the direct application of shared mobility data. Additionally, a shared two-wheeler industry representative (IR2) indicates that Idle vehicles are almost synonymous with nuisance. "*Therefore, it's our responsibility to use the data we collect to allocate our fleet across the service area in a way that ensures an optimal occupancy rate at all times.*"

Despite these efforts, municipalities experience more nuisances with two-wheelers than with carsharing providers (MR2; ER3; IR4). Dani Sprecher From MyWheels (IR4) argues that while scooters cause disturbances justifying certain data requests, cars do not present the same issues. "Officials had a specific basis for their concerns with scooters since they caused disturbances, prompting them to seek certain information. You might think this could also apply to cars, but cars do not cause the same disturbances," he explains. Jeanette van Eijk from Greenwheels (IR3) points out that car-sharing providers and shared two-wheeler providers are often grouped under the same term, despite significant differences between them. "Shared mobility often becomes an umbrella term that includes carsharing, which then also applies to data sharing requirements (...)Then it became a copy-paste situation: 'We also want to know this information from the car-sharing providers,''' Jeanette van Eijk from Greenwheels (IR3) says. This is in line with the municipality's perspective as they only mention nuisance concerning shared two-wheelers and do not mention any nuisance from shared cars (MR3; MR6; MR7).

To facilitate shared mobility and reduce nuisance and inconveniences, both municipalities and mobility providers are exploring the establishment of mobility hubs within cities. These hubs are designated locations where various modes of transportation are offered centrally (MR2; MR3; IR2). A shared two-wheeler industry representative (IR2) notes success in working with municipalities to create specific hubs for bikes, while a policy officer from municipality The Hague (MR3) mentions ongoing efforts to establish more hubs for shared scooters. "Mobility hubs are the places where this could happen, bringing various modes of transportation together. We are still working on linking different data to our data tool to visualize space requirements and determine the square footage needed for such a mobility hub. Where could we realize this within the city?" Emma de Wijs from municipality The Hague (MR2) explains.

5.1.2. USE PURPOSE & EXPECTED OUTCOME

The concrete use or purpose for demanding and using data is a crucial issue within shared mobility data sharing especially from car-sharing provider's perspectives. Shared two-wheel providers are generally already open to data sharing (MR3; MR6; MR7; ER3), but car-sharing providers tend to open up to municipalities when municipalities have a specific and unambiguous use case (IR3; IR4). This cooperative attitude stems from the clear identification of objectives in consultation with policymakers. As Gemma Schepers from municipality Amsterdam (MR7) points out, "*It's quite challenging because everyone thinks, 'I just want parking data,' but the question is, why?*" Car-sharing providers note that when municipalities articulate their data needs clearly, it facilitates cooperation. Dani Sprecher From MyWheels (IR4) illustrates this with an example: "*So you need to state, 'Okay, by 2025, we want only electric cars in the city. Therefore, we need to know if your car is electric or runs on gasoline.' Clear and well-founded. Here's my data. It's that simple.*"



However, collecting data without a justified interest is problematic. As Ferdinand Burgersdijk from the European Commission (ER5) emphasizes the need for municipalities to ask better questions and understand legal and technological constraints: "Collecting data for the sake of collecting is not a justified interest. I want municipalities to ask themselves better questions without being fed the answers. This means adopting a more critical stance, which involves, first, understanding and applying the law better, and second, having a better grasp of technology."

According to interviewed shared mobility providers, such as Amber Mobility (now MyWheels), they have shown openness to collaborative agreements that ensure privacy-sensitive and commercially sensitive information is protected. Niels Wiersma from municipality Eindhoven (MR1) explains their approach: "From the beginning, we stated that this would be a collaborative agreement where we would jointly determine what needs to be shared. From our side, we wanted to ensure that no privacy-sensitive information was shared, and from their side, we understood that no commercially sensitive information would be disclosed."

However, shared mobility providers perceive that they often encounter unwarranted and legally conflicting data demands from municipalities (IR3; IR4). Requests for personal customer characteristics, which providers are not permitted to collect, are an issue according to shared mobility providers. A shared two-wheeler industry representative (IR2) notes, *"Yes, municipalities often request personal characteristics of customers. This is data we simply do not collect from our customers, as we are not permitted to collect it. We do collect that through surveys."* Dani Sprecher From MyWheels (IR4) adds: *"They not only oversimplify by lumping concepts together, but they also consistently take a lax approach to legal matters. They often think, 'We think it's a good idea, so it must be allowed.' When we point out their legal authority, they generally don't appreciate it. "This signals that shared mobility providers perceive municipal data requests as disproportionate. At the same time, according to interviewed municipalities, as well as experts, they are aware of the need for municipalities to formulate clear, legally sound data requests where the municipality needs to think deeply about what the actual problem of the city is to formulate a very good questions and provide a legal rationale for determining the necessary data variables (MR7; ER5).*

Indiscriminate data requests are problematic and legally questionable from the shared mobility providers' perspective (IR2; IR3; IR4). Dani Sprecher From MyWheels (IR4) criticizes this approach, explaining that he is 99% sure that it is not permitted under administrative law: "In legal terms, this is called a 'fishing expedition': requesting all data indiscriminately and then searching for something useful to form policies with. This is not allowed." This is undesirable for the mobility provider as they do not want to be burdened with ineffective data sharing and certainly not with potential confrontation on business practices based on their own supplied data. As a result, mobility providers are often reluctant to address specific data requests, especially those that are or may be tied to privacy or business-related content (IR2, IR3, IR4). Dani Sprecher From MyWheels (IR4) explains, "Look, our data has value. We are happy to share it freely with the municipality, but I do not agree to another commercial party making money from our data."

5.1.3. CONTINUITY

The degree of data sharing depends significantly on the conditions established in agreements between municipalities and shared mobility providers (MR1; MR7; IR2; IR3; ER5) While live, automated data sharing is ideal for municipalities (MR1), this often involves extensive lobbying and negotiation. Gemma Schepers from municipality Amsterdam (MR7) notes, *"Shared mobility providers are willing*"



to share data, but they also want a place in the city and the necessary permits. However, the municipality imposes various requirements."

According to shared two-wheel providers, municipalities typically include fixed lists of data points in permits, with the frequency of data sharing varying by municipality (IR1; IR2). A shared two-wheeler industry representative (IR2) states, "Municipalities now include a fixed list of data points they want to receive from us in the permits. The frequency varies by municipality; some request it monthly, while others want it every six months."

Shared mobility providers also vary in their data-sharing practices. Some have continuous agreements (shared two-wheel providers), while others share data only when requested by municipalities (carsharing providers) or proactively by shared mobility providers in specific cases (essentially only shared two-wheel providers) (IR1; IR2; IR3; IR4). A shared two-wheeler industry representative (IR2) explains, "We share data purposefully with municipalities where we are active. Sometimes, we also share data with municipalities where we are not yet active but want to contribute to decision-making. Primarily, we share data in municipalities where we are active, and this has become a standard part of our permits with municipalities."

A notable difference exists in the continuity of data sharing between shared two-wheelers and carsharing companies. Two-wheeler providers are often required to share data as part of their permits, whereas car-sharing providers are less consistent (MR3; MR6; ER3; IR2; IR3; IR4). Dani Sprecher From MyWheels (IR4) highlights this discrepancy, noting the significant variation among car-sharing providers compared to the more regulated shared two-wheelers. Niels Wiersma from municipality Eindhoven (MR1) explains, "In recent years, we have made agreements with all mobility providers. If they receive a permit to operate shared mobility in the city, they are required to exchange data and make it available according to the established standard." However, according to the shared mobility providers, the continuity of data sharing depends on the municipality's requests and the level of detail required. Dani Sprecher From MyWheels (IR4) observes that issues arise when the questions become more detailed. Initially, municipalities just wanted to know what was happening with the vehicles, so providers gave them averages, but this changed when shared two-wheel providers arrived in the city. Jeanette van Eijk from Greenwheels (IR3) explains, "Initially, they just wanted to know what was happening with our vehicles. So we gave them averages. However, we now notice that more and more municipalities want to know exactly what is happening with each vehicle." This is in line with an explanation from a policy officer from municipality The Hague (MR3), which indicates that municipalities increasingly want to be closer to the data and are transitioning to being obligating actors, which means that it will be putting more pressure on mandatory data sharing. However, as municipalities increasingly seek to mandate data sharing, they must balance their demands to avoid driving providers away. A policy officer from municipality The Hague (MR3) articulates this balance: "On one hand, you want to have the data to create good policies and improve mobility in the city. On the other hand, you don't want to drive providers away with too many obligations and regulations." A policy officer from municipality Rotterdam (MR6) adds that municipalities should see shared mobility as a collective provision and a collective mobility, which indicates that municipalities should be careful with being too strict with shared mobility providers. Understanding the business operations of shared mobility providers is critical. Gemma Schepers from municipality Amsterdam (MR7) highlights that municipalities sometimes impose unrealistic data demands on young companies with slim margins, expecting them to be available for municipal purposes. "We sometimes impose too many unrealistic demands on their growth business model, especially since they are still young with very slim margins, yet we want them available throughout the city. We need to listen to them more, and I



think we, as a city, sometimes fail to do that," says Gemma Schepers from municipality Amsterdam (MR7)

5.2. SUPPLY OF SHARED MOBILITY DATA (AND ITS CHALLENGES)

5.2.1. DATA COLLECTION BY PRIVATE COMPANIES

Mobility providers note that they regularly receive data that might be privacy-sensitive. However, they tend to limit privacy-sensitive data collection, especially if it doesn't serve a specific use case, as they then tangle with legally established privacy restrictions (IR2; IR3). Jeanette van Eijk from Greenwheels (IR3) indicates that they do not track their users' destinations or refueling stops: "We currently do not know where our users go or where they stop to refuel. We do not track that at all. Only if something goes seriously wrong do we investigate, but we are very conscious of the information we request from people. We handle it with great care." A shared two-wheeler industry representative (IR2) explains that they do not collect personal data at all, so they cannot share it: "Municipalities sometimes request personal characteristics such as age or gender, but we are not allowed to collect this information due to General Data Protection Regulation (GDPR) regulations." A shared two-wheeler industry representative (IR2) further elaborates, "We are not allowed to collect personal data that is irrelevant to our work. In some cities, showing a driver's license is mandatory, but those details are processed and deleted in a GDPR-compliant manner." Often, this is also outsourced to a third party for better and more secure processing (IR2).

5.2.2. DATA AS INTELLECTUAL PROPERTY

Big data is more than just the collection of data; it also encompasses its analysis. In essence, big data has the typical characteristics of intangible properties. It requires a significant amount of time, energy, money, and effort to generate (Zhang & Xu, 2023). This results in the high value of data, and therefore it must be protected by the owner to maintain a competitive advantage over competitors (Teixeira & Ferreira, 2019). Therefore, shared mobility providers are generally cautious about sharing data because they consider it an essential resource for their business and necessary to compete with other mobility providers. It is crucial for capturing market share in users and winning contracts through tender processes. Additionally, privacy is one of the biggest concerns for these providers. Much of the data contains personal details or is in some way retraceable to an individual. Due to General Data Protection Regulation (GDPR) legislation, this data cannot be shared (CoE-DSC, 2023).

Dani Sprecher From MyWheels (IR4) emphasizes the value of private shared mobility data, stating, "Look, our data is valuable. We are happy to share it freely with the municipality, but I do not agree to another commercial party profiting from our data. That would be completely unacceptable." Dani Sprecher From MyWheels (IR4) further explains the internal views on data sharing, noting that not everyone within the company or in other companies thinks the same way about this issue. "I lean towards transparency, but conceptually, the line is drawn at information that is highly competitive and could be used by competitors. We are all colleagues, but if you have a small advantage, you want to maintain it. If things are going poorly, not everyone needs to know that." A shared two-wheeler industry representative (IR2) confirms and indicates that it will happily share commercially sensitive data with municipalities but need to be careful for the competition getting too much data, saying: "Because you also need to understand that there is a lot of commercially sensitive information. Information that we would like to share with cities because it can help them in decision-making. At the same time, we do not want to give our competitors too much information. This could help them build a business case for a particular city where we are active." This indicates that shared mobility providers



are essentially not against data sharing but against competition gaining an competitive advantage using the data of others, and are therefore careful about what they share and are demanding transparency from municipalities.

5.2.3. DATA TYPE & CONTENT

To address the issue of translating policy problems to data demand, municipalities need to engage in dialogue with shared mobility providers and remain open to learning from them (MR7; IR4) Rosanne Klerx from the NTM (ER1) emphasizes the importance of specifying data requirements in permits: "In your permit issuance to a shared mobility provider, you need a paragraph or a few sentences specifying what data must be shared so that I, as a municipality, can have a certain insight." A collaborative approach in policy-making, as suggested by Dani Sprecher From MyWheels (IR4), involves engaging in non-binding discussions with providers. For instance, municipalities might ask: 'Can you tell us more about this?' We, as a city, want to achieve the goal of reducing the number of cars driving through this street. What could we do to achieve this?" This approach fosters a cooperative relationship where shared goals can be more easily achieved.

All the shared mobility providers are generally open to sharing data and see it as beneficial for policymaking. Dani Sprecher From MyWheels (IR4) underscores this willingness: "Municipalities need to realize that we are highly positive about sharing data with them. They shouldn't think that we are unwilling to do so. Sharing data can be incredibly useful, and we are eager to contribute to policymaking." A Shared two-wheeler industry representative (IR1) adds by stating: "We want to be the responsible operator and collaborate with cities."

Both municipalities and shared mobility providers indicate that specific data requested by municipalities can vary. However, the typical type of data shared with municipalities includes trip data and related information. It includes standard KPIs such as the number of trips, trip duration, and average distance (IR2). (Car-sharing) providers may also share additional data if municipalities present a compelling reason (MR7; IR3; IR4) Niels Wiersma from municipality Eindhoven (MR1) provides an example: "We asked them to share the time when a trip started in one area and the time when it ended in another area. This way, we could see how many trips started and ended in each area. However, we removed the direct relationships between origins and destinations to avoid privacy concerns." Shared two-wheeler industry representative (IR2) notes, "There are also more detailed questions, like the start and end points of each trip. Complaints are, of course, an important aspect. However, there is actually much more underlying data". A shared two-wheeler industry representative (IR1) explains the general approach to data sharing: "We share the start and end locations if the city requests it. Of course, we have legal agreements in place ensuring that this information is anonymized." Additionally, some providers share road quality data along with aggregated trip data (IR1). This indicates that there is a significant need for a balanced approach that respects mobility providers' business cases while ensuring municipal data needs fulfillment. This mutual understanding and collaboration could then lead to more sustainable and effective urban development and policymaking, resulting in better shared mobility solutions.

The content and detail of data shared differ between car-sharing providers and shared two-wheel providers. Both provide data on trips and fleet availability, including average trip distance, fleet numbers, start and end locations, vehicle locations, and trip counts. However, two-wheeler providers generally share more comprehensive data, excluding only privacy and business-sensitive information (IR1). As a shared two-wheeler industry representative (IR1) states, "Any information that's not related to user or commercial information that you could see on our app". In contrast, car-sharing


providers are more reserved. Jeanette van Eijk from Greenwheels (IR3) explains the cautious approach of car-sharing providers: "We currently do not know where our users go. We handle it very carefully."

However, car-sharing providers are often more willing to share data if there is a clear, long-term use case from the municipality (IR3; IR4). Jeanette van Eijk from Greenwheels (IR3) states, "But we always talk about averages, not, or rarely, about specific vehicles." Dani Sprecher From MyWheels (IR4) advocates for sharing aggregated data rather than specific details: "Aggregated areas instead of specific locations. So, not this car but this neighborhood. The larger the area, the better." This approach helps avoid micromanaging and ensures the data is used effectively for policy-making, according to Dani Sprecher From MyWheels (IR4). The variability in data-sharing practices also depends on how each municipality arranges data use and its contribution to decision-making as, a shared two-wheeler industry representative (IR2) points out, "Because a lot of data is shared, and often it's unclear how that data is used and how it contributes to decision-making."

5.2.4. DATA LEVEL

Municipalities receive data at varying levels of specificity from shared mobility providers (IR2; ER1. While some providers supply raw data, others offer only general insights and averages (MR1; MR7; ER1; IR3). Additionally, there is a distinction between dynamic data, often provided by shared two-wheel companies, and static data, typically provided by car-sharing companies (MR3; IR1; IR3) A policy officer from municipality The Hague (MR3) explains that data from shared scooter providers is dynamic and received almost in real-time, within about 24 hours, which contrasts with the more static and slower data from car-sharing providers.

A shared two-wheeler industry representative (IR1) indicates that municipalities might either receive raw API data to handle as they wish or use platforms that gather and display data from all operators in a territory via a dashboard (IR1). However, some municipalities request detailed data tracking of each vehicle individually, which can be problematic for some providers as noted by the car-sharing providers (IR3; IR4). Jeanette van Eijk from Greenwheels (IR3) notes, "*It is really just an average overview, which is different from what many municipalities now ask from us. They often want to track each vehicle individually. For that, I say, maybe in the future, but first we need to set it up properly.*" In contrast, a policy officer from municipality The Hague (MR3) notes that it can exactly see where shared tow-wheelers are standing, indicating that these companies do share the live locations of their vehicles. However, it is not noted that individual vehicles could also be tracked while in use.

Despite their efforts to provide data, shared mobility providers are often cautious about the details they share. Car-sharing providers, in particular, struggle with sharing detailed data due to privacy concerns. Dani Sprecher From MyWheels (IR4) emphasizes that the more detailed the data, the more privacy issues arise. Jeanette van Eijk from Greenwheels (IR3) recalls providing a detailed overview to the G5, the five largest cities in the Netherlands, highlighting the number of new vehicles and the social impact of their service, such as saved space, as a way to thank municipalities for their collaboration. However, she stresses the complexity of complying with requests for very specific data: *"Do you really want to know this explicitly per vehicle? Then we'll set it up properly, so I know I won't be the only one complying"* (IR3).

Privacy concerns and the risk of data leakage are significant barriers to data sharing. In smaller markets, data can be easily traced back to the provider, which is problematic (IR3; IR4). Jeanette van Eijk from Greenwheels (IR3) recounts an incident where their data was shared with researchers without consent, leading to a halt in data sharing. A shared two-wheeler industry representative (IR2) adds, *"If you know a few key data points, like the number of trips and their duration, you can almost*



calculate revenue figures. That's highly confidential information that we don't want to make public." However, municipalities often find that abstract data can be very informative, even without privacy-sensitive information (MR1; MR5; MR7). Niels Wiersma from municipality Eindhoven (MR1) explains that simple data, like the number of trips starting and stopping at each location per hour, can effectively meet policy goals (MR1). Gemma Schepers from municipality Amsterdam (MR7) adds, *"With a set of, I think, eight variables or even fewer, you can answer almost all your policy questions."*

Despite this, experts often state that data only delivers its value when it can be combined with other data (ER2; ER4; ER5). Ferdinand Burgersdijk from the European Commission (ER5) notes that a single piece of data doesn't say much on its own but becomes significant when placed in context and cross-referenced with other data. However, combining data can pose privacy risks. Gemma Schepers from municipality Amsterdam (MR7) expresses concerns about future data collection, emphasizing the need for thorough privacy assessments to prevent identifying individuals. In addition, Niels Wiersma from municipality Eindhoven (MR1) highlights the importance of Privacy Impact Assessments for any data processing involving personal data (MR1). To protect user privacy, providers say they ensure that no one can identify users from the data shared. For example, scooter ID numbers and start and end locations are hidden to prevent user identification (IR1). According to Dani Sprecher From MyWheels (IR4) linking shared data to other government-held information could potentially identify users, which is not allowed (IR4). In some cases, shared mobility providers therefore might use the GDPR as a shield in sharing data with municipalities. As indicated by CDS-M and TOMP-API expert Edwin van den Belt (ER4): "Often, the GDPR or AVG (Algemene Verordening Gegevensbescherming) is cited, saying, 'Guys, data sharing isn't fully possible because it's personal data.'"

All municipalities indicate that, in essence, receiving raw data is often preferred, but can be hindered by poor agreements, provider reluctance, or the municipality's ability to process and analyze the data. Roland van der Heijden from municipality Rotterdam (MR5) points out that without well-defined contractual agreements, underlying data might not be accessible when needed: *We don't have much visibility into all the underlying data that goes into generating that information. Sometimes things go wrong, and if you then need the underlying data and it's not contractually well-defined, you won't get that data.*"

5.2.5. DATA SHARING MODE

In data sharing between municipalities and shared mobility providers, three distinct forms have emerged: compulsory data sharing, voluntary data sharing, and a hybrid model. However, municipalities are increasingly pushing towards mandatory data sharing, particularly aiming to include car-sharing companies (MR3). Over recent years, the market has seen a shift towards regulated data sharing through licenses or tenders. As a shared two-wheeler industry representative (IR1) explains, *"In these licenses and tenders, there are increasing requirements to share data in a specific way. This shows us how the city wants it. We then discuss with the city their preferences and the standards they want. It's an ongoing discussion that we can adjust anytime." A policy officer from municipality The Hague (MR3) observes this move towards a more mandatory approach, noting the necessity of shared mobility in the urban mobility transition: <i>"We need a solution. As a municipality, you can essentially say that if you don't provide data, we can revoke your permit. So, you must provide some data."*

Mandatory data sharing is common for two-wheel sharing companies, which must share data as part of their operating permits (MR3; MR6; IR1; IR2). The extent and nature of data shared can vary significantly. As Jeanette van Eijk from Greenwheels (IR3) points out, *"The two-wheelers needed*



permits, so they had to comply. It was explicitly stated in certain permit conditions." This gets confirmed by a shared two-wheeler industry representative (IR2) stating: "Primarily, we share data in municipalities where we are active. This has become a standard part of our permits with municipalities." In contrast, car-sharing companies are less subject to mandatory data sharing. They tend to be more resistant to such requirements, as Gemma Schepers from municipality Amsterdam (MR7) notes: "The car-sharing companies find it very difficult. We've been in discussions with them for about a year and a half or two now, and they are very resistant." Roland van der Heijden from municipality Rotterdam (MR5) adds that traditional companies are worried about compromising competitive advantage when sharing data: "Companies in traditional sectors are often hesitant to adopt new digital practices because they are concerned about data security, privacy and loss of control over their data. They may be reluctant to share data unless they are sure that it will not compromise their competitive advantage".

Voluntary data sharing is a phenomenon affecting both two-wheel and car-sharing companies. Carsharing providers, particularly those affiliated with large car brands, are often more cautious about data sharing and less keen on mandatory data sharing. Gemma Schepers from municipality Amsterdam (MR7) explains, "Some car-sharing companies are part of large car brands, which find data sharing more intimidating. In contrast, the shared scooter companies, which often start as small start-ups, have no problem sharing data." Dani Sprecher From MyWheels (IR4) elaborates on the differing attitudes towards data sharing: "The shared scooter companies come to the city with something new and ask for permission to operate. In contrast, car companies have been around for a long time and have been operating under certain regulations for years. Suddenly, they are asked to open up their databases." Car-sharing companies prefer to share data voluntarily and want to contribute to municipal goals and development plans in this manner. Dani Sprecher From MyWheels (IR4) emphasizes the importance of constructive dialogue over obligations: "You can still engage in policy-making by discussing and requesting data in a non-legally binding manner, like asking, 'Can you tell us more about this? We want to achieve the goal of reducing the number of cars on this street. What could we do to accomplish that?'''

A hybrid approach, involving a combination of mandatory and voluntary data sharing, is seen as beneficial by car-sharing providers (IR4). Dani Sprecher From MyWheels (IR4) suggests that municipalities set a minimum mandatory requirement for essential data while fostering collaboration for additional data sharing: *"It would help if municipalities saw mandatory requirements as a basic foundation and approached the rest through constructive collaboration."*

Both car-sharing and two-wheel companies can engage in proactive data sharing or respond to municipality's data demands. However, Jeanette van Eijk from Greenwheels (IR3) states, *"The only reason we would share data is because a municipality requests it from us."* Meanwhile, a shared two-wheeler industry representative (IR2) notes the benefits of proactive data sharing: *"We also proactively share data in areas where we believe we have a mutual interest and where it can improve our services."* These attitudes align with the cautious nature of older companies and their leadership styles. As indicated by a policy officer from the municipality of The Hague (MR3), shared two-wheel providers are generally led by younger people who are more technically capable and collaborative by nature. Speculatively this could also be explained as shared two-wheel companies might benefit more from the municipalities' idealistic changes in infrastructure aimed at reducing cars in public spaces and promoting sustainable mobility, therefore aiming at mutual benefits. This allows them to directly benefit from data sharing, which can be used to make the infrastructure less car-dependent.



5.2.5.1. DATA SHARING MOTIVATION

Following proactive data sharing, shared mobility (two-wheeler) providers see direct benefits in sharing data, often aligning with common interests such as service improvements and infrastructure enhancements (IR1; IR2). Shared mobility providers emphasize the proactive sharing of data to improve services and address gaps in supply (IR2). They also share data on missed demand, which highlights areas where potential customers open the app but do not find available vehicles, leading to actionable insights for better service placement. For example, "In the station area, we are not allowed to offer bikes, but that's where the highest demand for bikes is. This insight has helped us get permission in some municipalities to offer our bikes in specific areas" (IR2).

Shared mobility providers acknowledge their use of municipal infrastructure and feel a responsibility to give back to the municipalities. Jeanette van Eijk from Greenwheels (IR3) mentions, "We should be grateful for the opportunity to operate here, and we should be willing to give something back in return." By sharing data, both municipalities and mobility providers can gain insights that lead to infrastructure improvements, expansions, or changes (IR1; IR2; ER2). A shared two-wheeler industry representative (IR1) notes, "We want to help cities build more bike lanes, make it easier for people to bike, take a scooter, walk, etc." The data shared shows aggregated travel patterns, which can inform infrastructure development decisions. For instance, "We can say, 'Look, many people bike through this district every day, but there's a large lane for cars and no bike lane. Maybe in a few years, you should consider building new infrastructure there'" (IR1).

Despite the potential monetary value of the data they collect, shared mobility providers convincingly state they do not sell it for profit (IR1; IR3). A shared two-wheeler industry representative (IR1) stresses that there is no financial motivation behind it: *"We don't monetize our data"* (IR1).

5.2.6. DEGREE OF ACCESS

As essentially every stakeholder indicates, the extent to which municipalities have access to specific data varies greatly depending on the case. In some instances, municipalities obtain all requested data, including raw and real-time data. In other cases, the data is first processed or shared with a third party (see next paragraph).

A policy officer from municipality Rotterdam (MR6) explains that while for shared mobility providers some data is easy to share, other data can be highly competitive and privacy-intrusive. Gemma Schepers from municipality Amsterdam (MR7) adds that municipalities can receive data dumps, allowing them to process and analyze the data themselves. However, to access this data, municipalities must have a clear policy question and legal justification (MR7; IR4). Sometimes, additional information not included in API data variables could be requested separately and received in Excel files (MR7).

Municipalities often believe they own data generated within their boundaries, while companies see themselves as the rightful owners (MR5). Users also believe that data generated by individuals should belong to them, which makes the question of data ownership complex. As Gemma Schepers from municipality Amsterdam (MR7) notes, "Whose data is it? Sometimes it belongs to the government because it oversees public spaces. Sometimes it belongs to the provider. Sometimes it belongs to the owner." Ferdinand Burgersdijk from the European Commission (ER5) argues that data generated by users, such as bike usage data, should belong to the user. He criticizes the idea of bike-sharing companies sharing this data with municipalities without user consent, stating, "What I do with that bike is none of their business." Extending on this, municipalities have faced resistance from companies



over data ownership. Roland van der Heijden from municipality Rotterdam (MR5) recalls an incident where a company asserted ownership over the data, saying, "Yes, that's great, but the data isn't yours; it's ours".

The degree of data access also depends on agreements between mobility providers and municipalities and the specific data (MR1; MR7; IR2; IR3; ER5). Dani Sprecher From MyWheels (IR4) emphasizes that legal obligations must be met by mobility providers, and municipalities must respect data demand limits. Disagreements can lead to legal disputes. However, non-obligatory data demand and learning from shared mobility providers through constructive dialogue also form an alternative according to Dani Sprecher From MyWheels (IR4). Ferdinand Burgersdijk from the European Commission (ER5) points out that European and Dutch legislation requires municipalities to regulate data requests within their legal frameworks. Dani Sprecher From MyWheels (IR4) recounts a situation where a municipality demanded all data for cars in public spaces, leading to a potential court case: *"Your cars are in public spaces, so you must share all your data.' That's how he thought about it. I responded, 'Okay, then we'll see each other in court.'"*

Data sharing and access are rarely straightforward. Roland van der Heijden from municipality Rotterdam (MR5) notes that it's almost always about the circumstances under which data can be shared (MR5). Building trustful relationships is crucial for effective data sharing. Martijn Arets from Professional Outsider Consultancy (ER3) emphasizes the importance of being mutually reliable partners in making agreements, and Rosanne Klerx from the NTM (ER1) adds that trust is essential before exchanging data.

Sharing data is a big task for many market participants (Royal HaskoningDHV, 2017). They see data as an asset that (potentially) provides a competitive advantage. As a result, they want to limit access to it significantly and keep it mostly for themselves (IR2; IR3; IR4). However, when the data's purposes and usage are clear, transparent, and secure, shared (car) mobility providers are more willing to share (some) data for public benefits (ER5; IR3; IR4).

5.2.7. FACILITATION

When municipalities lack the technical capability to read or collect data themselves, they often turn to third parties for assistance. They use data dashboards provided by organizations like Vianova and CROW. This approach has proven effective for shared bicycles but remains challenging for carsharing (MR3; MR6; MR7). "The CROW shared mobility dashboard is super useful for us as a *municipality*," notes a policy officer from municipality The Hague (MR3). These dashboards help municipalities analyze data effectively, especially when responding to council questions, as highlighted by a policy officer from municipality Rotterdam (MR6): "Often, specific data is used in an ad-hoc manner. For example, when there are council questions, you can use the shared mobility dashboard, which really proves its value." Gemma Schepers from municipality Amsterdam (MR7) explains, "We are not technically capable of accessing that data ourselves, so we work with third parties like Vianova or CROW. They have the APIs to access the data and display it on a dashboard, giving us access." The goal is to integrate shared mobility data into these dashboards. While this integration works well for shared two-wheelers, it remains a challenge for car-sharing. "For shared two-wheelers, everything is included and functional. However, for car-sharing, it is still quite a challenge," states a policy officer from municipality Rotterdam (MR6). There is a desire to include car-sharing data in these dashboards, as noted by a policy officer from municipality The Hague (MR3): "Ideally, we would like to have a dashboard that includes the car-sharing providers as well."



For data dashboards to be effective, they need a sufficient data supply to provide value and reduce concerns about data tracing back to individual providers (MR5; IR4). Public and essentially open data platforms can act as barriers to data sharing. Roland van der Heijden from municipality Rotterdam (MR5) explains, "An open data platform uses open standards, and anyone can connect. However, it doesn't mean all data is open. Companies decide under what conditions someone gets access to their data." When data is open, it is freely accessible and therefore a concern for private companies because of their business case (Geonovum & Ministerie van Binnenlandse Zaken en Koninkrijksrelaties, 2021).

Car-sharing providers indicate their hesitance to supply data due to concerns about data handling and competition. Dani Sprecher From MyWheels (IR4) notes, "*The relationship that Vianova currently has with the municipality of Utrecht and the data from the providers is not okay. They are given too much freedom with it.*" He also highlights the competitive risk: "*Greenwheels knows its data, so if they pull their data from the dashboard, they also get ours. You have to be realistic about that.*" To mitigate these issues, some mobility providers offer their own dashboards and analyses (IR2; IR4). However, not all municipalities trust the reliability of these proprietary surveys and data presentations. Jeanette van Eijk from Greenwheels (IR3) mentions, "We also conduct research with our own users. *This type of information is included in the research, but some people don't fully believe it.*"

5.3. DIGITALIZATION AND DATA SHARING DEVELOPMENTS AND OPPORTUNITIES IN THE NETHERLANDS

5.3.1. LEGAL AGREEMENTS AND TRUST DEVELOPMENT

Municipalities embrace new mobility solutions, but mutual agreements should be made (Gemeente Den Haag, 2020b). For municipalities and shared mobility providers to have a mutual understanding of expectations, establishing clear conditions is essential. These conditions must address what data will be shared, how privacy and security are managed, and who owns the data, ensuring that consent is always obtained (Maltha et al., 2021). From a municipal perspective, flexibility in current policies is beneficial for implementing innovations. Niels Wiersma from municipality Eindhoven (MR1) notes the advantage of having an experimental clause in the parking ordinance that allowed for flexible data-sharing agreements. Martijn Arets from Professional Outsider Consultancy (ER3) adds that agreements should include buffers for changes to manage risks effectively.

Jeanette van Eijk from Greenwheels (IR3) highlights the need for clarity in governance and liability: "How is cybersecurity managed? Also, how is governance arranged? When I share data, who owns it? And if something goes wrong, who is liable? These are things that are still not clearly and precisely regulated." This illustrates the current state of data sharing and the need for further development in regulation and standardization to increase trust within data sharing and handling. Jeanette van Eijk from Greenwheels (IR3) explains that harmonizing and understanding data is acceptable as long as it is properly set up and standardized. This ensures that data sharing is less risky and more manageable.

Both municipalities and shared mobility providers must take considered risks and agree to changes which takes trust, "*Ultimately, it is about being a mutually reliable partner,*" notes Martijn Arets from Professional Outsider Consultancy (ER3). Gemma Schepers from municipality Amsterdam (MR7) emphasizes the need for municipalities to be consistent and reliable partners, citing the example of Amsterdam's inconsistency with scooter permits, where certain partners were banned from operating within the city even though they complied with every demand from the municipality, compromising trust. Rosanne Klerx from the NTM (ER1) underscores the necessity of informing each other about



changes to maintain this trust, stating: "Parties are realizing more and more that they can't do everything themselves. I don't know everything, and I don't have all the data, so I gladly use your data. But I want to know that it's reliable and that you won't change it without notice. You'll see that trust will play a bigger role in this. This can be addressed by making mutual agreements. Under what conditions and how will we handle changes?" Transparency in data sharing and utilization lowers barriers and fosters collaboration for shared (car) mobility providers. Jeanette van Eijk from Greenwheels (IR3) states, "If you keep it transparent and clearly specify what you want from whom, it becomes a solid foundation to build on. And yes, trust is the key in this." Ferdinand Burgersdijk from the European Commission (ER5) adds that compliance with laws and regulations, combined with transparent operations, ensures trustworthiness.

Understanding each other's interests and expectations is fundamental. Jeanette van Eijk from Greenwheels (IR3) asks municipalities, *"Why do you want this data? And why should I share it?"* highlighting the need for clarity and mutual understanding. Roland van der Heijden from municipality Rotterdam (MR5) warns that the lack of transparency and accountability is a significant concern. Gemma Schepers from municipality Amsterdam (MR7) concludes that careful consideration is needed when sharing data via technical APIs, including defining roles and responsibilities clearly to remain trustworthy.

5.3.1.1. DIGITAL WALLET

Starting in 2025, a European digital wallet will be introduced for citizens, enhancing transparency in data sharing and ownership. This initiative aims to help users better understand what data they are sharing and ensure that data sharing is limited to clear, justified use cases (ER4). Gemma Schepers from municipality Amsterdam (MR7) highlights that the digital wallet system will improve transparency, allowing users to specify which data they wish to share. The digital wallet will enable minimal identification and allow citizens to manage the information they share. CDS-M and TOMP-API expert Edwin van den Belt (ER4) explains, "For example, if you want to rent a scooter in Rome, you can use your Digital Wallet to easily share your name and bank account details with the rental organization to sign up and use the scooter." This system simplifies the data-sharing process while maintaining user control. Companies will still be able to set conditions for using their products, but these must be legally justified. CDS-M and TOMP-API expert Edwin van den Belt (ER4) emphasizes, "They may only request information necessary for their operation, such as verifying the validity of a driver's license." However, since data ownership is expected to be put at the individual user level, data acquisition from a municipal perspective might be compromised. Municipalities might need to get data from way too many individual digital wallets. Therefore there might arise opportunities for data collection initiatives where individuals directly allow municipalities to handle their data and therefore supplement municipal data demand (ER4).

5.3.2. DIGITAL CAPACITIES & INFRASTRUCTURE

The national and local government acknowledges that digital skills are becoming increasingly important for businesses and urban projects (Rijksoverheid, 2021b). The municipality of Rotterdam acknowledges the challenges of digitizing existing departments and modifying internal processes, while new departments can be easily adapted and set up in a non-traditional way, aiming to work data-driven and encourage integration with other departments and stakeholders (de Lange, 2020).



5.3.2.1. VNG COMMON GROUND INITIATIVE

The adoption and implementation of IT systems within municipalities are exceptionally slow, and meeting privacy legislation requirements is highly challenging. A critical aspect of this issue is the complexity of combining and sharing data across different domains. Since each IT system is often slightly different from the next, and data standards are not yet widely adopted, IT is frequently seen as a barrier rather than an enabler (Keur, 2020). A policy officer from municipality Rotterdam (MR6) describes the situation vividly: *"ICT systems are almost like tectonic plates that continuously rub against each other. If one decides to slide open, then another one goes in the opposite direction. That's very complicated."* This analogy highlights the lack of synchronization between various IT systems, making effective data sharing and implementation of IT solutions particularly difficult.

Therefore, the Common Ground (CG) vision was developed. Through CG, municipalities can handle data easier, faster, and smarter and comply with privacy legislation. CG is a movement in which municipalities work together with the VNG, chain partners, market parties, and the state (VNG, 2020). The goal is to discover and organize a new way of working (using ICT and data). Despite a strong willingness among municipalities to realize CG, breaking away from the status quo is difficult (Keur, 2020). Antoine Gribnau from municipality The Hague (MR4) notes that the theoretical promises do not always match practical experiences: *"Theoretically, I completely agree, but that just means a lot of agreements have to be made with all sorts of parties."*

The VNG roadmap clarifies how municipalities can achieve collaborations to realize decision-making and developments (Keur, 2020). Most large municipalities are actively participating in the CG initiative, intending to convince smaller parties and municipalities to join through practical success stories, with the VNG indicating opportunities for innovation (VNG, 2020). The technology to achieve all of this is already available, however, the significant challenge lies in establishing the right collaborations with chain partners and market parties (Keur, 2020). Additionally, Antoine Gribnau from municipality The Hague (MR4) points out that the usual problem-solving culture involves short-term thinking: *"My strategy is to focus on the short term. What do my clients want? I will arrange that for them. Then we will see if we can integrate that into a system."* This contrasts with what car-sharing providers believe is necessary for shared mobility data sharing, which is a long-term perspective (IR4). In this respect, Common Ground is essential to extend the internal short-term problem-solving culture within municipalities and reduce the individual demand on shared mobility providers, as data can be better shared within and among municipalities (VNG, 2020).

5.3.3. STANDARDIZED DATA SHARING PROCEDURE

Currently, many municipalities do not know how to handle data responsibly and often improvise, which can lead them to not comply with EU law and regulation (Berretta, 2023; European Commission, 2023c). This uncertainty makes shared mobility providers reluctant to share data. Dani Sprecher From MyWheels (IR4) highlights this issue, noting that misinterpretation of data can lead to incorrect policy decisions: *"If they just request a lot of data and start looking at it, yes, it's going to go wrong."* Different interpretations of data values are common, and proper data interpretation requires understanding the context in which the data was collected. Jeanette van Eijk from Greenwheels (IR3) argues that data generation and analysis are best performed by the same party or at least using the same standards. A standardized data protocol is a solution to facilitate interoperability between the different parties and solve the current problems (Borsboom-Van Beurden et al., 2021).

Jeanette van Eijk from Greenwheels (IR3) illustrates the need for this standardization: "If an organization wants to analyze all the offerings in a city and says, 'I get one sheet from Greenwheels



and another from a different provider, and I can't combine them. I want to harmonize and understand the data,' I understand that but we need to set it up properly and ensure that we're all doing it in the same way." Inconsistent definitions and calculations further complicate data sharing. Jeanette van Eijk from Greenwheels (IR3) points out that terms like occupancy rate can be defined differently by various providers, affecting how data is interpreted: "How do you calculate the occupancy rate of a car? That very much depends on how you set it up in your system."

In the view of Dani Sprecher From MyWheels (IR4) officials often oversimplify data sharing without understanding its complexities. According to Jeanette van Eijk from Greenwheels (IR3, governments are also often unaware of how private mobility providers operate. Jeanette van Eijk from Greenwheels (IR3) recalls finding basic data requirements in local ordinances, such as respecting user privacy, which providers already prioritize: "*I have to laugh a little bit at that because I look at that and think; yes, but we already have much more respect for our users in the first place than you do right now.*" This illustrates the critical view form car-sharing providers on municipalities and the distrust in them handling data correctly, which hampers their data sharing.

Conversely, shared mobility providers can also learn from the municipality's perspective. Gemma Schepers from municipality Amsterdam (MR7) stresses the importance of mutual learning and collaboration: "If I talk to Check, for example, they tell me things I never thought about because I'm not in their business. And when I talk to them, I provide the city's perspective, which they might not know. Continuous consultation is very important." A policy officer from municipality The Hague (MR3), confirms and recognizes that visions from younger people (among shared mobility providers) can enhance municipal knowledge, "You notice that these shared scooter companies were often, generally speaking, started by young guys. They also had a lot of knowledge about how to do all of this technically very quickly. So, that's perfect for us as a municipality."

5.3.3.1. CDS-M

To integrate shared mobility into urban development the Netherlands is developing a national program called 'Natuurlijk Deelmobiliteit'. A key component of this program is the CDS-M, a standardized data-sharing method (MR7). "We have ensured that data sharing is standardized in the Netherlands through the 'Natuurlijk Deelmobiliteit' program. As part of this standardization, we have also specified technical APIs," explains Gemma Schepers from municipality Amsterdam (MR7). The program is a collaborative effort, promoting the use of shared transportation and CDS-M (Curzon-Butler, 2022; ER4). Using European grant to share mobility data, the CDS-M was initiated in 2019 (Berretta, 2023; European Commission, 2023c). The TO-MP working group, Ministry of I&W, and the five largest Dutch cities are currently collaborating to further develop this initiative (Nijhof, 2020). CDS-M is a data-sharing manual that enables municipalities to analyze data responsibly while safeguarding user privacy (Berretta, 2023; European Commission, 2023c). The CDS-M program recognizes that shared mobility, partly driven by growing sustainability movements, is increasingly being adopted and utilized in cities (Berretta, 2023; European Commission, 2023c). Governments benefit from CDS-M by gaining better insights from shared mobility in the urban environment, allowing for better policy development, urban development, and identifying social problems (Gemeente Amsterdam, 2024). The model is based on GDPR and includes a step-by-step explanation of how municipalities can learn from use cases (Berretta, 2023; European Commission, 2023c).

Municipalities start by identifying a policy problem, referred to as a use case in CDS-M terms. These use cases are often set by policymakers in consultation with shared mobility providers (Beckers & Aurup Poolsem, 2022). "*The goals are actually set by policymakers in consultation with the shared mobility providers*" says Gemma Schepers from municipality Amsterdam (MR7). Data requests must



be substantiated with a clear use case, as unsubstantiated data requests are prohibited. "The GDPR and AVG legislation prescribe data minimization. If you start exchanging trip data without a substantiated question, that's not allowed," explains CDS-M and TOMP-API expert Edwin van den Belt (ER4). Ultimately, this creates a centralized and unified data query that meets all privacy requirements, which makes for easier collaboration between governments and mobility providers (Gemeente Amsterdam, 2024). Gemma Schepers from municipality Amsterdam (MR7) notes, "What's very nice about that is that it's very easy for cities because they don't have to think about how to share data anymore."

"CDS-M's goal was not to develop a separate standard, but to use existing standards to answer policy questions in a privacy-compliant way," says Niels Wiersma from municipality Eindhoven (MR1), noting that it is an improvement to establish a data link between carriers and cities. This standardization is welcomed by both municipalities and mobility providers, facilitating fair and structured data exchange (MR1; MR3; MR7; ER4; IR3; IR4). "For the shared mobility providers, it is very nice because they have to install an API once, and then they are done. The basic data is always just unlocked via an API," explains Gemma Schepers from municipality Amsterdam (MR7). Jeanette van Eijk from Greenwheels (IR3) emphasizes the importance of setting up this data sharing correctly: "I understand that the municipality wants to know what happens here. We are using public space after all. But we need to set it up right. So we are really working with CDS-M to harmonize data sharing and arrange it as best we can."

Most of the current work regarding CDS-M focuses on car-sharing companies, but efforts are expanding to include two-wheelers. *"We are now working to include the two-wheelers as well. For that, we are at the table with Amsterdam, Rotterdam, and Eindhoven,"* says CDS-M and TOMP-API expert Edwin van den Belt (ER4). The CDS-M procedure has been tested in 2021 through several pilots in which the use cases have been worked out. However, laws and regulations still have to be adapted to the CDS-M procedure and the standards within the mobility sector. To this end, a plan of action is being developed for the Netherlands, as well as ambitions have been put out for international cooperation and implementation (Beckers & Aurup Poolsem, 2022).

5.3.4. DATA STANDARD

Standardized data sharing within the CDS-M program is facilitated through an API, ensuring interoperability between mobility providers, municipalities, and other stakeholders. As is illustrated in Figure 4 and explained by a policy officer from municipality The Hague (MR3): *"With data sharing between those parties, that magic word of interoperability always comes up. That means communication between technical systems. So that can be helped with an API."*



Figure 4 Display of API function (Gemeente Amsterdam, 2021a)

Niels Wiersma from municipality Eindhoven (MR1) discusses the evolution of data standards: "The first standard that we have tested quite thoroughly and that is reasonably used is GBFS. It's an old bike-sharing standard that only reports status changes, resulting in inaccurate start and end times and



making it difficult to track the availability of bikes and such. Then we adopted MDS, which was originally an American standard. It provides a much more continuous data stream." A policy officer from municipality The Hague (MR3) confirms, "In the beginning, there were various APIs in the MaaS world. We started with GBFS, then moved to GBFS+, and finally to the TOMP API." The TOMP API, imposed after negative experiences with shared bicycles in Dutch cities (MR3), is an open-source code that facilitates data exchange among all parties (Nijhof, 2020). Multiple data standards are necessary to address specific aspects and fulfill various purposes. Rosanne Klerx from the NTM (ER1) emphasizes, "I have come to realize that having multiple standards, each addressing specific aspects, is essential. Otherwise, you end up with a single, overly broad standard that does not adequately meet the local and individual needs of parties, users, or governments."

Integrating data standards can be challenging due to technical issues and legal obligations regarding data standardization and sharing. For example, U.S. data standards required adjustments to comply with European regulations. CDS-M and TOMP-API expert Edwin van den Belt (ER4) recalls, "We did push back against OMF for a while because it wasn't fully GDPR compliant. We looked into it thoroughly and made various adjustments for the European market."

Shared mobility providers are technically capable of making necessary adjustments to comply with these standards. However, they require clarity about the intended outcomes and the course of action, as these changes can be costly (IR3; IR4). Jeanette van Eijk from Greenwheels (IR3) notes, "*Technically, a lot is possible, but many other aspects still need to be properly arranged.*" Dani Sprecher From MyWheels (IR4) adds, "*We did have a TOMP-API, which we developed a long time ago. We acquired Amber Mobility, a much younger and newer car-sharing company that had developed such a TOMP-API but it also cost a lot of money.*"

5.3.5. NEUTRAL DATA FACILITATOR

Governments are looking for opportunities to acquire all kinds of data, but are limited by their capacities to form collaborations and guarantee data quality (MR4). Additionally, new laws and regulations need to be incorporated into the CDS-M procedure, and cooperation in the domain of mobility must be expanded. For this, besides frontrunners, a central governing body is also necessary to monitor and encourage data exchange (Beckers & Aurup Poolsem, 2022). Also, a comprehensive solution is needed to address concerns among private parties about competition, privacy, and mistrust when data is made public. Neutral parties must develop solutions and establish pre-set regulations to alleviate these concerns and facilitate data sharing and collaboration (ER1).

Governmental bodies can potentially serve as neutral actors in managing shared data because they typically have no vested interest in private companies. Rik Braams from TNO and the Ministry of I&W (ER2) suggests, *"A government could do that because a government has no interest in a private company. You would say a government is a neutral actor."* However, it is essential to recognize that even governmental bodies can be influenced by political streams and their effects (ER2).

Although a neutral data facilitator is not interested in the individuals behind the data but solely in the transport movements and associated data, a major impediment for a mobility data hub is the GDPR, which ensures the privacy of personal data. Therefore, an adjustment to the rules on purpose limitation is necessary. Both government and market parties should be able to share data and be a source of innovation (Geonovum & Ministerie van Binnenlandse Zaken en Koninkrijksrelaties, 2021).



5.3.5.1. NATIONAL ACCESS POINT (NAP)

Within Europe, every member state must have a National Access Point (NAP) for mobility data. In the Netherlands, this is the National Access Point for Mobility Data (NTM) (ER1). It serves as a connecting hub between various parties in shared mobility data sharing. It establishes and ensures standardization and interoperability in data sharing and brings public and private entities closer together (NTM, 2023a). The NAP's core mission lies in making mobility data discoverable and linking different data together to form a chain connection. It recognizes data can provide a lot of value, but is also associated with risks. Therefore, good cooperation between all stakeholders is essential (ER1). Additionally, Gemma Schepers from municipality Amsterdam (MR7) notes the multiple functions of the NTM as follows: *"The NTM is also going to determine, throughout the data chain, who has what role and what responsibility? So who is the owner, who is the controller, who invents it, who tests it if you want to exchange data? Because that is also not yet sufficiently described."*

The ultimate goal is to have the NAPs of each member state working together to create an international platform that facilitates international transportation (ER1). Data must not only be known but also accessible, functional, reliable, and interoperable, which is why NTM focuses on various MaaS standards such as TOMP-API and CDS-M data sharing procedure (NTM, 2023b). As Rosanne Klerx from the NTM (ER1) explains, data must be shared in a manner that transforms it into useful information and knowledge. Data must be accessible so that not only the transporter but also the municipality that issued the permit can see it. NTM focuses on the data aspect, while the dashboards communicates the knowledge derived from it. Thus, several steps are involved in this process (ER1).

5.4. CHALLENGES AND OPPORTUNITIES IN UTILIZATION OF AND LEVERAGING DATA IN URBAN PLANNING AND POLICYMAKING

5.4.1. SILO THINKING AND INDIVIDUAL EXPECTATIONS

Urban planning is complicated by shrinking available space and differing interests within public space as indicated by most municipalities. Municipalities and shared mobility providers both indicate they must reach concessions in expectation management to share data and develop public space effectively. Municipalities should seek overlapping goals among municipalities, mobility providers, and citizens (ER5). A policy officer from municipality Rotterdam (MR6) notes, "*The public space and the spatial integration of all ambitions we have as a city is becoming an increasingly important and difficult question. There are so many ambitions, and everyone is on their own island, but how does it all come together?*" This issue is compounded by mismatched expectations between municipalities and providers. "*There is continuously a mismatch between the expectations we have from a policy perspective and what providers can or want to deliver*," says a policy officer from municipality Rotterdam (MR6). Ferdinand Burgersdijk from the European Commission (ER5) describes the relationship as a triangle in which an overlap of common interests should be identified between government, private companies, and citizens.

In addition, there is often too little collaboration within municipalities. Operational departments are too individualistic, making it difficult to achieve cohesive results. *"It is already difficult enough for many municipalities to get collaborations done within their own organization,"* observes Antoine Gribnau from municipality The Hague (MR4). Technical systems may differ from one department to another, making it more difficult for employees to adjust (MR4; MR6). In addition, data sharing between municipalities is also still perceived as difficult, despite promising great potential (MR4). For municipalities to make a successful decision, it is incredibly important to drive together from the same reality. Different views and results should be examined to understand how they came about (MR5).



Mobility is too often approached from a silo, resulting in limited interdisciplinary collaboration (ER3; MR5). To create unity and promote cooperation, direction, and control from a higher level of government are needed to bridge gaps between different municipalities (ER1). *"What strikes me is that people often think in isolated themes,"* notes Roland van der Heijden from municipality Rotterdam (MR5). *"However, as a government, we often buy data and applications from companies, which end up as silo solutions within our organization,"* he adds.

This silo thinking creates non-optimal results, as external scenarios are not included. Strengthening intra-municipal cooperation creates opportunities for data (models) to interact and provide closer-to-reality insights (MR5). Roland van der Heijden from municipality Rotterdam (MR5) finds this particularly interesting but challenging: *"In Rotterdam, we have a fantastic mobility model that calculates road capacity and flow rates. We also have a great water model that shows flood scenarios. However, integrating these models to understand road capacity during a flood is very difficult because everyone works in silos."* Currently each tool is only focused on one goal and needs only limited data to do so. This data must be combined for both tools to get the same result and thus better present reality (MR5). This illustrates the need for all kinds of data including shared mobility data to create a holistic and integral simulation of reality. However, these intra-municipal collaborations are limited by differences in technical state and infrastructure, leading to problems regarding interoperability and human knowledge. A policy officer from municipality Rotterdam (MR6) reflects on these silo approaches: *"Everything changes so often. We have experiments with mapping structures. The mobility department still uses them, while the sustainability department operates through Teams and SharePoint."*

5.4.2. MUNICIPAL KNOWLEDGE AND CAPABILITIES

To work with data and collaborate with others, urban planners and policymakers must be (digitally) proficient and have the right facilities to analyze and process data into policy and urban development decisions (Maltha et al., 2021). Staff must be knowledgeable about processing as well as gathering data to deal with the complexities and deliver value within the public sector (Rijksoverheid, 2019b). In practice, most civil servants have too little knowledge and/or (technical) capacity to accurately analyze data. In addition, they need to be able to translate this analysis into policy and decision-making, which also requires a lot of knowledge, capabilities, and capacities (Maltha et al., 2021). Most civil servants do not have these capacities as Gemma Schepers from municipality Amsterdam (MR7) observes, "*The average official doesn't understand what to do. And then you get that every civil servant is going to make up his own mind and that's not very efficient.*" Dani Sprecher From MyWheels (IR4) adds his view on policymaking when civil servants are not (digitally) capable of making decisions, "*If you draw the wrong conclusion, you are also going to make the wrong policy.*"

A shared two-wheel providers (IR2) mentions that while a lot of data is shared with municipalities, the main challenge is the municipalities' capacity to use that data effectively for decision-making. Dani Sprecher From MyWheels (IR4) further explains, "*Certainly, the substantial knowledge is still lacking to work with the data. Even if the data analysis is done, translating that into policy and making decisions is something most civil servants cannot do.*" This illustrates the critical shared mobility providers' perspective on municipalities and sharing data with them, suggesting data provision is not the essence of the problem, but being able to cope with them is.

When municipalities lack the knowledge and capacities, they seek it from external parties. Larger municipalities often have more internal knowledge and capacities, while smaller municipalities are highly dependent on consultants (ER2). Rik Braams from TNO and the Ministry of I&W (ER2) notes



that smaller municipalities and regions often struggle with knowledge building, contrasting with larger teams like Amsterdam. However, Gemma Schepers from municipality Amsterdam (MR7) indicates that they also often do not have the capacity and capabilities to work with certain data, "We often can't do it ourselves, so we hire agencies. For example, we use third parties like Vianova or CROW to handle data through APIs and provide dashboards for us to access the data." Gemma Schepers from municipality Amsterdam (MR7) continues, "I think a city like Eindhoven or Rotterdam are better at this because they are more technical." "A lot of expertise is hired by governments, and it leaves with the consultants, risking the buildup of internal knowledge," Rik Braams from TNO and the Ministry of I&W (ER2) concludes.

In addition, the lack of expertise and knowledge can also lead to incorrect handling and interpretation of data. *It's really about how many people are on your digital team and whether those people understand data and have time to dive into it,* " explains a shared two-wheeler industry representative (IR2). Gemma Schepers from municipality Amsterdam (MR7) agrees in this, "*We can't do it as a municipality because we're not technically capable of unlocking the data.*" To address these challenges, the Natuurlijk Deelmobiliteit/CDS-M program is being set up in the Netherlands (MR7).

5.4.3. DATA STANDARDS AND INTEROPERABILITY

If as many municipalities as possible draw up a common vision and ground rules for the digital infrastructure, such as the use of IoT and data, innovation and developments can be safer and faster, as cities can learn from each other's experiences (Depla & Ollongren, 2017). In addition, an integral vision for the design of public space, in which cooperation with the state, municipalities, the business community, and (knowledge) institutions is central, is also a way to develop further (Depla & Ollongren, 2017).

To work together, central data standards must be developed and integrated into business operations and municipal infrastructure. In doing so, it is important that data can be read by the recipient and used for policy development (Royal HaskoningDHV, 2017). A policy officer from municipality The Hague (MR3) acknowledges the challenges in adopting a central data standard: *"There are still major challenges in getting those inter parties to work together, especially within a central platform where they are forced to adopt the same data standards. That makes the private parties nervous."* Additionally, Roland van der Heijden from municipality Rotterdam (MR5) highlights a practical issue with proprietary data formats: *"At one point, we had the data, but I couldn't do anything with it because it was in a language only that one company understood. It wasn't an open standard; it was written in a company-specific standard. I needed the codebook, which I could only get if I paid for it."* This lack of data standards, open data formats, and protocols creates barriers to city development (van Beurden, 2017).

5.4.4. DATA RESPONSIBILITY AND PRIVACY & SECURITY

"Even though there is no law prohibiting a certain action, it doesn't mean it should be done without careful consideration," says Roland van der Heijden from municipality Rotterdam (MR5). He further emphasizes the importance of being responsible, prudent, and transparent in decision-making: "We must always consider whether it is responsible, good, and smart, and we should always be transparent about it. As a government, we often fail in this regard." Niels Wiersma from municipality Eindhoven (MR1) underscores the importance of good governance in data sharing, especially when addressing new questions or adding new aspects to area development: "It's crucial to organize governance around data sharing properly. Nationally, the first steps have been taken, but there is still room for



improvement." Niels Wiersma from municipality Eindhoven (MR1) also acknowledges the challenge of transparency in data collection and usage by government institutions: "*This is tricky because it could involve a long list of data. People might comply, but we must question whether such practices are transparent enough for a governmental institution.*"

Within mobility providers, opinions on openness differ. While companies can be transparent about many things, sharing competition-sensitive information is a different matter. Dani Sprecher From MyWheels (IR4) explains, "*If you have a small lead, you want to keep it. If things are going badly somewhere, then not everyone needs to know.*" This highlights the delicate balance between transparency and maintaining a competitive edge, emphasizing the need for thoughtful governance and ethical considerations in data sharing. Additionally this underlines the importance of addressing privacy and security concerns proactively and ensuring transparency in data handling to maintain public trust (IR4).

Antoine Gribnau from municipality The Hague (MR4) highlights the need for balance in data management, stating, "We live in such a society that everything has to be buttoned up with all kinds of contracts. It must be *necessary, but there has to be a balance somewhere between having everything tightly framed and having everything open."* This emphasizes the challenge of finding a middle ground between stringent data protection measures and the openness required for innovation and effective data sharing.

5.4.5. COMMAND AND CONTROL

The government's role in managing technological advancements has often been minimal, with a tendency to leave it to big tech companies, particularly from the U.S. A policy officer from municipality Rotterdam (MR6) questions this approach: "We've mostly let big tech handle it all for us, but the question is whether that's wise." Roland van der Heijden from municipality Rotterdam (MR5) highlights the ethical implications of this passive stance: "You see that this position is becoming increasingly untenable. You see that the whole ethical aspect is becoming more and more unsustainable. This is a development we are very much involved in, and we find it very important that companies understand this. But we are working very hard on it." The municipality itself is very aware of the excessive data collection of large corporations and therefore appointed a philosopher to incorporate data ethics into the digitization program (de Lange, 2020).

Entering the technological world where the rules are not yet established, makes it very challenging for both the government and market players to determine the boundaries within which they operate (ER1). In recent years, various municipalities have primarily left the market in charge, giving the market free rein to develop and innovate in the area of shared transportation (MR6). This also applies to other digital developments. Municipalities have allowed these advancements to progress with little oversight and have shown little initiative, resulting in them now lagging (MR5).

Shared mobility apps collect data from every trip, which can be valuable for identifying issues, opportunities, and solutions in the urban domain (Berretta, 2023; European Commission, 2023c). Often, big tech companies have a better understanding of how certain urban aspects function and how to address related issues. Additionally, more physical infrastructure data is being discovered and processed by private companies. While they often gather this data to assist municipalities directly, they can also charge for this service. This situation, however, makes cities directly dependent on these companies, which is both democratically and economically undesirable (MR5). Initially, the municipality might be able to cover the financial costs, but as their dependence on the data grows, private companies have the freedom to increase their prices (MR5). Roland van der Heijden from



municipality Rotterdam (MR5) warns, "If we continue down the current path, you could say that we will end up as a government, as a customer in our own city." CDS-M and TOMP-API expert Edwin van den Belt (ER4) provides an example of a situation which could happen when municipalities become too dependent on a certain saying, "We tried to ensure that agreements are made to prevent companies like Apple and Google from having the opportunity to, say, independently contract with an organization like NS. Towards the end of the contract term, these companies might say, 'Hey NS, we see that about 70% of your revenue flows through our platform. We want a bit extra for that.'"

Municipalities must be vigilant about entering into data delivery contracts with (large and dominant) private parties (ER2). They must have clear terms to avoid the tendency to vendor lock-in and ensure that they are not overly dependent on specific data standards or providers (Veronelli, 2016; (Bagheri, 2024; Maltha et al., 2021). Especially when companies have a monopoly position, they have the ability and freedom to set conditions for the use of data, which can deprive municipalities of access, restrict it, or essentially become blackmailed when the municipalities depend on it directly (Gemeente Amsterdam, 2021b; Bagheri, n.d.). Municipalities should contractually establish ownership of source data to avoid later problems (Meerman, 2023; ER4), such as indicated by Roland van der Heijden from municipality Rotterdam (MR5) where it has not been able to receive underlying raw data from a certain data provider (for free).

5.4.5.1. HIGHER-LEVEL GOVERNANCE

Currently, municipalities often struggle with the balance between autonomy and centralization, as well as unity within decision-making and problem-solving (Keur. Peter-August, 2019). Rik Braams from TNO and the Ministry of I&W (ER2) notes, "It's very difficult to speak about one vision because I don't think it exists." A policy officer from municipality Rotterdam (MR6) emphasizes the need for stronger interaction between national and local governments: "The interaction between the national government and municipalities should be much stronger." Niels Wiersma from municipality Eindhoven (MR1) adds that having a national legal framework would simplify coordination: "If there were a national legal guideline or framework, it would be much easier to regulate centrally." According to the Masterplan Mobiliteit 2050 of municipality Eindhoven shared mobility is something that goes beyond municipal barriers and therefore should be regulated in cooperation with regional authorities (Gemeente Eindhoven, 2023). These are all indications that municipalities can not effectively innovate and integrate shared mobility alone. Central regulation and cooperation is needed to provide fair and transparent collaboration between municipalities and shared mobility providers.

Gemma Schepers from municipality Amsterdam (MR7) suggests more centralized steering through national programs like Natuurlijk Deelmobiliteit (including CDS-M) to ensure uniformity in data sharing: "We have now a national program, Natuurlijk Deelmobiliteit, which ensures uniformity in data sharing and defines technical APIs." In which Gemma Schepers from municipality Amsterdam (MR7) praises the national collaboration, "It's very good that we have organized this nationally, so not every city comes up with something different."

Contrarily, policies are often developed from another or higher government position to be implemented by others. In practice, however, this is much more difficult than previously estimated by the higher government. They often lack concreteness about the practice and implementation of policies. As a policy officer from municipality Rotterdam (MR6) says: "A legion of optimistic technicians who are eager to play the innovation card. There is a great need for assistance in a lagging field like mobility. The thinking is, if we just do this now, you'll quickly see how fast we can grow into a beautiful new mobility system. Many things that are not fulfilled but are often initially embraced by politicians."



5.4.6. COMMUNICATION AND COLLABORATION

Cities often find it challenging to engage in meaningful dialogue with businesses. It is crucial to listen to the experiences of these parties, as they possess intricate knowledge of their operations. Engaging with them allows for a better understanding and more effective collaboration (MR7). Gemma Schepers from municipality Amsterdam (MR7) highlights this, stating, "For instance, during discussions with Check, they share insights that are novel to me due to my different perspective. Conversely, I provide them with the city's viewpoint, which they might not be fully aware of."

This continuous exchange of information ensures mutual awareness and fosters collaboration, which is also indicated by Dani Sprecher From MyWheels (IR4), noting that entering into dialog with shared mobility providers opens up opportunities to learn from each other and fosters collaboration. However, this process can be difficult, and the ease of communication can vary from one city to another (MR7). Gemma Schepers from municipality Amsterdam (MR7) notes that trust is essential for effective collaboration: *"Without trust, collaboration would not be possible" (MR4)*. Moreover, including representatives of frequent users of shared mobility solutions or even regular residents can add a valuable perspective to these discussions, as they view these solutions differently (MR1). Niels Wiersma from municipality Eindhoven (MR1) emphasizes the importance of incorporating these perspectives to enrich the dialogue and improve outcomes.

5.5. MUNICIPALITY AND SHARED MOBILITY PROVIDER COMPARISON

Although many aspects in this section are already described in previous sections it is developed to give a quick insight into the differences, similarities, and exceptions among shared mobility providers and Dutch municipalities. Based on initial codes and thematic analysis this section provides a cross-case analysis of the results found between car-sharing and shared two-wheel providers as well as Dutch municipalities. In doing so, this comparison contributes to the degree of generalizability of the results and the interpretations given to them. For substantiation and direct thematic coding results see **Appendix E**.

5.5.1. SHARED MOBILITY DATA SUPPLY COMPARISON

During the research, it became clear that within shared mobility data sharing a distinction can and should be made between shared two-wheel providers and car-sharing providers to form a clear picture of the shared mobility data supply. By looking at these parties individually, municipalities can get a better understanding of the companies and their vision for data sharing and build better policies and frameworks accordingly.

In essence, the type of data shared by both parties is similar. Both parties primarily share trip data, but in some cases also include aspects such as usage, complaints, and potential initiatives regarding road quality. A two-wheeler provider also mentioned proactively sharing mismatch data. This data covers the time and location when someone opens the app but cannot find a shared vehicle within reach. This data can be beneficial for both municipalities and shared mobility providers. However, the specific content of shared data significantly differs between two-wheeler and car-sharing companies.

Both parties indicate that data sharing heavily depends on agreements with different municipalities. What municipalities request and what providers are willing to offer can vary greatly. Generally, it can be concluded that shared two-wheeler providers are more willing and able to share extensive data with municipalities, partly because they are bound by legal agreements and permits. On the other hand, carsharing providers prefer not to go into detail when sharing data with municipalities. They believe that



municipalities can manage with less specific data, thus they prefer to share annual averages rather than direct data about individual vehicles. The level of specificity is the main difference between the two parties.

Despite both parties recognizing the benefits of data sharing for urban development and policymaking, they are both extremely cautious about privacy and business-sensitive information. An important example of maintaining privacy is only sharing start and end locations instead of the entire trip, or anonymizing and lightly editing data before sharing it.

While shared two-wheel providers have already agreed to mandatory data sharing, car-sharing providers are not in favor of this and indicate that they are only willing to share very limited data on a mandatory basis. They suggest that much more is possible if municipalities engage in dialogue with the provider and the provider shares data voluntarily. Both parties can and want to proactively share data with municipalities to contribute to urban development and policymaking and to give something back to the municipality. Shared two-wheel companies explicitly mention creating win-win scenarios through proactive data sharing, where the provider shares data to improve public spaces and infrastructure, potentially tailored for (two-wheel) shared mobility.

Both parties, however, see the risks of open data and are therefore cautious about what they share and how it is used and presented in practice. To facilitate this data sharing, there are various initiatives in place. Two-wheel sharing companies do not explicitly mention third parties or facilitators. However, car-sharing providers frequently mention CROW and Vianova as third-party data dashboards for municipalities. It is noted that this can be challenging in practice because there are too few different providers in an area, making it too easy to trace data back to a specific provider. In addition to data dashboards, car-sharing providers state that they want to operate according to the CDS-M data-sharing model from the program Natuurlijk Deelmobiliteit. While shared two-wheel providers do not mention this, car-sharing providers see it as the core of (potential) data sharing.

The extent to which municipalities have access to the providers' data depends on legal agreements and the attitude of the provider. While shared two-wheel providers sometimes even share data proactively with municipalities where they do not operate, car-sharing providers typically only share data in response to municipal requests.

5.5.2. MUNICIPAL DATA DEMAND COMPARISON

To realize data sharing, municipalities emphasize that communication with shared mobility providers is essential. Both parties need to know and understand each other's perspectives to establish good and clear agreements. Within these agreements, the use of APIs is crucial to read the data. Several municipalities also indicate that they use third parties to read data. Specifically, the Municipality of Rotterdam and the Municipality of The Hague mention the Open Urban Platform. This platform aims to facilitate data sharing from various parties by offering a data marketplace.

Most municipalities state that data should be open and accessible to everyone. In practice, various municipalities acknowledge that while they possess a lot of data, specific (external) data is often not acquired yet. The Municipality of Rotterdam cites a specific case where it sought access to the underlying raw data of an application but was denied access by the data owner.

All municipalities focus on using data for better decision-making, public tasks, and improving the quality of life. Several municipalities specifically mention their efforts to manage urban growth, public space, and especially mobility. The Municipality of Amsterdam notably highlights the use of data to



reduce the nuisance caused by shared mobility. Similarly, the Municipality of The Hague also indicates a desire to use data to facilitate mobility hubs to mitigate nuisance.

All municipalities recognize that they should not request data from shared mobility providers unless they have a direct use purpose. Additionally, most municipalities indicate that mobility data can be used for multiple purposes beyond just mobility-related questions. In some cases, new, unexpected results can be obtained based on mobility data. Some municipalities also directly state that personal data has no use purpose and is therefore not important in their data requests. Various municipalities acknowledge that data leads to opportunities that have not yet been identified.

The continuity of data sharing varies between municipalities and depends on permits and other agreements with shared mobility providers. Municipalities seek automated data sharing where as much data as possible is shared, but in many cases, data is only shared periodically and in a less specific manner.

Acquiring data from shared mobility providers requires effort. However, many municipalities lack the capacity for this effort and often do not know what to do to obtain the data or which data they need, as mentioned by the Municipality of Rotterdam. In contrast, the Municipalities of Eindhoven and The Hague state that there are few obstacles and that data acquisition is generally proceeding well. The Municipalities of Rotterdam and The Hague note that the challenge mainly lies in the organizational capabilities of the parties rather than technical difficulties. On the other hand, the Municipality of Amsterdam states that technology is still a challenge within the municipality.

For data utilization, many municipalities often do not have the right capacities and capabilities to analyze data and integrate it into policy decisions and urban development. Additionally, a central theme among the municipalities is ensuring the privacy and security of the data. The Municipality of Amsterdam points out that there are frequent changes in the market, which complicate data usage. The Municipality of Rotterdam has a unique perspective on the mismatch in data utilization. The Municipalities of Rotterdam and The Hague also acknowledge that data application must be done at various levels and within different municipal departments.



6. DISCUSSION

This chapter critically examines the results of this study by putting it into contrast with existing literature, again distinguishing between data demand, supply, and challenges found through a systematic literature study which can be found in **Appendix D**. In addition, the limitations of the study are acknowledged and the significant contributions of the study are named. Lastly, recommendations for further studies are made based on the limitations and results of the study.

6.1. EMPIRICAL FINDINGS AND IMPLICATIONS

In addition to the deductive coding strategy using an initial codebook, this research also employed inductive open coding. Many of the initially divergent themes align with the challenges identified by Diran & van Veenstra, (2020). Examples include: 'Data ownership', which corresponds to 'lacking access rights and scattered distribution of data'; 'market dependence', which aligns with 'cautious to make decisions based on insights from data'; and 'resistance to change', which corresponds to 'lack of awareness, trust, and openness in data sharing.' Emergent themes, however, include 'central vision,' 'role division,' and 'higher level governance,' all of which suggest a new theme described as '(central) governance and role division' in data sharing. Additionally, other themes pertain to specific aspects of the urban development and policymaking sector.

6.1.1. DATA DEMAND

Although little literature exists on the specific research topic, there is a literature base in the area of data sharing for public benefits and urban planning & policymaking. X. Liu & Dijk, (2022) indicate that there is a data need for real-time traffic and mobility-related data. This is important in short-term policy development. This research is in line with this in terms of data demand. Municipalities often want to obtain shared mobility data to solve short-term policy problems, but shared mobility providers indicate that municipalities should not do this and should look at the longer term, which directly corresponds to more concrete policy problems and use cases for data. According to the mobility providers, this is the main reason why shared mobility providers would be reluctant to provide data, and therefore supply and demand do not match. A tension is seen in this since if mobility providers are continuously asked to provide data it incurs more costs and can thus lead to even less willingness of companies to share data. Therefore data demand standardization is desired from both parties, but need close cooperation in establishing it.

Leveraging data and integrating it with smart technologies can revolutionize urban living and management (França et al., 2021; Wang et al., 2021). By harnessing the power of big data, cities can make more informed, timely, and effective decisions, leading to more sustainable and efficient urban environments. Borghys et al. (2020); Hawken et al. (2020); Susha et al. (2019), highlight the benefits of data collaboration in city developments, including improved decision-making, increased process efficiency, and innovation. Businesses could contribute to the development of smart and sustainable cities (Hudović Kljuno & Krivošić Dizdarević, 2021). This research connects to this and provides specific insights into shared mobility and its data in Dutch cities. Municipalities see direct benefits of shared mobility for addressing urban issues like densification and urbanization. Additionally, they aim to reduce private vehicle ownership to free up public space and achieve sustainability goals. This often drives municipalities to request data from shared mobility providers to see where and how demand for this modality is currently established and might transition in the future. This aligns with the literature by França et al., (2021) and Borghys et al. (2020) emphasizing the critical role of data in smart city initiatives, noting that data-driven approaches can significantly improve urban infrastructure



management and sustainability efforts. The demand for data is also influenced by increasing digitization and the need for municipalities to adapt to rapid technological changes (França et al., 2021). This is also confirmed by this research, noting that municipalities are afraid of 'a second Uber' which might disrupt the status quo in the mobility regime within Duch municipalities. This indicates that the municipal data needs to regulate NMS initiatives.

Hawken et al., (2020) and Verhulst, (2021) emphasize that while private companies collect vast amounts of valuable data, sharing this data with municipalities is still in the early stages. This is partly confirmed by this research. Shared two-wheel providers are often already sharing data and comply with municipal data demand due to obligation through permits, which aligns with the reason noted by Klievink, Van Der Voort, et al., (2018). However, currently, shared mobility providers perceive the data requests of municipalities often as unfounded, asking for excessive data with suboptimal purposes, leading to reluctance from providers to share data. Shared mobility providers, especially carsharing providers, indicate that municipalities must avoid unjustified data demands, also called 'fishing expeditions,' which are legally questionable. Providers are open to data sharing if municipalities have specific, justified use cases, which are termed policy problems according to CDS-M terms. CDS-M and its corresponding program of Natuurlijk Deelmobiliteit are still developing in collaboration with shared mobility providers and municipalities. However, municipalities express mixed opinions about these statements. Some municipalities admit that they do not always fully understand what data they need, and therefore cannot issue clear data demands to the shared mobility providers. On the other hand, municipalities assert that they have a right to certain data and should receive it from the shared mobility providers. The right to and ownership of data remains a significant point of discussion. CDS-M aims to address these issues and provide clarity regarding data demand and supply, as well as the methods of data sharing. Essentially this program aims to tackle most of the challenges identified in this research, specifically focusing on matching shared mobility data supply and demand through the development of standardized data sharing and use cases.

The need for clear, actionable data demands supports the conclusions of Verhulst, (2021), who identifies the gap between data supply and demand as a major challenge in data-driven urban planning. More and more Dutch municipalities are trying to incorporate data demand into permits for shared mobility providers. While this has been successful for most shared two-wheeler providers, the goal is to organize this for car-sharing providers as well. The data requirements municipalities include in these conditions, or their data demand in general, vary greatly between municipalities but also depend on the shared mobility provider. The larger the provider, the greater its influence, and the larger the data demand from the municipality. This also affects the continuity of data sharing. Ideally, municipalities want to receive as much and as direct data as possible from each provider, but often, static, less frequent data sharing is already sufficient for municipalities. Municipalities also recognize the need to be cautious with the data-sharing conditions they include in permits because they depend on the availability of shared transport and do not want to drive providers away from the city.

6.1.2. DATA SUPPLY

Vigorito, (2022) emphasizes that data sharing presents risks such as privacy, and security. Several studies point out the challenges of data sharing to municipalities. For example, Zhang, (2019) and Benli-Trichet & Kübler, (2022) discuss the need for strong governance structures and regulations to keep disrupting initiatives under control. The issues of data privacy and the legal constraints imposed are significant barriers, as noted by (França et al., 2021) and further supported by this study's findings. Shared mobility providers are vigilant about collecting privacy-sensitive data. European legislation (GDPR) limits the collection of personal data, and mobility providers generally avoid collecting



privacy-sensitive data unless necessary for specific use cases. However, in some cases, this is not yet fully understood by municipalities. The data request is then not only not wanted but also impossible. Additionally, the competitive landscape plays a significant role in data-sharing practices. Data is valuable and considered intellectual property that requires protection to maintain a competitive advantage. Mobility providers are cautious about sharing data, especially with commercial parties, to avoid compromising their business interests. Privacy sensitivity remains at the core of concerns in data sharing. To mitigate privacy risks, data is anonymized and aggregated as much as possible. Ensuring individual user privacy is a central responsibility of mobility providers. This is in line with (Klievink, Janssen, et al., (2018), which note that companies must be cautious of over-compliance with information-sharing regulations, as this can lead to unintended consequences. Striking a balance between compliance and value creation is crucial (Klievink, Janssen, et al., 2018)

(Hawken et al., (2020) indicate that big internet businesses use vast amounts of geospatial data to improve their functioning and financial growth, but they often overlook the potential public benefits of their data. Additionally, governments are slow to recognize that cities might have ownership of this data or could integrate this data into their development. According to this study's findings, it is shown that there are roughly 3 perspectives on data ownership. These are the data generator- often the transport user, the data collector- often the transport provider, and the municipality that facilitates the infrastructure. Throughout the research it is indicated that data ownership is extremely hard to determine and most respondents did not want to explicitly state who should or would be the rightful owner of data generated through mobility usage. Even though sometimes it is said that the individual data generator should be the owner one needs to address to complexities of access and storage of this data, which individuals find hard to accomplish. In addition, municipalities often do not know which data they need for a specific policy question. Engaging in dialogue with shared mobility providers allows municipalities to learn from the market's expertise. However, this dialogue and learning are mutual, meaning that both parties can benefit from collaboration and communication.

As for the level of specificity and specific types and content of data, it depends on the case. Some municipalities demand more and other data than others and some shared mobility providers provide more or other data than others, with some municipalities receiving raw data and others only general insights. Generally speaking, there is a lack of coherence and centralization in data sharing, which indicates an immediate demand for centralization and handling data sharing ethically. This is in line with research from (Estevez et al., (2016) indicating that technological solutions should not neglect legal, social, and ethical impacts. This is also what directly influences the differences between carsharing and shared two-wheel providers. Whereas the two-wheel providers are the disruptive innovator and therefore need permits to operate within the city and car-sharing providers are the incumbent firm which is already settled within the city. On top of that, these companies create far fewer nuisances so there is no immediate reason for municipalities to request data, whereas two-wheelers do. Therefore dynamic data is often shared by two-wheel providers and more static data is shared by car-sharing companies. However, shared mobility providers become more cautious when data requests become more specific, privacy-sensitive, or business-critical. Shared mobility providers primarily share trip data, such as start and end points, trip duration, and aggregated information to protect privacy. However, there is a clear distinction between car-sharing and two-wheel providers, with the latter often sharing more comprehensive data. Car-sharing companies typically wait for municipalities to present a substantiated data demand and specific use case for the data.

Throughout the literature, most practical evidence has been found for voluntary B2G data sharing and win-win scenarios (Rukanova et al., 2020; Susha, Rukanova, et al., 2019; Klievink, Van Der Voort, et al., 2018; Vigorito, 2022). Within the shared mobility domain, in principle, all parties are willing to



share data voluntarily. However, mandatory data sharing is currently only in place for shared twowheel providers. Share car providers often only share data voluntarily because they are not (yet) subject to licensing requirements. However, it should be noted that this data sharing is only done after concrete municipal data demand, with clear use cases for the data. Car-sharing providers are generally a lot older and mature and therefore already often present within cities before data-sharing demand emerged and shared two-wheel providers began to integrate within the urban public space. Shared two-wheel providers were taken hand in hand with nuisance, causing municipalities to implement direct regulation through permits and therein also data sharing. This does not go up for car-sharing providers, as they do not cause the same amount of nuisance and disruption. However, municipalities are increasingly pushing towards mandatory data sharing for all mobility providers.

However, this study indicates that that shared mobility data sharing could be both on voluntary or mandatory basis. Although voluntary data sharing is also confirmed by this study as it indicates that shared mobility providers are sharing data to potentially increase municipal infrastructure and even specific mobility hubs for the providers, it also provides evidence for successful mandatory data sharing, confirming with Rukanova et al., (2020) and Vigorito, (2022). One of the interviewed carsharing providers even opted for a mandatory data-sharing basis, where additional data can be shared with municipalities voluntarily. This could contain data that municipalities regularly need to confirm shared mobility providers operate according to the given permit (such as operating vehicles and frequency of use) and for other use cases/ policy problems, municipalities could come to a dialog with shared mobility providers to provide additional information voluntarily. This way shared mobility providers are given the opportunity to understand the municipal problem, and allow them to find the appropriate data to share to solve this problem. This is consistent with the findings of Vigorito, (2022), which indicates that in many cases an optimum of data sharing can be achieved when both methods are combined. In addition, shared mobility providers feel a responsibility to contribute to municipal goals and infrastructure improvements. By proactively sharing data, they often see benefits in the form of improving services and infrastructure, therefore enabling mutual benefits.

6.1.3. DATA ACQUISITION AND UTILIZATION CHALLENGES

This research started with an expectation of problems using the theory of Diran & van Veenstra, (2020). Although they specifically dedicate their research to heat transition policymaking, the barriers they are mentioning were not expected to differ significantly for policymaking based on shared mobility data. This expectation was right. This research has confirmed almost all the problems experienced by Diran & van Veenstra, (2020). Especially, the barriers such as lack of expertise and skills, determining the value and purpose of data, legal limitation/ GDPR, and difficulties in linking, analyzing, and visualizing data are dominant throughout the research. Although municipalities seem to know the value of data, they lack the expertise and skills to deal with them and often do not address specific purposes of data use. This also refers to the effect that municipalities find it difficult to analyze data, and make connections. In this study, this is mainly referred to as "silo thinking. Municipalities are often too individualistic, limiting collaboration, connections, and mutual learning opportunities. It has also been shown that data, especially from car-sharing providers, often do not meet the municipalities' intended level of detail. However, shared mobility providers often think that detailed data is often not necessary to share. This also ties in with the legal limitations of data use and privacy laws. Shared mobility providers in some cases hide behind this legislation to share as little data as possible with municipalities. Municipalities must also comply with the legal frameworks and powers in not only the data demand, but also its use.



As municipalities often lack the knowledge and technical capabilities to access and analyze data, they often rely on third-party platforms, such as Vianova and CROW. These platforms help integrate and visualize data, though challenges remain, especially with car-sharing data. However, dependence on these parties may compromise internal knowledge accumulation, which is a unique challenge identified in this study. This dependence is not extensively discussed in the literature, which often assumes internal capabilities.

This research highlights the mismatch of expectations between policymakers and shared mobility providers. There is a significant challenge due to different expectations and perceptions of the usefulness of data between policymakers and providers as well as divided perspectives on providing a clear use purpose for the data demand. This aspect is indicated by Diran & van Veenstra, (2020), but receives a lot more body within this research. Meanwhile, this study found no significant indication that data was scattered distributed, or untraceable by municipalities. Only when looking at shared mobility as a whole one can speak of scattered distribution. In practice, each provider has its own data, over which the municipality can express its need. Municipalities actually do know where to find data. However, the degree of access is significantly dependent on agreements and the willingness of the shared mobility provider to share data. Also, it should again be noted that data often gets demanded from single municipal departments without other departments knowing. This indicates that data might be demanded redundantly and is scattered among municipal departments, asking for central data points within municipalities.

6.2. LIMITATIONS

This research contains several limitations. The first limitation is the specific focus on medium to largesized cities in the Netherlands. The findings may not be generalizable to smaller cities or cities in other countries with different urban dynamics and regulatory environments. In addition, the four municipalities each have at least one interview, with some being examined more thoroughly. The limited and varying number of interviews means that the findings may not fully capture the complexities and variations within each category. This limitation reduces the depth and reliability of the insights and hampers the identification of consistent patterns and themes across municipalities. The comparative findings are preliminary and should be seen as indicative rather than definitive.

The second limitation is the large focus on only municipalities, shared mobility providers, and experts/researchers on the research topic. Residents and/or shared mobility users are not included in the research, which can lead to a biased understanding of the data usage and its implications. However, it is indicated from different angles that also the vision of these stakeholders can contribute to the development of data sharing and that the interests of these stakeholders should be included in the discussion about data ownership and the purposes of use.

Third, this research is primarily limited only to the collaboration and interaction of data sharing between municipalities and shared mobility providers. However, the research also revealed that NMS contains more modalities, and MaaS is a direct extension of this data sharing. The MaaS platforms are another (often (semi) private) party that needs data from shared mobility providers.

Fourth, the research topic is very current so there is still much changing in the field of shared mobility within cities, but even more so in the data they collect and share with municipalities. Following this, the much-discussed policy program "Natuurlijk Deelmobliteit" was released on 30/05/2024, so it was not possible to include it in this research.



Lastly, this research is based on qualitative information, which introduces subjectivity in data interpretation by the researcher. In this researchers' preferences or background could have influenced the interpretation and direction of the research. Focusing on qualitative data provides a brought and deep understanding of the phenomenon of mobility data sharing, but hinders statistical presence and underpinning for generalizability. One should consider this study indicative and exploratory rather than definitive and exclusive.

6.3. CONTRIBUTIONS AND FUTURE DIRECTIONS

This research supports existing knowledge on sharing data for evidence-based decision-making. By providing empirical insights into the demand, supply, and challenges in acquiring and using shared mobility data, this research highlights critical aspects, differences, and conditions to effectively use data for smart and sustainable urban development and policy making, potentially improving sustainability and urban space, and reducing (mobility) nuisance. However, future research should explore best practices for data collaboration and integration, focusing on case studies of successful implementations to provide practical guidelines for municipalities and mobility providers. According to some respondents, the municipality of Utrecht is currently working with shared mobility providers, especially car-sharing providers, to reach data-sharing agreements.

This research is one of the first substantial movements in capturing supply and demand from shared mobility data. The qualitative approach has provided a broad exploration of the current situation, but could better demonstrate its value when statistical and quantitative underpinnings are given to the specific supply and demand variables needed for urban planning and policymaking.

This research demonstrates that shared mobility providers have a significant need for a wellsubstantiated and clear use case and data handling framework when municipalities request data sharing. However, this study has not fully clarified what constitutes a clear and justified use case and interest that would persuade shared mobility providers to share data. Further research should delve into both municipal and shared mobility providers' perspectives on creating this use case and data handling framework.

This research also shows that there are many issues involved besides sharing data, such as who should provide the data, data ownership, why it is sometimes not readily available or used, and any potential repercussions and solution. However, this provides opportunities for further research into data collaboration (in urban development) in which certain aspects can be explored in greater depth.

This research provides a comprehensive approach to the supply and demand of shared mobility data from the municipal demand perspective. However, the research also shows that MaaS is a movement that requires shared mobility data. Municipalities also see potential in this and are looking for ways to take advantage of it. Further research can provide its value when focusing on data sharing between shared mobility providers and Maas operators. This allows for triangulation between the three stakeholders, in which an optimum mutual state (of collaboration) could be identified, facilitating all stakeholders' ambitions.

This research provides a comprehensive approach to the current state of data sharing between shared mobility providers and municipalities, however, mobility crosses national borders. Besides improving quality and supporting the reliability of the results, it is interesting to investigate the current state of affairs in other (European) countries. This can provide valuable insights into differences and similarities, which can contribute to the discovery of best practices, centralization in data sharing, and further interoperability in data sharing and usage.



In closing, the research shows that a significant hope for central control, interoperability, and reliability in the area of shared mobility data sharing is placed on the CDS-M procedure, which is part of the Natuurlijk Deelmobiliteit program. Further research can delve into this particular program to provide guidance on the quality of the program and potentially suggest improvements. In addition, this can also be drawn to other countries, as the CDS-M method has ambitions to roll out further within the EU.



7. CONCLUSION

This research aimed to understand how Dutch municipalities (could) leverage shared mobility data in municipalities in the Netherlands. Therefore the research focused on three sub-questions: understanding the shared mobility data supply, understanding the data needs of Dutch municipalities, and examining the challenges and opportunities in data acquisition and utilization faced by Dutch municipalities. The framework of Susha et al., 2017) has been adapted to structure this research around supply and demand. However to identify the data sharing mode, this study included the distinction of data sharing mode based on theory from (Rukanova et al., 2020; Susha, Rukanova, et al., 2019; Klievink, Van Der Voort, et al., 2018; Vigorito, 2022). Additionally, to structure the challenges in data acquisition and utilization in the urban development context, the theory of Diran & van Veenstra, (2020) has been adopted. Following this theoretical framework and by conducting interviews with multiple Dutch municipalities, researchers and experts in the field of shared mobility data sharing and urban planning, and shared mobility providers, primary data has been gathered to identify the core of the problem as well as the (potential) solutions for the shared mobility data supply and demand mismatch. Additionally, document and literature reviews have provided context on the issue, clarifying the advantages and disadvantages of shared mobility data sharing, its operational mechanisms, and ways to improve it. Together, this has led to an answer to the main question: "What is the supply and demand for shared mobility data within Dutch municipalities to inform (smart) urban development and policymaking, and which access and utilization challenges and opportunities arise in the data sharing process?" The answer to this central research question is elaborated in this chapter based on the 3 sub-questions.

Sub-questions:

1. "How is the Dutch municipalities' shared mobility data demand formed and what data do they need for urban planning and policymaking purposes?"

Dutch municipalities acknowledge they often cannot innovate on their own and want to keep control over (possible) disruptive innovation. For municipalities to answer their policy questions and solve problems, they are looking for opportunities to collaborate with and acquire data from shared mobility providers as they note their positive potential to reduce car usage, fix space shortage, and stimulate sustainability. However, especially shared two-wheel providers lead to problems within urban areas, resulting in municipalities integrating data sharing as a condition in operating permits for shared two-wheel providers to gather data on vehicle usage and trips, aiming to inform urban development and policymaking.

The Dutch municipal shared mobility data demand is a combination of legal, practical, and policydriven factors. The data demand is mainly fueled by the will to overcome challenges related to urbanization, densification, and sustainable mobility. To request this data, municipalities must take into account legal restrictions and need to clearly articulate their needs and establish a legal basis for their requests to increase the willingness to share data from the perspective of shared (car) mobility providers.

The specific content of municipal shared mobility data demand varies and depends on factors such as market share, frequency of use, and data already acquired. The approach to data sharing also differs between shared two-wheeler and car-sharing companies, with the former often needing to share more data through licensing agreements and the latter tend to be more hesitant about data sharing. Clear communication and legally sound data requests facilitate collaboration, while indiscriminate data



requests and handling can even lead to conflict and ineffective data sharing. To ensure continuity of data sharing, municipalities, and shared-mobility providers must establish robust agreements that balance municipal data needs with providers' operational interests. This includes establishing clear standards for data sharing, understanding technological and legal constraints, and fostering collaborative approaches that benefit both parties.

2. "What is the current state of data-sharing initiatives between shared mobility providers and Dutch municipalities, and how is this data sourced?"

Shared mobility data provision involves a complicated mix of legal, operational, and collaborative dynamics and aspects. While shared mobility providers collect a lot of data on vehicle trips and usage patterns, they are bound by privacy regulations such as GDPR, which prohibits them from collecting and sharing personal information. This limitation means that while municipalities often seek detailed information, providers note they often can and will only share anonymized and aggregated data. Providers also protect their data as intellectual property to achieve a competitive advantage, essential for maintaining market share and winning contracts and licenses from municipalities.

Shared two-wheeler providers generally share more data, often in real-time or near real-time, while car-sharing providers are more cautious. Municipalities need to come up with better cooperation strategies, as they cannot directly apply their strategy of data obligations for shared two-wheeler providers to car-sharing providers, as this strategy was mainly based on obtaining data to prevent nuisance. Shared cars often do not cause nuisances, eliminating the municipal legal basis for data collection.

Data sharing between municipalities and mobility providers can be mandatory, voluntary, or a combination of both. Mandatory data sharing is more common for shared two-wheelers and is often included in city operating permits. However, car-sharing companies often resist mandatory data sharing due to competition and privacy concerns. Voluntary data sharing is often driven by mutual interests, such as improving services and infrastructure, and is seen as a way of increasing cooperation and flexibility. This indicates that there are opportunities for municipalities to directly encourage shared mobility providers to provide data, as they can mutually benefit. However, car-sharing providers advocate for a centralized approach to data sharing that clearly states the purpose for which the data is used and how it is processed, interpreted, and handled, meaning transparency in the whole data sharing and utilization process, leading to trust.

As most municipalities lack technical capabilities, shared mobility data could also be shared through the use of third-party dashboards. However, shared mobility providers also question the transparency of parties such as CROW, raising concerns about data handling and competitive risks.

3. "What are the technical and organizational challenges and opportunities experienced by Dutch municipalities in accessing and using data from shared mobility providers active in the Netherlands?"

Additionally, shared mobility providers are cautious about sharing data that could compromise user privacy or their competitive advantage and market positioning. This value of data makes providers protective of their datasets, especially when there is a risk of sharing sensitive information with competitors or third parties. However, the threshold for data sharing depends on the obligation of data sharing, but it also relies on the internal organizational structure and leadership. Shared two-wheel providers are often a lot younger as well as their teams when compared to car-sharing providers which



directly affects the willingness to share data. Therefore, municipalities are often sufficiently supplied with data from shared two-wheel providers but lack access to data from car-sharing providers.

For successful data sharing, trust is at the core of collaboration. Constructive dialogue and mutual respect are crucial for fostering cooperative data-sharing agreements. Municipalities and providers must both be transparent, prudent, and ethical in their handling of data. Ensuring that data-sharing agreements are clear, legally sound, and aligned with the public interest to maintain public trust and achieve (sustainable) urban development. However, there are still many obstacles to conquer for municipalities to translate the data into evidence and make policy and urban development decisions. Municipalities often lack the technical expertise and capacity to effectively analyze and utilize the data they receive. Many municipalities rely on third-party platforms like Vianova and CROW to access, process, and visualize shared mobility data. However, municipalities should be careful and limit dependence on these platforms as internal (technical) knowledge and capacity development might be at risk. Municipalities must invest in building their technical capacities or continue to leverage third-party platforms to manage and utilize the data effectively.

Urban planning and policymaking are often hindered by silo thinking and mismatched expectations between municipalities and providers as well as municipalities with higher-level government and internal departments. Breaking down these silos, higher-level governance, and a more centralized approach to data regulation and standards can help address these challenges. This also has great potential to solve the lack of standardized data formats and interoperability. Proprietary data formats and siloed data systems further complicate data sharing. Therefore, the centralized and currently developed CDS-M procedure should be used to come to concessions and stimulate mutual benefits between municipalities and shared mobility providers. This procedure ensures interoperability, guarantees equal data interpretation and standardizes data sharing which increases data availability for municipalities and decreases stress on shared mobility providers as transparent data handling and clear policy problems are presented as uses cases.

Overall, this research illustrates the current state of supply and demand in shared mobility data sharing. It enables municipalities to identify problems and opportunities in this area and potentially learn from the perspectives of shared mobility providers to reach better agreements. The research highlights the key aspects of shared mobility data sharing, most of which can be addressed by the CDS-M program. Additionally, the research provides valuable insights for shared mobility providers, allowing them to learn from the perspectives of municipalities to reach joint solutions and mutual benefits, and actively contribute to the further development of the CDS-M program.



8. RECOMMENDATIONS FOR PRACTITIONERS

To form clarity for municipalities on what they can do to strengthen data sharing between them and shared mobility providers, some clear points are briefly identified below.

4. Establish clear data-sharing agreements

Municipalities should establish a standardized way of data-sharing based on agreements with shared mobility providers. Within these agreements it should be clearly defined which data types and content municipalities need and at which frequency. For this, the municipalities should make clear what they need the data for, as demonstrated within the CDS-M data-sharing method. This will help in building trust and ensuring compliance with privacy regulations.

5. Strengthen collaborations with shared mobility providers.

Municipalities can start the conversation with shared mobility providers, as well as establish new communications and collaborative platforms. In doing so, it facilitates ongoing dialogue to solve problems and further innovate in the field of shared mobility. By facilitating publicprivate partnerships, both parties can better collaborate, understand and learn from each other, which can have the direct result of making data easier or earlier to share.

6. Promote voluntary data sharing

Municipalities can stimulate and encourage shared mobility providers to share data voluntarily by demonstrating mutual benefits. This can be done by showing best practices in which data sharing has contributed to urban planning and mobility solutions benefiting shared mobility providers.

7. Invest in data and digital infrastructure and human capabilities

Municipalities need to invest in solid data and digital infrastructure to be able to analyze, manage, and use (shared mobility) data. This includes setting up proper, secure data storage systems, employing data analytics tools, commitment to interoperable data standards, and above all, bringing technically savvy employees in-house through, for example, more and better employee training.

8. Ensure data trust and privacy

Municipalities must ensure that it is clear what data is used for, but also in what way and how it is handled. Again, having data standards is essential, but the method of reading data must also be consistent. Municipalities must demonstrate compliance with privacy laws and regulations and protect the interests of both mobility providers and user. This builds mutual trust and alleviates privacy concerns.



9. ACKNOWLEDGEMENTS

In terms of both content and process, I would like to express my gratitude to a select group of people for their support and assistance throughout this research. First and foremost, I extend my heartfelt thanks to my supervisor, Iryna Susha, for her frequent help, guidance, and support throughout the research process. The specific direction of shared mobility data sharing is a fairly new phenomenon which made it difficult to find an appropriate theoretical framework that also fits the supply and demand as well as the challenges and opportunities in data sharing. Fortunately, through constructive feedback from Iryna, I managed to develop a sound research method and structure to answer my research question and thereby making a significant contribution to the literature of data sharing as well as urban development and policy making.

Additionally, I cannot thank enough all the municipal officers, researchers and experts, and industry representatives who participated in the interviews for this research. Without their kindness to participate and share their knowledge this research would not be possible. I also wish to show my appreciation to those who, although unable to participate directly, referred me to colleagues, acquaintances, or friends for interviews. Lastly, I would like to thank my friends and family for their mental and process support. Their involvement in the urban development sector proved invaluable as they served as test interviewees.



10. BIBLIOGRAPHY

- Al-Rodhan, N. R., & Gérard Stoudmann Director, A. (2006). Program on the Geopolitical Implications of Globalization and Transnational Security Definitions of Globalization: A Comprehensive Overview and a Proposed Definition Senior Scholar in Geostrategy and Director of the Program on the Geopolitical Implications of Globalization and Transnational Security Geneva Centre for Security Policy. www.gcsp.ch
- Aryandoust, A., van Vliet, O., & Patt, A. (2019). City-scale car traffic and parking density maps from Uber Movement travel time data. *Scientific Data*, 6(1). https://doi.org/10.1038/s41597-019-0159-6
- Bagheri, S. (2024). Digital City Rotterdam: Open Urban Platform.
- Bakri, R., El-Sayad, Z., & Bakr, A. F. (2022). BIG DATA IN SMART TRANSPORTATION PLANNING: TOWARDS AN APPLICABLE SMART SOCIO-ECONOMIC TRANSPORTATION PLATFORM IN HISTORICAL MEGA CITIES. WIT Transactions on the Built Environment, 212, 13–24. https://doi.org/10.2495/UMT220021
- Beckers, F., & Aurup Poolsem, L. (2022, September 22). *Manual For Secure Exchange of Shared Mobility Data*. CDS-M. https://www.cds-m.com/
- Bednarska-Olejniczak, D., Olejniczak, J., & Svobodová, L. (2019). Towards a smart and sustainable city with the involvement of public participation-The case of Wroclaw. *Sustainability (Switzerland)*, *11*(2). https://doi.org/10.3390/su11020332
- Benli-Trichet, M. C., & Kübler, D. (2022). The political origins of platform economy regulations. Understanding variations in governing Airbnb and Uber across cities in Switzerland. *Policy and Internet*, 14(4), 736–754. https://doi.org/10.1002/poi3.312
- Berretta, D. (2023). Amsterdam wants to save your digital privacy. *EuroCities*. https://eurocities.eu/stories/amsterdam-wants-to-save-your-digital-privacy/
- Bezerra, A., Alves, G., Silva, I., Rosati, P., Endo, P. T., & Lynn, T. (2019). A Preliminary Exploration of Uber Data as an Indicator of Urban Liveability. http://www.globalcitiesinstitute.org/
- Bian, C. (2023). Data as Assets in Foreign Direct Investment: Is China's National Data Governance Compatible with its International Investment Agreements? *Asian Journal of International Law*, 13(2), 342–364. https://doi.org/10.1017/S2044251322000595
- Bibri, S. E., & Krogstie, J. (2017). Smart sustainable cities of the future: An extensive interdisciplinary literature review. In *Sustainable Cities and Society* (Vol. 31, pp. 183–212). Elsevier Ltd. https://doi.org/10.1016/j.scs.2017.02.016
- Bidasca, L., & de la Quintana, J. (2020). URBAN AGENDA FOR THE EU EXPLORING THE DEPLOYMENT OF NEW MOBILITY SERVICES.
- Boeije, E. (2019). "Er is geen vraagstuk waar de overheid alleen over gaat." *Morgens*. https://www.morgens.nl/actueel/er-is-geen-vraagstuk-waar-de-overheid-alleen-overgaat/?_gl=1*34tavz*_up*MQ..*_ga*MzM2Nzk1ODIyLjE3MTE0NDMzNTM.*_ga_GZKK1D6JEM *MTcxMTQ0MzM1Mi4xLjEuMTcxMTQ0Mzc5My4wLjAuMA



- Bollars, S., Verbeet, M., Boijens, M., Willems, W., Dogger, W., van Willigen, M., Mink, E., Teuben, H., Smith, F., Bosker, H., & Szabo, zsolt. (2021). *De toekomst van Mobility as a Service (MaaS)*.
- Borghys, K., van der Graaf, S., Walravens, N., & Van Compernolle, M. (2020). Multi-Stakeholder Innovation in Smart City Discourse: Quadruple Helix Thinking in the Age of "Platforms." *Frontiers in Sustainable Cities*, 2. https://doi.org/10.3389/frsc.2020.00005
- Borsboom-Van Beurden, J., Kruizinga, E., Rodrigues De Almeida, J., Kallaos, J., & Gindroz, B. (2021). *Smart City Guidance Package Summary*.
- Brakema, S. (2020, February 3). *Data Sharing Coalition has officially started*. SureSync. https://info.suresync.nl/news/data-sharing-coalition-officially-started
- Brinkerhoff, D. W., & Brinkerhoff, J. M. (2011). Public-private partnerships: Perspectives on purposes, publicness, and good governance. *Public Administration and Development*, *31*(1), 2–14. https://doi.org/10.1002/pad.584
- Burns, R., & Andrucki, M. (2021). Smart cities: Who cares? *Environment and Planning A*, 53(1), 12–30. https://doi.org/10.1177/0308518X20941516
- Butler, L., Yigitcanlar, T., & Paz, A. (2020). Smart urban mobility innovations: A comprehensive review and evaluation. In *IEEE Access* (Vol. 8, pp. 196034–196049). Institute of Electrical and Electronics Engineers Inc. https://doi.org/10.1109/ACCESS.2020.3034596
- Carlino, G. A., Chatterjee, S., & Hunt, R. M. (2007). Urban density and the rate of invention. *Journal of Urban Economics*, *61*(3), 389–419. https://doi.org/10.1016/j.jue.2006.08.003
- Clarck, T., Foster, L., Sloan, L., & Bryman, A. (2021). *Bryman's Social Research Methods* (6th ed.). Oxford University Press.
- CoE-DSC. (2023, September 4). *Proving the value potential of MPC for MaaS*. CoE-DSC. https://coe-dsc.nl/use-cases/proving-the-value-potential-of-mpc-for-maas/
- Cohen, T. (2018). Being ready for the next Uber: can local government reinvent itself? *European Transport Research Review*, *10*(2). https://doi.org/10.1186/s12544-018-0330-8
- Curzon-Butler, R. (2022). City Data Standard Mobility: a data standard for mobility.
- de Lange, P. (2020). Datagedreven werken bij gemeente Rotterdam Morgens. Morgens.
- De Vries, H., Bekkers, V., & Tummers, L. (2014). Innovation in the Public Sector: A Systematic Review and Future Research Agenda. www.lipse.org
- Depla, P., & Ollongren, K. (2017). brief Digitale Stad wethouders Ollongren en Depla 20feb17.
- Diran, D., Hoekstra, M., & van Veenstra, A. F. (2022). Applications of Data-Driven Policymaking in the Local Energy Transition: A Multiple-case Study in the Netherlands. *Lecture Notes in Computer Science (Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, 13392 LNCS, 55–72. https://doi.org/10.1007/978-3-031-23213-8_4
- Diran, D., & van Veenstra, A. F. (2020). Towards Data-Driven Policymaking for the Urban Heat Transition in The Netherlands: Barriers to the Collection and Use of Data. *Lecture Notes in Computer Science*



(Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics), 12219 LNCS, 361–373. https://doi.org/10.1007/978-3-030-57599-1_27

- Dunarintu, A., & Dociu, M. (2012). The Socio-Economic Impact of Urbanization. Dociu Madalina and Dunarintu Anca. International Journal of Academic Research in Accounting, 2(Special Issue 1), 47– 52.
- Ecorys. (2021). Maatschappelijke impact van Slimme en Duurzame verstedelijking.
- Eisenhardt, K. M. (1989). Building Theories from Case Study Research. In *Source: The Academy of Management Review* (Vol. 14, Issue 4). https://www.jstor.org/stable/258557
- El-Dardiry, R., & Overvest, B. (2019). *Position Paper Digitalisering*. https://www.cpb.nl/sites/default/files/omnidownload/CPB-Notitie-20nov2019-Position-paperdigitalisering.pdf
- Elstgeest, M. (2022). *Beginnen bij enthousiasme: zo werkt gemeente Eindhoven datagedreven*. https://www.morgens.nl/actueel-en-publicaties/
- ERTRAC. (2021). New Mobility Services Roadmap Status: final for publication. www.ertrac.org
- Estevez, E., Vasco Lopes, N., & Janowsk, T. (2016). *Smart Sustainable Cities Reconnaissance Study*. https://collections.unu.edu/view/UNU:5825#viewMetadata
- European Commission. (n.d.-a). Algemene verordening gegevensbescherming (AVG). 2022. Retrieved March 25, 2024, from https://eur-lex.europa.eu/NL/legal-content/summary/general-data-protection-regulation-gdpr.html
- European Commission. (n.d.-b). Van de richtlijn overheidsinformatie (PSI) tot de richtlijn inzake open gegevens. 2022.
- European Commission. (2016). REGULATION (EU) 2016/679 OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL.
- European commission. (2020). European Commission-Press release Shaping Europe's digital future: Commission presents strategies for data and Artificial Intelligence.
- European Commission. (2020a). *Governance of B2G data sharing across the*. https://digitalstrategy.ec.europa.eu/en/related-content?topic=70
- European Commission. (2020b). *The EU will create a single market for data where: THE EUROPEAN* DATA STRATEGY. https://doi.org/10.2775/645928
- European Commission. (2020c). Towards a European strategy on business-to-government data sharing for the public interest Final report prepared by the High-Level Expert Group on Business-to-Government Data Sharing. https://doi.org/10.2759/406717
- European Commission. (2022). European Commission-Press release Commission announces 100 cities participating in EU Mission for climate-neutral and smart cities by 2030.
- European commission. (2022). European Commission-Press release Data Act: Commission proposes measures for a fair and innovative data economy.



- European Commission. (2023a). *Het programma Digitaal Europa*. https://digitalstrategy.ec.europa.eu/nl/activities/digital-pr
- European Commission. (2023b). *Proposal for a Regulation laying down harmonised rules on artificial intelligence*. https://digital-strategy.ec.europa.eu/en/related-content?topic=96
- European Commission. (2023c). Saving digital privacy from shared mobility in Amsterdam. https://urbanmobility-observatory.transport.ec.europa.eu/news-events/news/saving-digital-privacy-shared-mobilityamsterdam-2023-03-30_en
- European Commission. (2024a). A European strategy for data. https://eur-lex.europa.eu/legalcontent/EN/TXT/?uri=CELEX%3A52020DC0066
- European Commission. (2024b). *European Data Governance Act*. https://digitalstrategy.ec.europa.eu/en/policies/data-gover
- European Parliament, & Council of the European Union. (2023). *REGULATION (EU) 2023/2854 OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL*. https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A32023R2854&qid=1710928686533&print=truehttp://data.europa.eu/eli/reg/2023/2854/oj
- Fiore, S., Elia, D., Pires, C. E., Mestre, D. G., Cappiello, C., Vitali, M., Andrade, N., Braz, T., Lezzi, D., Moraes, R., Basso, T., Kozievitch, N. P., Fonseca, K. V. O., Antunes, N., Vieira, M., Palazzo, C., Blanquer, I., Meira, W., & Aloisio, G. (2019). An Integrated Big and Fast Data Analytics Platform for Smart Urban Transportation Management. *IEEE Access*, 7, 117652–117677. https://doi.org/10.1109/ACCESS.2019.2936941
- França, R. P., Monteiro, A. C. B., Arthur, R., & Iano, Y. (2021). Smart Cities Ecosystem in the Modern Digital Age: An Introduction. In *Advanced Sciences and Technologies for Security Applications* (pp. 49–70). Springer. https://doi.org/10.1007/978-3-030-72139-8_3
- Friedmann, J. (2005). Globalization and the emerging culture of planning. In *Progress in Planning* (Vol. 64, Issue 3, pp. 183–234). https://doi.org/10.1016/j.progress.2005.05.001
- Gemeente Amsterdam. (2021a). City Data Standard-Mobility Smart Cities Strategy.
- Gemeente Amsterdam. (2021b). Datastrategie Gemeente Amsterdam.
- Gemeente Amsterdam. (2023). Agenda Digitale Stad 2023 2026.
- Gemeente Amsterdam. (2024, March 19). *City Data Specificatie Mobiliteit (CDS-M)*. Gemeente Amsterdam. https://www.amsterdam.nl/innovatie/mobiliteit/city-data-specificatie-mobiliteit-(cds-)/
- Gemeente Den Haag. (2020a). Datagedreven werken voor onze stad.
- Gemeente Den Haag. (2020b). Haagse visie Smart Mobility.
- Gemeente Den Haag. (2021). Omgevingsvisie Den Haag 2050.
- Gemeente Den Haag. (2024). OB08 Datamodel Smart City Den Haag.
- Gemeente Eindhoven. (2023). Masterplan mobiliteit 2050.



- Gemeente Rotterdam. (2021). LIVING LAB SENSIBLE SENSOR LIVING LAB SENSIBLE SENSOR IF THE PUBLIC SPACE COULD TALK, WHAT WOULD IT TELL YOU?
- Geonovum, & Ministerie van Binnenlandse Zaken en Koninkrijksrelaties. (2021). Verkenning Publiek Gebruik Data van Derden.
- George, J. J., Yan, J., & Leidner, D. E. (2020). Data Philanthropy: Corporate Responsibility with Strategic Value? *Information Systems Management*, 37(3), 186–197. https://doi.org/10.1080/10580530.2020.1696587
- Geropanta, V., Karagianni, A., Mavroudi, S., & Parthenios, P. (2021). Exploring the relationship between the smart-sustainable city, well-being, and urban planning: An analysis of current approaches in Europe. In *Smart Cities and the UN SDGs* (pp. 143–161). Elsevier. https://doi.org/10.1016/B978-0-323-85151-0.00010-5
- Gerte, R., Konduri, K. C., & Eluru, N. (2018). Is there a limit to adoption of dynamic ridesharing systems? Evidence from analysis of uber demand data from new york city. *Transportation Research Record*, 2672(42), 127–136. https://doi.org/10.1177/0361198118788462
- Grace Dobush. (2019, September 9). *Uber has troves of data on how people navigate cities. Urban planners have begged, pleaded, and gone to court for access. Will they ever get it?* Medium. https://marker.medium.com/ubers-real-advantage-is-data-e54984ff524c

Gunterman, J. (2020). Onderzoek 'De gemeente van Morgen.' *Morgens*. https://www.morgens.nl/actueel/resultaten-onderzoek-de-gemeente-van-morgenbeschikbaar/?_gl=1%2A1k22wax%2A_up%2AMQ..%2A_ga%2AMzM2Nzk1ODIyLjE3MTE0NDM zNTM.%2A_ga_GZKK1D6JEM%2AMTcxMTQ0MzM1Mi4xLjEuMTcxMTQ0MzUwMS4wLjAuM Aer-is-geen-vraagstuk-waar-de-overheid-alleen-overgaat%2F%3F_gl%3D1%2A34tavz%2A_up%2AMQ..%2A_ga%2AMzM2Nzk1ODIyLjE3MTE0ND MzNTM.%2A_ga_GZKK1D6JEM%2AMTcxMTQ0MzM1Mi4xLjEuMTcxMTQ0Mzc5My4wLjAu MA

- Hartley, J., Sørensen, E., & Torfing, J. (2013). Collaborative Innovation: A Viable Alternative to Market Competition and Organizational Entrepreneurship. *Source: Public Administration Review*, 73(6), 821– 830. https://doi.org/10.1111/puar
- Hawken, S., And, H., & Pettit, C. (2020). *OPEN CITIES OPEN DATA COLLABORATIVE CITIES IN THE INFORMATION ERA*.
- Hudović Kljuno, A., & Krivošić Dizdarević, L. (2021). DIGITALIZATION of CITIES-SMART CITY INTERVENTIONS in the CASE of SARAJEVO. International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences - ISPRS Archives, 46(4/W5-2021), 273–278. https://doi.org/10.5194/isprs-Archives-XLVI-4-W5-2021-273-2021
- Kalhor, K., & Mahdisoltani, M. (2015). Urbanization and its Effects on the Environment and Society Along with Sustainable Development Third International Symposium On Environmental and Water Resources Engineering. https://www.researchgate.net/publication/328732625
- Kamargianni, M., Li, W., Matyas, M., & Schäfer, A. (2016a). A Critical Review of New Mobility Services for Urban Transport. *Transportation Research Procedia*, 14, 3294–3303. https://doi.org/10.1016/j.trpro.2016.05.277


- Kamargianni, M., Li, W., Matyas, M., & Schäfer, A. (2016b). A Critical Review of New Mobility Services for Urban Transport. *Transportation Research Procedia*, 14, 3294–3303. https://doi.org/10.1016/j.trpro.2016.05.277
- Keur, P.-A. (2020, July 2). Gemeenten en hun informatievoorziening: transitie naar Common Ground. https://www.morgens.nl/actueel/gemeenten-en-hun-informatievoorziening-transitie-naar-commonground/?_gl=1*51bqwy*_up*MQ..*_ga*MzM2Nzk1ODIyLjE3MTE0NDMzNTM.*_ga_GZKK1D6J EM*MTcxMTQ0MzM1Mi4xLjEuMTcxMTQ0Mzc5My4wLjAuMA

Keur, P.-A., & Gunterman, J. (2021). De gemeente van Morgen.

- Keur. Peter-August. (2019). VNG: samen naar de gemeente van Morgen. *Morgens*. https://www.morgens.nl/actueel/vng-samen-naar-de-gemeente-vanmorgen/?_gl=1*zlw9g*_up*MQ..*_ga*MzM2Nzk1ODIyLjE3MTE0NDMzNTM.*_ga_GZKK1D6JE M*MTcxMTQ0MzM1Mi4xLjEuMTcxMTQ0Mzc5My4wLjAuMA
- Khokhar, S., Pathan, H., Raheem, A., & Abbasi, A. M. (2020). Theory Development in Thematic Analysis: Procedure and Practice. *Review of Applied Management and Social Sciences*, *3*(3), 423–433. https://doi.org/10.47067/ramss.v3i3.79
- Klievink, B., Janssen, M., van der Voort, H., & van Engelenburg, S. (2018). Regulatory compliance and over-compliant information sharing – Changes in the B2G landscape. *Lecture Notes in Computer Science (Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, 11020 LNCS, 249–260. https://doi.org/10.1007/978-3-319-98690-6_21
- Klievink, B., Van Der Voort, H., & Veeneman, W. (2018). Creating value through data collaboratives: Balancing innovation and control. *Information Polity*, 23(4), 379–397. https://doi.org/10.3233/IP-180070
- Lee, J., Babcock, J., Pham, T. S., Bui, T. H., & Kang, M. (2023). Smart city as a social transition towards inclusive development through technology: a tale of four smart cities. *International Journal of Urban Sciences*, 27(S1), 75–100. https://doi.org/10.1080/12265934.2022.2074076
- Liu, J., Cui, E., Hu, H., Chen, X., Chen, X., & Chen, F. (2017). Short-Term Forecasting of Emerging On-Demand Ride Services.
- Liu, X., & Dijk, M. (2022). How more data reinforces evidence-based transport policy in the Short and Long-Term: Evaluating a policy pilot in two Dutch cities. *Transport Policy*, 128, 166–178. https://doi.org/10.1016/j.tranpol.2022.09.022
- Liu, X., Dijk, M., & Colombo, C. (2024). Improving multilevel policy mixes for sustainable urban mobility transition. *Environmental Innovation and Societal Transitions*, 50, 100808. https://doi.org/10.1016/j.eist.2023.100808
- Loa, P., Hossain, S., & Nurul Habib, K. (2020). On the Influence of Land Use and Transit Network Attributes on the Generation of, and Relationship between, the Demand for Public Transit and Ride-Hailing Services in Toronto. *Transportation Research Record*, 2675(3), 136–153. https://doi.org/10.1177/0361198120966601
- Ludwig, C., Psotta, J., Buch, A., Kolaxidis, N., Fendrich, S., Zia, M., Fürle, J., Rousell, A., & Zipf, A. (2023). TRAFFIC SPEED MODELLING TO IMPROVE TRAVEL TIME ESTIMATION IN OPENROUTESERVICE. International Archives of the Photogrammetry, Remote Sensing and Spatial



Information Sciences - ISPRS Archives, 48(4/W7-2023), 109–116. https://doi.org/10.5194/isprs-archives-XLVIII-4-W7-2023-109-2023

- Luk, J. (2022). Datagedreven werken als vanzelfsprekend om opgavegericht te werken. Morgens.
- Maltha, S., Driesse, M., & Verhagen, P. (2021). *Digitaliseringsstrategie Een handreiking voor MRDH-gemeenten*.
- Martin, C. J., Evans, J., & Karvonen, A. (2018). Smart and sustainable? Five tensions in the visions and practices of the smart-sustainable city in Europe and North America. In *Technological Forecasting and Social Change* (Vol. 133, pp. 269–278). Elsevier Inc. https://doi.org/10.1016/j.techfore.2018.01.005
- Maselli, I., Lenaerts, K., & Beblavy, M. (2016). Five things we need to know about the on-demand economy. *CEPS Essay*. https://deliverypdf.ssrn.com/delivery.php?ID=501005104009101093100019003124102095100069023 0420440440770060051270710911010920001001241000570060570630500841160060840800290860 3303509202808300400006501309111902811902205402000507408911509902311507308612709402 3072000003067123066002029123115096013066031&EXT=pdf&INDEX=TRUE

Meerman, K. (2023). Een digitale replica van je gemeente: waarom?

Meijer, A. (2020). Eindhoven zet stap richting datagedreven mobiliteit met innovatielab. Sogeti.

- Meijer, A., & Thaens, M. (2018). Urban Technological Innovation: Developing and Testing a Sociotechnical Framework for Studying Smart City Projects. Urban Affairs Review, 54(2), 363–387. https://doi.org/10.1177/1078087416670274
- Mills, D. E., Izadgoshasb, I., & Pudney, S. G. (2021). Smart city collaboration: A review and an agenda for establishing sustainable collaboration. In *Sustainability (Switzerland)* (Vol. 13, Issue 16). MDPI AG. https://doi.org/10.3390/su13169189
- Mills, D., Pudney, S., Pevcin, P., & Dvorak, J. (2022). Evidence-based public policy decision-making in smart cities: Does extant theory support achievement of city sustainability objectives? *Sustainability* (*Switzerland*), 14(1). https://doi.org/10.3390/su14010003
- Ministerie van Binnenlandse Zaken en Koninkrijksrelaties. (2021). NL DIGITAAL: Interbestuurlijke Datastrategie Nederland.

Ministerie van Economische Zaken en Klimaat. (2018). Nederlandse Digitaliseringsstrategie.

- Muñoz, L. A., & Bolívar, M. P. R. (2021). Different levels of smart and sustainable cities construction using e-participation tools in european and central Asian countries. *Sustainability (Switzerland)*, 13(6). https://doi.org/10.3390/su13063561
- Nederland Digitaal. (2021, May 4). *Data Sharing Coalition: datadelen tussen sectoren*. https://www.nederlanddigitaal.nl/initiatieven/data-sharing-coalition
- Nesti, G. (2020). Defining and assessing the transformational nature of smart city governance: insights from four European cases. *International Review of Administrative Sciences*, *86*(1), 20–37. https://doi.org/10.1177/0020852318757063



- Neun, M., Eichenberger, C., Xin, Y., Fu, C., Wiedemann, N., Martin, H., Tomko, M., Ambuhl, L., Hermes, L., & Kopp, M. (2023). Metropolitan Segment Traffic Speeds From Massive Floating Car Data in 10 Cities. *IEEE Transactions on Intelligent Transportation Systems*, 24(11), 12821–12830. https://doi.org/10.1109/TITS.2023.3291737
- Nijhof, N. (2020). Aan de slag met deelmobiliteit! Case study onderzoek naar het gemeentelijk beleidsinstrumentarium om het gebruik van deelmobiliteit te stimuleren.

Nijman, H., & Alberts-de Gier, B. (2023). Opgavegericht en verantwoord datagedreven werken.

- NTM. (2023a). Mobility as a Service (Maas). https://www.toegangspuntmobiliteit.nl/onderwerpen/maas
- NTM. (2023b, January 12). *NTM aan de slag met datastandaarden Mobility as a Service (MaaS)*. https://www.toegangspuntmobiliteit.nl/actueel/nieuws/2023/1/12/ntm-aan-de-slag-met-maas-standaarden
- Ooms, W., Caniëls, M. C. J., Roijakkers, N., & Cobben, D. (2020). Ecosystems for smart cities: tracing the evolution of governance structures in a dutch smart city initiative. *International Entrepreneurship and Management Journal*, *16*(4), 1225–1258. https://doi.org/10.1007/s11365-020-00640-7
- Oregi, X., Roth, E., Alsema, E., Van Ginkel, M., & Struik, D. (2015). Use of ICT tools for integration of energy in urban planning projects. *Energy Proceedia*, 83, 157–166. https://doi.org/10.1016/j.egypro.2015.12.206
- O'Reilly III, C. A., & Tushman, M. L. (2021). *Lead and Disrupt : How to Solve the Innovator's Dilemma, Second Edition*. Stanford University Press. http://ebookcentral.proquest.com/lib/uunl/detail.action?docID=6647558
- Palinkas, L. A., Horwitz, S. M., Green, C. A., Wisdom, J. P., Duan, N., & Hoagwood, K. (2015). Purposeful Sampling for Qualitative Data Collection and Analysis in Mixed Method Implementation Research. *Administration and Policy in Mental Health and Mental Health Services Research*, 42(5), 533–544. https://doi.org/10.1007/s10488-013-0528-y
- Pardo Del Val, M., & Martínez Fuentes, C. (2003). Resistance to change: a literature review and empirical study. In *Management Decision* (Vol. 41, Issue 2, pp. 148–155). https://doi.org/10.1108/00251740310457597
- Parra-Domínguez, J., Alonso-García, M., & Corchado, J. M. (2023). Fuzzy Logic to Measure the Degree of Compliance with a Target in an SDG—The Case of SDG 11. *Mathematics*, 11(13). https://doi.org/10.3390/math11132967
- Paskaleva, K., Evans, J., & Watson, K. (2021a). Co-producing smart cities: A Quadruple Helix approach to assessment. *European Urban and Regional Studies*, 28(4), 395–412. https://doi.org/10.1177/09697764211016037
- Paskaleva, K., Evans, J., & Watson, K. (2021b). Co-producing smart cities: A Quadruple Helix approach to assessment. *European Urban and Regional Studies*, 28(4), 395–412. https://doi.org/10.1177/09697764211016037
- Perlman, J., & Roy, S. Sen. (2021). Analysis of human movement in the Miami metropolitan area utilizing Uber Movement data. *Cities*, 119. https://doi.org/10.1016/j.cities.2021.103376



- Rasche, A., Morsing, M., & Wetter, E. (2021). Assessing the Legitimacy of "Open" and "Closed" Data Partnerships for Sustainable Development. *Business and Society*, 60(3), 547–581. https://doi.org/10.1177/0007650319825876
- Ratcliff, C., De Bono, J., & Mratinello, B. (2023). *DE ALOMTEGENWOORDIGE DIGITALE EENGEMAAKTE MARKT*. www.europarl.europa.eu/factsheets/nl
- Rijksoverheid. (2016). Gemeentewet. https://wetten.overheid.nl/BWBR0005416/2024-01-31/0/afdrukken
- Rijksoverheid. (2019a). Nederland Digitaal De Nederlandse visie op datadeling tussen bedrijven.
- Rijksoverheid. (2019b). NL Digitaal Data Agenda Government.
- Rijksoverheid. (2020a). NL DIGIbeter 2020 Agenda Digitale Overheid.
- Rijksoverheid. (2020b). NL DIGITAAL NL DIGIbeter Agenda Digitale Overheid.
- Rijksoverheid. (2021a). Nederlandse Digitaliseringsstrategie 2021.
- Rijksoverheid. (2021b). The Dutch Digitalisation Strategy 2021.
- Rijksoverheid. (2021c). Toekomstverkenning Digitalisering 2030.
- Rijksoverheid. (2023). IBDS Jaarplan 2024.
- Romano, A. (2021). Pandemic and (Im)mobility: the spatial effects of the lockdown through digital platform's Big Data. *Rivista Geografica Italiana*, *4*, 5–22. https://doi.org/10.3280/RGIOA4-2021OA12956
- Roy, S. Sen, Perlman, J., & Balling, R. C. (2020). Analysis of urban mobility in South Florida using Uber Movement. *Case Studies on Transport Policy*, 8(4), 1393–1400. https://doi.org/10.1016/j.cstp.2020.10.003
- Royal HaskoningDHV. (2017). metropool regioamsterdam programma Smart Mobility.
- Ruijer, E. (2021). Designing and implementing data collaboratives: A governance perspective. *Government Information Quarterly*, *38*(4). https://doi.org/10.1016/j.giq.2021.101612
- Rukanova, B., Tan, Y. H., Huiden, R., Ravulakollu, A., Grainger, A., & Heijmann, F. (2020). A framework for voluntary business-government information sharing. *Government Information Quarterly*, 37(4). https://doi.org/10.1016/j.giq.2020.101501
- Sarwat, M. (2015). Interactive and Scalable Exploration of Big Spatial Data A Data Management Perspective. *Proceedings - IEEE International Conference on Mobile Data Management*, *1*, 263–270. https://doi.org/10.1109/MDM.2015.67
- Silva, B. N., Khan, M., Jung, C., Seo, J., Muhammad, D., Han, J., Yoon, Y., & Han, K. (2018). Urban planning and smart city decision management empowered by real-time data processing using big data analytics. *Sensors (Switzerland)*, 18(9). https://doi.org/10.3390/s18092994
- Sun, Y., Ren, Y., & Sun, X. (2020). Uber movement data: A proxy for average one-way commuting times by car. *ISPRS International Journal of Geo-Information*, 9(3). https://doi.org/10.3390/ijgi9030184



- Susha, I., Grönlund, Å., & Van Tulder, R. (2019). Data driven social partnerships: Exploring an emergent trend in search of research challenges and questions. *Government Information Quarterly*, 36(1), 112– 128. https://doi.org/10.1016/j.giq.2018.11.002
- Susha, I., Janssen, M., NI, M. F. W. H. A. J., & Verhulst, S. (2017). Data Collaboratives as a New Frontier of Cross-Sector Partnerships in the Age of Open Data: Taxonomy Development. http://hdl.handle.net/10125/41481
- Susha, I., Rukanova, B., Ramon Gil-Garcia, J., Tan, Y. H., & Hernandez, M. G. (2019). Identifying mechanisms for achieving voluntary data sharing in cross-sector partnerships for public good. ACM International Conference Proceeding Series, 227–236. https://doi.org/10.1145/3325112.3325265
- Susha, I., Schiele, J., & Frenken, K. (2022). Business-to-Government Data Sharing for Public Interests in the European Union: Results of a Public Consultation. *Lecture Notes in Computer Science (Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, 13391 LNCS, 515–529. https://doi.org/10.1007/978-3-031-15086-9_33
- Swofford, J. L. (2020). Uber's early entry decisions in the US. *Applied Economics*, 52(25), 2723–2729. https://doi.org/10.1080/00036846.2019.1694633
- Tao, X., Fu, Z., & Comber, A. J. (2019). An Analysis of Modes of Commuting in Urban and Rural Areas. *Applied Spatial Analysis and Policy*, *12*(4), 831–845. https://doi.org/10.1007/s12061-018-9271-9
- Teixeira, A. A. C., & Ferreira, C. (2019). Intellectual property rights and the competitiveness of academic spin-offs. *Journal of Innovation and Knowledge*, 4(3), 154–161. https://doi.org/10.1016/j.jik.2018.12.002
- The World Bank. (2023, April 3). Urban Development. IBRD.
- TNO. (2020). POLICY OPTIONS TO STEER MOBILITY AS A SERVICE: INTERNATIONAL CASE STUDIES.
- Tomor, Z. (2019). The Citipreneur: How a local entrepreneur creates public value through smart technologies and strategies. *International Journal of Public Sector Management*, *32*(5), 489–510. https://doi.org/10.1108/IJPSM-02-2018-0060
- Türk, U., Östh, J., Kourtit, K., & Nijkamp, P. (2021). The path of least resistance explaining tourist mobility patterns in destination areas using Airbnb data. *Journal of Transport Geography*, 94. https://doi.org/10.1016/j.jtrangeo.2021.103130
- UITP. (2020). NEW MOBILITY SERVICES MANAGING CHANGE AND DISRUPTION. https://bit.ly/3f2ZiwQ

UNECE. (2023). Smart Sustainable Cities. In UNECE. https://unece.org/housing/smart-sustainable-cities

UNEP. (2017). MAKE CITIES AND HUMAN SETTLEMENTS INCLUSIVE, SAFE, RESILIENT AND SUSTAINABLE Environmental dimension of SDG 11. https://www.unep.org/explore-topics/sustainable-development-goals/why-do-sustainable-development-goals-matter/goal-11

van Beurden, H. (2017). NL Smart City Strategie.



- Van De Wiel, A. (2023). *Je reis zoeken, boeken en betalen in één app: wanneer breekt mobility as a service door?* https://mtsprout.nl/author/amy-van-de-
- van den Berg, A., Hoekstra, M., & van Veenstra, A.-F. (2023). *Systeemfuncties voor datagedreven werken in de publieke sector*. www.vector.tno.nl
- Van Schijndel-de Nooij, M. (2022, May 16). *Smarter cities through better data sharing and accessibility*. Eindhoven University of Technology.
- Verhulst, S. G. (2021). The Value of Data and Data Collaboratives for Good: A Roadmap for Philanthropies to Facilitate Systems Change Through Data (pp. 9–27). https://doi.org/10.1007/978-3-030-78985-5_2
- Verhulst, S. G., Follow, , Susha, I., & Kostura, A. (2016). *Data Collaboratives: Matching Demand with Supply of (Corporate) Data to solve Public Problems*. https://sverhulst.medium.com/
- Veronelli, E. (2016). *Smart cities vs "locked-in" cities*. https://www.youris.com/energy/ecocities/smartcities-vs-locked-in-cities.kl?print=ok
- Vigorito, A. (2022). Government Access to Privately-Held Data: Business-to-Government Data Sharing: Voluntary and Mandatory Models. *European Journal of Comparative Law and Governance*. https://doi.org/10.1163/22134514-bja10030
- VNG. (n.d.). Vragen en antwoorden over gemeentewet. Retrieved March 25, 2024, from https://vng.nl/vragen-en-antwoorden?onderwerp%5B0%5D=380625
- VNG. (2020, June 12). Common Ground Slim gebruik van data. VNG. https://vng.nl/artikelen/commonground-slim-gebruik-van-data
- VNG. (2024). Voorwoord Trendrapport Informatiesamenleving 2024. https://vng.nl/artikelen/voorwoord-trendrapport-informatiesamenleving-2024
- Wang, A., Zhang, A., Chan, E. H. W., Shi, W., Zhou, X., & Liu, Z. (2021). A review of human mobility research based on big data and its implication for smart city development. In *ISPRS International Journal of Geo-Information* (Vol. 10, Issue 1). MDPI AG. https://doi.org/10.3390/ijgi10010013
- Wang, J., Kong, X., Rahim, A., Xia, F., Tolba, A., & Al-Makhadmeh, Z. (2017). IS2Fun: Identification of Subway Station Functions Using Massive Urban Data. *IEEE Access*, 5, 27103–27113. https://doi.org/10.1109/ACCESS.2017.2766237
- Wesselink, J.-W. (n.d.). *Wat gaan we doen?* Retrieved March 28, 2024, from https://agendastad.nl/wat-gaan-we-doen/
- Wesselink, J.-W. (2024, February 20). *Een slimme stad, zo doe je dat*. https://agendastad.nl/citydeal/een-slimme-stad-zo-doe-je-dat/
- Williams, R. (1997). Universal Solutions or Local Contingencies: Tensions and Contradictions in the Mutual Shaping of Technology and Work Organisation (Issue 8). International Thompson Business Press.



- Yang, B., Tian, Y., Wang, J., Hu, X., & An, S. (2022). How to improve urban transportation planning in big data era? A practice in the study of traffic analysis zone delineation. *Transport Policy*, 127, 1–14. https://doi.org/10.1016/j.tranpol.2022.08.002
- Yigitcanlar, T., Han, H., Kamruzzaman, M., Ioppolo, G., & Sabatini-Marques, J. (2019). The making of smart cities: Are Songdo, Masdar, Amsterdam, San Francisco and Brisbane the best we could build? *Land Use Policy*, 88. https://doi.org/10.1016/j.landusepol.2019.104187
- Zhang, C. (2019). China's new regulatory regime tailored for the sharing economy: The case of Uber under Chinese local government regulation in comparison to the EU, US, and the UK. *Computer Law and Security Review*, *35*(4), 462–475. https://doi.org/10.1016/j.clsr.2019.03.004
- Zhang, C., & Xu, Y. (2023). Institutional innovation essence and knowledge innovation goal of intellectual property law in the big data era. *Journal of Innovation and Knowledge*, 8(4). https://doi.org/10.1016/j.jik.2023.100417



11. APPENDIX A – (DATA AND DIGITALIZATION STRATEGY)

Based on the rapidly growing amount of data generated by both private and public institutions, Europe aims to become a leader in the data economy. Recognizing the untapped value and potential of data Europe aims to use its technologies, skills, industry, and vision to become a frontrunner in data sharing and create a society empowered by data (European commission, 2020).

Therefore the European Data Strategy (EDS) has been developed. It indicates the creation of data spaces where data is freely available to everyone, whether public or private, start-up or giant. The European Union (EU) acknowledges that creating incentives for data sharing and establishing practical, fair, and clear rules on data access and use, which comply with European values and rights such as personal data protection, consumer protection, and competition rules, is necessary. Furthermore, the strategy highlights the importance of making public sector data more accessible by opening up valuable datasets across the EU for reuse, thereby paving the way for innovation (European Commission, 2024a). To fully exploit the opportunities of data sharing, the European Commission wants to improve the development of technologies and infrastructure. The Commission will also work to further narrow the digital skills gap among Europeans and explore how to give citizens better control over who can access their machine-generated data (European commission, 2020).

Through policy development and investments, B2G data sharing is facilitated within the EU (European Commission, 2020a). Each country is expected to cooperate in setting up governance structures and data stewards. Transparency is also seen as very important, with citizens at the heart of acceptance and provision of data. Also, the technological and information aspects is of great importance in the development of efficient and effective B2G data sharing. The (mutual) benefits need to be discovered and explained, and support is needed for urban development based on data and digital infrastructure (European Commission, 2020c).

To achieve the goals of the EDS, the Data Governance Act (DGA), Digital Markets Act (DMA) and Digital Service Act (DSA), among others, were drafted. The DGA aims to promote data sharing mechanisms and trust in the availability and use of data and aims to facilitate the exchange of data between different sectors, including through data intermediaries, with the AVG (GDPR) at its core when it comes to personal data. It also deploys safeguards to strengthen trust in the exchange and use of data, thereby promoting the availability of data in the market. The DMA aims to regulate and set requirements for interoperability, trust, transparency, and market position of online platforms. The DSA ensures a level playing field for innovation, growth, and competition. All are designed to protect the interests and rights of citizens while encouraging industrial and technological development (European Commission, 2020b).

The Data Act aims to make data clear and visible for both users and other parties. By applying central formats to data and providing access to it, evidence-based decisions can be made. Moreover, the transparency of the process potentially allows users better insight into the potential of their data in enhancing urban planning and policymaking. Additionally, the Act focuses on updated contracts to facilitate data access, sharing, and use, and to counteract market dominance by large companies. This is achieved by making data sharing easier and more secure through fair, reasonable, and non-discriminatory terms. This ensures not only better innovation and economic growth but also preserves the rights of companies and stimulates a competitive market. The Data Act ensures that the compensation for data access does not exceed the costs directly associated with making the data available. This consideration supports the innovation ecosystem by enabling smaller players to more



fully participate in the data economy, promoting diversity and competition (European Parliament & Council of the European Union, 2023).

In 2023, the Data Governance Act went into effect, which aims to build trust in data sharing, and data availability and break technical barriers (European Commission, 2024b). In addition, the AI Act and the Data Act are under development. The AI Act assesses risks associated with the use of algorithms to generate trust in this development (European Commission, 2023b). The Data Act aims to facilitate the collection of data by private parties and access to the data by those from whom it has been collected (European commission, 2022)

Dutch digitalization strategy

The Dutch Digitization Strategy describes the increasing reliance on collaborations among various stakeholders such as businesses, governments, knowledge institutions, and societal organizations to optimize opportunities (Ministerie van Economische Zaken en Klimaat, 2018). It creates space for companies to deploy and test innovations that can address Dutch societal problems and maintain the Netherlands' position as a digital frontrunner in Europe. Here, it emphasizes trust, which further translates into security, privacy protection, and fairness in digital data (Ministerie van Economische Zaken en Klimaat, 2018).

Echoing the European data and digitalization strategy, in 2018, the Dutch government published the Digitalization Strategy (DS). To map out digitalization across various public sector levels, the Digital Government Agenda (DGA) was established. This agenda also mentions the Data Agenda Government (DAG). In this agenda, the Dutch government indicates that it is looking for opportunities to optimally utilize data for policy-making and providing solutions to social issues (Rijksoverheid, 2019b). This agenda highlights the opportunities for new technologies and innovations to contribute to data-driven action. Implementing this is the responsibility of both national and local governments, such as municipalities (Rijksoverheid, 2020b). In 2021, the Dutch government released an update to this strategy that makes it clear that the government is using digital resources for development, solving societal problems, and improved government services (Rijksoverheid, 2021a). In addition to the DAG, the individual vision of each municipality is also important. The Association of Netherlands Municipalities (VNG) is responsible for developing data-driven development goals and visions of municipalities in the Netherlands (Rijksoverheid, 2019b).

Within the Netherlands, since 2019, The Ministry of the Interior and Kingdom Relations and the VNG are responsible for the approved use of data coming from the public domain and for taking stock of the complications in the use of this data. The VNG should also come up with solutions to these complications based on the inventory of the different municipalities (Rijksoverheid, 2019b). In addition, the Ministry of Economic Affairs and Climate Policy developed the Dutch vision on data sharing between companies. This document, unlike Data Agenda Government, is aimed purely at the business sector (Rijksoverheid, 2019a).

For using digital tools and data to enhance urban environments and the development of smart cities, the "NL DIGIbeter 2020" program has been developed. Through Public-Private Partnerships (PPPs), the Dutch government aims to promote innovation in smart city projects. These partnerships involve sharing data and insights to enable more informed decision-making and to develop holistic solutions that meet the needs of various stakeholders (Rijksoverheid, 2020a). To get to optimal use of digitization to achieve economic and societal goals the Data Sharing Coalition was officially launched in the Netherlands in 2020 (Brakema, 2020). To ensure ownership of data and conditions for data use by the respective party, the Data Sharing Coalition is actively engaged in establishing general



agreements for data sharing, testing the use of data in practical cases, and supporting data sharing development and initiatives (Nederland Digitaal, 2021).

Dutch data strategy

The Inter-Governmental Data Strategy (IBDS) is a policy that aims to use data responsibly to solve societal problems and contributes to the "Work on Implementation" (WoI) and the "Value-Driven Digitalization" work agenda. Since 2023, an inter-governmental multi-year plan has been established to work responsibly with data within the government based on use cases with societal value. Through cooperation, data can be utilized from various perspectives to influence the four central pillars: What is Allowed, What is Possible, What Helps, and What Inspires (NL DIGITAAL: Interbestuurlijke Datastrategie Nederland, 2021; van den Berg et al., 2023). In the multi-year approach of the IBDS 2024, there is an effort to follow up on innovations and new developments concerning data and to identify needs within the context of data sharing and data usage. This should collectively lead to leveraging and guiding innovations to create a solid trust framework for data sharing. To this end, it focuses on increased use of data and data usage in consultation with public values (Rijksoverheid, 2023).



12. APPENDIX B – (LAW AND REGULATIONS IN DATA-DRIVEN DECISION-MAKING)

The EU promotes but limits data-driven decision-making at the same time. To begin with, the EU is promoting the Digital Single Market for Europe, which aims to ensure that the economy, business, and society in Europe take full advantage of the new digital era. Unity between different stakeholders on different policy/geographic levels in the field of digital processes, technologies, and data use aims to innovate and increase acceptance of change (Ratcliff et al., 2023). Additionally, the EU Commission has developed the Public Sector Information (PSI) guideline, which is also known as the directive on open data, with a specific focus on open data from public enterprises to make data more accessible and further used for development. The directive aims to prescribe a transparent process for public-private data sharing to avoid risks such as data lock-in. Thus, it seeks to remove barriers in the reuse of public sector information and includes scopes for obtaining and understanding dynamic data to facilitate real-time services. These high-value datasets include geospatial, environmental, meteorological, statistical, mobility, and business datasets because they have great potential commercial value. Obtaining data is partly done by engaging especially SMEs in data sharing through financial incentives and organizing EU data spaces to stimulate access and interoperability and bring about data-driven innovation (European Commission, n.d.).

City Deal

The City Deal focuses on creating smart, sustainable, and democratic cities in the Netherlands, rooted in a community of practice framework. A broad coalition of partners collaborates to seize opportunities related to digitalization and technology, with a strong emphasis on ethical principles and democratic decision-making. Innovation in digitalization and technology is leveraged to address significant societal challenges, including housing shortages, mobility pressures, energy transition, climate change, and nitrogen issues. Furthermore, the City Deal serves as a platform for collaboration, knowledge sharing, and innovation, providing smart city solutions(Wesselink, 2024; Wesselink, n.d.).

Gemeentewet

The Municipal Law forms the foundation for municipal government, facilitating decentralization from the national government to local municipalities. This autonomy allows local governments to make individual choices for urban development and respond to local needs. The law stipulates that the municipal council establishes the frameworks for urban planners and policymakers, and accomplishes zoning plans, budget allocations, and project approvals. Urban planners and policymakers are required to work closely with the executive board (mayor and aldermen) to implement plans and operate within these frameworks. The Municipal Law encourages citizen participation and involvement in the planning and decision-making process. Plans must address not only technical and social aspects but also financial feasibility within the municipality and provide accountability for this.

Although the Municipal Law does not directly address decision-making based on data from private parties, it does imply certain uses of data within the legal framework. The decentralized autonomy of municipalities allows them to form their policies and decisions and to determine how they collect, analyze, and apply data for urban development (within legal constraints). The principles of transparency and accountability from the law also apply to the use of data in decision-making processes. Collaboration with citizens in the realm of data and transparency can enhance the acceptance of policies among residents.

The openness of the law enables the formation of collaborations with various stakeholders for project implementation, as well as data-driven partnerships with private entities. However, the conditions



within these collaborations must be transparent and serve the public interest (Rijksoverheid, 2016; VNG, n.d.).

Handreiking Data-gedreven werken VNG

Data-driven work leads to opportunities for reliability and quality in policy development, but public values and digital civil rights must be respected. The Association of Dutch Municipalities (VNG) emphasizes the achievement of goals for 2030 in its guidance on Data-Driven Working. These goals include responsible data use and municipal participation in the VNG Knowledge Network for Data & Society. The aim is to share effective information and experiences based on practical cases among municipalities to identify successful processes and optimize various policy processes. By gathering information from different municipalities, the VNG can build on datasets, knowledge, and experiences to map the impact of national laws and regulations, centralizing the interests of the municipalities (Nijman & Alberts-de Gier, 2023).

GDPR

However, the Dutch government acknowledges complications in data-sharing processes (Rijksoverheid, 2019b). The main data sharing limiting regulation is the General Data Protection Regulation (GDPR), which exists to protect the individual rights of people and to simplify and standardize data use within the EU. It focuses on giving individuals control over their data. This approach not only aims to modernize data handling practices but also to foster consumer trust through reduced bureaucracy and enhanced privacy measures. Both companies and governments are required to handle user and resident data carefully by complying with safety and privacy requirements. The GDPR sets innovation as a goal of data use and emphasizes the realization of privacy by design, which means that privacy is the starting point of every good and service (European Commission, 2016; European Commission, n.d.-a). Since governments are not the only actors and stakeholders in the public domain, it is necessary to set certain boundaries regarding the collection and use of data in agreement with society and businesses (Rijksoverheid, 2019b)

Decentralization and data sovereignty

In addition, the Future Exploration of Digitalization 2030 identifies 11 trends within digitalization in the Netherlands. Among these, the development of Mega Ecosystems, also known as multifunctional apps by large tech companies, raises concerns about centralization, privacy, and security. For this reason, there is a surge in technological decentralization, allowing data, finance, and applications to operate without direct control by a central (technological) power.

Data sovereignty is a growing trend where citizens, companies, and governments are becoming more aware of their data and its sharing, particularly with the rise of Artificial Intelligence (AI). This has raised concerns about data responsibility, especially as AI is increasingly integrated into daily tasks. Digitalization and smarter technology enable people to group digitally and transmit information through platforms, resulting in less direct government oversight. Humans are increasingly dependent on technology for optimization and economic growth, but this has led to increased vulnerabilities on various levels, from individuals to society as powers aim to spread ideas and stimulate technology development (Rijksoverheid, 2021c).



13. APPENDIX C – (MAAS FINDINGS)

Although the rise of micro-mobility is often prohibited by the Dutch government to date, MaaS is seen as an opportunity to use new mobility options to achieve public social goals. New Mobility Services have increasingly been added to the Dutch transportation system, and new modalities have been integrated into travel planner apps. The number of stakeholders involved in MaaS is significantly large, and the diverse nature of these stakeholders makes collaboration complex but essential for urban development. Although most public transport companies are exploring MaaS options, governments often do not provide MaaS services directly but collaborate with private players and encourage growth (TNO, 2020). The difficulty lies in the separate silos in which MaaS is conceived and developed (ER1; ER4). Rik Braams from TNO and the Ministry of I&W (ER2) highlights the challenge of this integration: *"Imagine if you could bring all those apps and all those mobility providers together on one platform; that would be quite a challenge."* Furthermore, Emma de Wijs from municipality The Hague (MR2) emphasizes the environmental benefits, stating, *"Clean transportation in the city is an important task in which Mobility as a Service can be key and connects to the goals of the municipality."*

MaaS was presented as the solution for traveling from A to B, regardless of the modes of transportation used (Van De Wiel, 2023). As Emma de Wijs from municipality The Hague (MR2) notes: *"You have one way to travel, but how you get there can be with different providers."* The generated travel data from the MaaS trip would be immediately analyzed to contribute to sustainability goals, solve traffic congestion issues, and reduce pressure on public transport and accessibility. However, complications arose during its development. It turned out to be more challenging than anticipated, usage was lower than expected, and the concept resolved fewer problems than anticipated. Even when shared mobility providers were already integrated with MaaS, they withdrew because it did not work as promised and caused more problems (Van De Wiel, 2023). Table 4 below adds a comprehensive overview of current challenges and findings about MaaS and its integration into the Dutch mobility system.

A MaaS app might be financially challenging for providers	Connection to MaaS is difficult for car-sharing providers
Maas cannot be developed by a single party	Planning ahead and making reservations makes MaaS integration more difficult
Concession boundaries	MaaS has privacy issues
MaaS takes hurdles with downloading, onboarding etc.	MaaS is not really needed
Integrating standards other than their mobility providers' own technological development in MaaS and shared transportation costs too much money	The answer for facilitating and developing MaaS has not yet been found
MaaS experiences struggles with integration of many different mobility providers	The municipality is not really sure yet how it wants to facilitate or drive MaaS
MaaS is only a small share of total mobility	MaaS is not yet successful as it has high fees
MaaS is still too often in separate silos	MaaS is difficult for a government to market

Table 4 MaaS challenges



Most people are not yet familiar with MaaS	MaaS lacks in operationable scale
Maas development takes collaboration	The accountability aspect is a major barrier to MaaS
Native apps are better than MaaS apps	The challenge of MaaS is organizational
People want mulitple apps instead of one central MaaS app	The impact of MaaS is uncertain
MaaS apps are only used to book a ride to a limited extent	

The user experience through a MaaS app is always inferior and less reliable than in the native app. Developments in the providers' own apps could only be made available later in the MaaS app, causing them to quickly fall behind the competition. The development costs for MaaS are high, and if no added value is delivered, partnerships quickly dissolve (Van De Wiel, 2023). Gemma Schepers from municipality Amsterdam (MR7) notes: *"Most people download an app from the provider anyway, so the Check app, or the Felix app, or the Greenwheels app and don't book the trip through MaaS,*

because it's a bit cumbersome anyway. It doesn't work that well yet."

The expectation is that MaaS can contribute to urban developments when challenges regarding data use and privacy are addressed, but therefore is a need for both public and private data. The Netherlands directly targets achieving MaaS level 4, which means that data from public-private partnerships is used to guide policy objectives and regulations that have been developed for MaaS (see Figure 5) (Bollars et al., 2021).



Figure 5 Levels of MaaS (TNO, 2020)

On the other hand, MaaS apps are effective for facilitating billing, integrating with administrative systems, and registering and deregistering users. Based on data, the app serves excellently as a map or travel planner in which mobility supply and demand are efficiently and effectively structured (CoE-DSC, 2023), but users prefer the ability to compare different mobility providers. This was confirmed by Jeanette van Eijk from Greenwheels (IR3), noting: *"Identifying those vehicles is going well, but when it comes to reporting damage, indicating where someone has parked the car, or wanting to know how much fuel is in the car or the battery level if it's electric, it would be convenient if we could show that in our own app and continue to develop it further."* Additionally, the app must be at least as easy to use as the native apps (ER2). The high level of required personal data also creates a barrier to usage (Van De Wiel, 2023). Therefore, behavioral change is needed in various ways. If people are willing and able to travel more flexibly, it could significantly impact MaaS usage. The responsibility for the functioning of MaaS lies with the service provider, who must ensure it meets the needs of the intended audience (Van De Wiel, 2023). Additional use purposes and Maas opportunities as posed in Table 5.

Table 5 MaaS use purposes

Municipalities	Industry representatives	Researchers & experts



MaaS as a route planner	As a map of current vehicles	A solution for the first and last mile of a journey.
MaaS as a solution for space constraints	To facilitate the customer	Facilitating international transport
MaaS as an opportunity for clean transportation		Facilitating payments
MaaS for vulnerable groups		Facilitating seamlessness for stimulating shared mobility use
MaaS in non-urban areas can be a solution for the lack of train and metro services.		Identifying accessible vehicles and modalities
MaaS is a way to offer transportation to various groups in an intelligent manner.		Solution for an unplanned trip
MaaS to achieve sustainability goals		Tailored travel advice
People need an incentive to use MaaS.		
First and last mile transport is from the city outskirts to the center and vice versa.		
A complete MaaS trip consists of small segments before and after transport, integrated with public transit.		

MaaS in practice

In 2017, a whitepaper was published by the Ministry of Infrastructure and Water Management (Ministry of IenW), outlining the definitions and models of MaaS. Simultaneously, Dutch governments collectively set up the MaaS program, showing public-private cooperation for 7 pilot initiatives to understand the impact of MaaS in urban environments. In 2018, budgets for these initiatives were set, with co-financing totaling 20 million euros. This involves municipalities imposing strict regulations on new and shared mobility based on fair competition and protection of consumer interests. Private parties must contribute 50% of the capital needed themselves and to present a business model that aims to be (financially) self-sustaining within 2 years. Additionally, the use of the data string, collaboration, and as much use as possible of the Transport Operator to Mobility/MaaS Provider API (TOMP-API) is mandatory. The data enters an open ecosystem to explore insights from practice and develop proper regulations (TNO, 2020).



14. APPENDIX D – (INITIAL LITERATURE REVIEW)

To fit the research within the larger context of existing knowledge and literature, this chapter describes the current state of the literature through a systematic review. In this way, literature gaps can be identified and important theories and concepts can be incorporated into the theoretical framework. This literature was found based on keywords entered in the Scopus database. Here, the time criteria were set on 10 years of research, from 2014 until 2024. However, as this research has been built in response to the Uber Movement Initiative, the literature search has been specified for Keywords mentioning "Uber" to exclude the timeframe in which Uber Movement was not yet available and thus may lead to different research findings about Uber as a ride-hailing company in the city. The search term for the development of cities is additionally framed in the Netherlands, where the goal is to gather specific insights into the development plans of Dutch cities.

Table 6 shows the main concepts and corresponding search code. It also explains why these concepts are important to the study and how many results were found and selected in the literature review. After this, the full and elaborated literature review is presented in which the main concepts for understanding the research topic become clear and research gaps are being identified.

Main concepts	Search code	Reasoning of use	Search results	Selected
Urban mobility	TITLE-ABS-KEY ("urban mobility" AND "dutch" AN D data)	To examine current practices regarding movements of people within urban environments in Dutch cities and lay a foundation for applying data in urban mobility.	2	2
Smart and sustainable cities in the Netherlands	TITLE-ABS-KEY ("smart cities" AND "netherlands" AND "sustainability")	To discover the intersections and connections between smart cities and sustainability in the Netherlands and understand how Dutch cities use technology and data for efficient urban development.	23	8
Sustainable urban planning	TITLE-ABS- KEY ("sustainable urban planning" AND dutch OR n etherlands)	To understand the broader Dutch urban planning framework and discover environmental and social sustainability in urban planning.	6	1
Platform/ride -hailing data	TITLE-ABS-KEY ("ride- hailing data" OR "platform data" AND "urban transportation" OR mobility	To find out how platform/ride-hailing data influences urban planning transportation and mobility.	24	4
Data collaborative in the city	TITLE-ABS-KEY ("DATA COLLABORATIVE") AND CITY	To discover which and how data collaboratives are currently taking place in cities.	32	6

Table 6 Search keywords and results



Data partnership	TITLE-ABS-KEY ("DATA PARTNERSHIP")	To discover which and how data partnerships are currently taking place in cities.	26	3
Business to government data sharing	TITLE-ABS-KEY ("BUSINESS TO GOVERNMENT DATA SHARING") TITLE-ABS-KEY (B2G OR BUSINESS-TO- GOVERNMENT AND "DATA SHARING")	To explore how data sharing between businesses and government is formed and what is involved.	4 / 16	3/8
Urban planning and policymakin g	TITLE-ABS-KEY ("urban planning" OR policymakin g AND "data analytics" AND transportation OR mobility)	To explore the use of data analytics in urban planning and policymaking, specifically focused on the transportation and mobility sector.	39	4
Platform/ride -haling policymakin g and urban planning	TITLE-ABS- KEY ("transportation" OR mobility AND policy OR pl anning AND "ride- hailing" OR platform AND municipality)	To explore the ways in which municipalities involve platform-based services in urban planning or policymaking developments.	28	2
Data driven urban planning and policymakin g	TITLE-ABS-KEY ("data- driven" AND "urban planning" OR policymakin g AND "netherlands")	To explore the role of data in urban planning and policymaking processes in the Netherlands.	8	2
Uber Movement	TITLE-ABS-KEY ("UBER MOVEMENT")	To uncover how ride-hailing companies can potentially share data, as Uber Movement is seen as the largest and most successful data sharing innitiative	22	6
Uber and local governments	TITLE-ABS-KEY (UBER " LOCAL GOVERNMENTS") TITLE-ABS-KEY (UBER AND PLANNING AND DATA AND CITY OR URBAN)	To discover how Uber, by being the largest ride-hailing company, is being taken on by local governments	15 / 54	9 / 10

Smart and Sustainable Cities in the Netherlands

The integration of smart city governance into traditional systems may lead to challenges that require sustainable and multi-stakeholder participation strategies (Nesti, 2020). The changes in governance structures provide opportunities for collaboration and inclusive approaches in which stakeholders can



contribute to city developments. In this development, the use of data is an opportunity for smart and sustainable development (Yigitcanlar et al., 2019). Similarly, D. E. Mills et al. (2021) recognize the importance of authentic collaboration between urban governments, private organizations and citizens for achieving urban development goals. (Paskaleva et al., 2021b) add academia as a fourth stakeholder and suggest that co-production with diverse perspectives leads to more holistic solutions and could be the key to effectively leveraging data for urban planning and policy making, using the Quadruple Helix model to analyze projects in cities such as Manchester, Eindhoven and Stavanger.

Research by Tomor (2019) introduces the concept of "Citipreneurship" in which traditional public sector management is combined with strategies in smart city contexts. The unique position of these entrepreneurs can provide a link between civil society, the market, and the state to create public value by using smart technologies (Tomor, 2019). Based on four different smart city models in different urban contexts, focusing on socio-cultural and organizational processes in urban development, Lee et al. (2023) add the role of integrated platforms in facilitating collaborative governance in local contexts.

The evolution of governance structures within smart city projects can be divided into several stages (Ooms et al., 2020). In the initiative stage, factors such as leadership, strategy, and community building are crucial. As development continues, it enters the growth phase. In which the focus shifts to managing larger numbers of participants and increasing competition, with an emphasis on collaboration and co-creation strategies. In the maturity phase, ecosystem management returns to orchestration, but with a focus on control and value appropriation (Ooms et al., 2020).

To assess compliance with Sustainable Development Goal 11 (SDG 11), (Parra-Domínguez et al. (2023) introduced a fuzzy logic-based model. This includes data availability and data quality challenges, highlighting the potential of this model for a better understanding of urban smart and sustainability performance. The findings suggest that this fuzzy logic approach can effectively capture the complexity of urban sustainability and provide insight into areas where cities need improvement. However, the research has limitations, such as the potential impact of missing data on assessment results and the subjectivity involved in designing membership functions (Parra-Domínguez et al., 2023).

Urban Mobility

The complex policy mixes within governments, specifically in the context of sustainable urban mobility transitions in Dutch cities have been examined by Liu et al., (2024). The authors acknowledge the importance of cooperation between different levels of government and other stakeholders and the need for European funding for local initiatives. Based on case studies in Maastricht and Groningen, Liu & Dijk (2022) highlight the importance of using data in short-term policy cycles for traffic regulation and policy adjustments, emphasizing the need for more competent urban planners and policymakers to interpret data and the importance of data for promoting smart and sustainable mobility. Specifically, to achieve traffic adaptation, there is a greater need for real-time traffic and mobility-related data in short-term policy cycles (X. Liu & Dijk, 2022).

Urban Planning and Policymaking

França et al. (2021) and Wang et al. (2021) emphasize the role of interconnected and data-driven systems in enhancing urban transportation systems. They highlight how leveraging data and integrating it with smart technologies can innovate urban living and management. However, understanding challenges such as data isolation and limited knowledge derivation from data is essential (A. Wang et al., 2021; Sarwat, 2015). Sarwat (2015) and Silva et al. (2018) underscore the



importance of data management techniques like IoT to develop effective smart cities that efficiently manage and process large amounts of urban data, leading to a better understanding of urban mobility patterns and more informed urban planning and management. By leveraging the power of big data, cities can make more informed, timely, and effective decisions, leading to smarter and more sustainable urban environments (Sarwat, 2015; Silva et al., 2018).

Sustainable Urban Planning

The application of Information and Communication Technology (ICT) tools in urban planning for sustainable energy integration is investigated by Oregi et al. (2015). Based on a city expansion plan in Nijmegen, The Netherlands, the authors found that the use of ICT tools can significantly improve sustainability performance and facilitate stakeholder communication. However, conflicts and synergies exist between different policy levels, and better policy alignment for sustainable urban mobility still needs to be addressed (Oregi et al., 2015).

Data collaboration

Data partnerships enable urban planners and policymakers to access more data and expertise, allowing them to make evidence-based decisions to improve transparency, accountability, and effectiveness (Hawken et al., 2020). The integration of data-driven approaches is essential in developing smart and sustainable cities (Hudović Kljuno & Krivošić Dizdarević, 2021). Different datasets offer insights into urban challenges, enabling more effective policy formulations and help address urban challenges and stimulate innovation and economic growth, especially in developing countries (Hawken et al., 2020). Evidence-based decision-making allows for appropriate responses to emergencies, transportation, and environmental management (Susha et al., 2017). New data promotes direct decisions and democratic decision-making through increased citizen engagement (Muñoz & Bolívar, 2021).

Although Klievink, Van Der Voort, et al., (2018) do not directly address collaboration structures in urban environments, they do indicate that successful data collaboratives involve trustful collaborations and active stakeholder engagement. Better collaboration between users and data providers enhances the positive impact of data and trust, enhancing the legitimacy of this practice (Verhulst, 2021). These partnerships improve decision-making, optimize services, and enhance efficiency (Rasche et al., 2021; Hawken et al., 2020; Susha et al., 2019). By providing access to new data and resources, these collaborations help the decision-making process and integrate public participation (Rasche et al., 2021; Bednarska-Olejniczak et al., 2019). Public-Private Partnership (PPP) and the Quadruple Helix model are examples of collaborations that combine joint efforts to achieve urban development goals (Brinkerhoff & Brinkerhoff, 2011; Paskaleva et al., 2021).

Data collaborations also present challenges like data privacy and changing governance frameworks (Swofford, 2020; Borghys et al., 2020). Here, citizen engagement, public participation, and inclusivity are essential (Muñoz & Bolívar, 2021; Bednarska-Olejniczak et al., 2019). Strong political leadership in city governance, stakeholder collaboration, and enough funding can lead to data-informed decision-making (Hawken et al., 2020; Paskaleva et al., 2021). Hudović Kljuno & Krivošić Dizdarević, (2021) and Zhang, (2019) indicate that challenges in this process are associated with financial difficulties, administrative complexities, and data privacy concerns. To overcome these challenges, Paskaleva et al., (2021) indicate that innovation and ethical technology usage need support from regulatory frameworks.

Types of Data in City Data Collaborations

Evidence-based decision-making relies on data collaboration, including transportation, environmental, energy, and social data (Hawken et al., 2020). This data covers city functions like traffic, utilities,



water supply, waste management, and community services (Hudović Kljuno & Krivošić Dizdarević, 2021). E-participation initiatives help understand civic needs and preferences through social media and citizen feedback (Muñoz & Bolívar, 2021). Data on citizen well-being also plays a role in data collaborations, analyzing the impact of urban environments on mental and physical well-being (Geropanta et al., 2021).

Benefits and Risks of Data Sharing

B2G data sharing benefits public authorities by improving the quality and quantity of data (Vigorito, 2022; Rasche et al., 2021). According to Borghys et al., (2020); Hawken et al., (2020); Susha et al., (2019), the benefits of data collaboration in city developments can be seen in improved decisionmaking, overall increased process efficiency, enhanced citizen engagement, and innovation. Data collaboration in city developments can influence situational awareness, analyze causes, problems, and effects, and make connections. By combining different data sets, a global overview of a case can be provided, ensuring that those responsible for the solution have insights into the causes of problems they need to solve. Data can also improve predictive capability by analyzing previously unavailable data sets, enabling institutions to proactively respond to potential changes and crises. Finally, data can reflect the actual impact of policies and interventions, allowing for evidence-based development of policies and services through an iterative process and experimental methodology (Verhulst, 2021).

Data sharing is a crucial aspect of reducing social and economic disparities (Vigorito, 2022), but it also presents risks such as privacy, security, and potential increased social and economic disparities. Clear governance frameworks and regulatory frameworks are necessary to guide data sharing and achieve a balance between public and private sector interests (Zhang, 2019; Ruijer, 2021; Hawken et al., 2020; Susha et al., 2017). Strong leadership, stakeholder engagement, and adequate resources are essential for reducing inequalities (Hawken et al., 2020; Zhang, 2019). Companies must be cautious of over-compliance with information-sharing regulations, as this can lead to unintended consequences. Striking a balance between compliance and value creation is crucial (Klievink, Janssen, et al., 2018). The primary issue is the mismatch between data supply and demand, which can be addressed through data collaboratives (Verhulst et al., 2016). By ensuring the right institutions and individuals can analyze and use data, new innovative social solutions can be discovered (Verhulst, 2021).

Business-to-Government Data Sharing

Businesses could contribute to the development of smart and sustainable cities (Hudović Kljuno & Krivošić Dizdarević, 2021). However, this does not exist without risks for the companies. Privacy and potential misuse of data could damage trust in private businesses by consumers (Burns & Andrucki, 2021). For this reason, there is an immediate demand for strong governance frameworks to handle data sharing ethically (Bian, 2023).

Collaborative success relies on a shared vision, commitment, and mutual trust among stakeholders (Susha et al., 2022; Hawken et al., 2020). However, challenges include effective governance structures, transparency, financial resources, and administrative complexity (Bednarska-Olejniczak et al., 2019; Hudović Kljuno & Krivošić Dizdarević, 2021). Burns & Andrucki, (2021) indicate that citizen engagement and social inclusivity are crucial for successful partnerships, but more research is needed. Ethical issues like data ownership, privacy, and security pose significant obstacles in data collaborations, necessitating further research (Ruijer, 2021).

Private businesses can benefit from data sharing with government agencies when they align their interests with societal needs. This can be achieved by balancing government leadership with businesses' interests (Susha et al., 2022; George et al., 2020). However, data philanthropy requires



significant control over complementary assets like data expertise (George et al., 2020). Businesses must carefully consider the information they want to share, the conditions for sharing, and how to add value while considering the pros and cons of data sharing (Klievink, Janssen, et al., 2018).

Platform/Ride-Hailing Data

Despite only focusing on Toronto, Canada, Loa et al. (2020) reveal that ride-hailing and public transit demand are more complementary than substitutive and suggest that the demand for one service tends to increase the demand for the other. Yang et al. (2022) propose an innovative approach to utilizing diverse mobility data sources, including ride-hailing data, which can potentially significantly improve urban planning and policy-making. Through a multi-source data-driven methodology for delineating Traffic Analysis Zones (TAZs), Yang et al. (2022) propose a way to offer valuable insights into urban transportation planning. However, its specific application to Beijing highlights potential limitations in different urban contexts.

Big data can also forecast demand for on-demand ride services. Through data from DiDi's platform in Hangzhou, China, J. Liu et al. (2017) examine the effectiveness of the random forest model in predicting ride service demand. This approach aligns with the dynamic nature of urban transportation and could be instrumental for real-time policy decisions in cities. Also, Romano (2021) explores the role of platform data in understanding changes in mobility patterns during and after lockdowns, highlighting the critical role of such data in managing urban mobility during crises. However, due to its focus on pandemic-related mobility patterns, it may not fully represent regular conditions.

Platform/ride-hailing policymaking and urban planning

Fiore et al. (2019) show how advanced data management and analytics can contribute to smart city development, specifically focusing on transportation management. By using cloud-based big data platforms and a mix of technologies, focusing on data privacy, data quality and the integration of different data sources, more informed decisions and efficient urban mobility planning can be achieved. Another way to use big data analytics in urban traffic management has been explored by Bakri et al. (2022). By developing a smart transportation platform that uses big data to address transportation challenges in historic megacities, the authors highlight the potential of big data analytics to provide real-time insights into traffic conditions, enabling more efficient and responsive traffic management. However, the study's specific focus on historic megacities may limit its applicability to other urban contexts.

Data-Driven Urban Planning and Policymaking

Diran et al. (2022) examine the adoption of data-driven approaches in local policymaking, focusing on local energy transitions within four Dutch municipalities: Amsterdam, Rotterdam, The Hague, and Utrecht. They identify a significant gap between the potential applications of data-driven methods and current practices, underlined by challenges related to data availability, technical capabilities, and readiness of institutions. Despite the value of data-driven approaches, barriers such as the integration of these methods into local policymaking persist. Dutch municipalities should navigate these challenges to harness data effectively for urban planning and policymaking. For this, the authors advocate for future research to delve into integrated and actionable strategies for the adoption of data-driven applications (Diran et al., 2022). Complementing this perspective, Diran & van Veenstra (2020) explore the specific challenges encountered by Dutch cities in implementing data-driven policymaking for urban heat transition. Their findings, based on eight Dutch cities, illuminate barriers including GDPR (General Data Protection Regulation) and privacy concerns, the substantial investment of time and resources, scattered data distribution, and data processing and analysis challenges. This detailed



examination of challenges and barriers contributes to a broader understanding of the impediments to data-driven policymaking in urban settings.

Uber and Local Governments

Big internet businesses use vast amounts of geospatial data to improve their functioning and financial growth. However, public benefits are often overlooked, and governments are slow to recognize the city's ownership of this data (Hawken et al., 2020). To address global challenges, new guidelines and rationales need to be established for data sharing between governments and companies (Mayer-Schoneberger and Ramge 2018). Mayer-Schoneberger and Ramge (2018) argue that large companies should pay taxes through data sharing, rather than monetary taxation, to work together and address global challenges (Hawken et al., 2020).

Route planning relies heavily on time-dependent traffic speeds, but many OpenStreetMap (OSM) platforms lack real-time or historical data. One reason for this is that commercial companies like Google and Here own these global traffic speed data. However, combining different data sets is challenging due to different data system structures. Uber Movement, released by Uber, is the most promising open data set for open-source routing systems so far (Ludwig et al., 2023).

Understanding Uber Movement Data

Uber Movement Data is an initiative that releases anonymized GPS data from millions of Uber trips and users worldwide, providing insights into transportation patterns across major cities through information about trip origins, destinations, travel times, and other statistics while securing user privacy (Sun et al., 2020; Bezerra et al., 2019; Aryandoust et al., 2019; Roy et al., 2020; Perlman & Roy, 2021; Ludwig et al., 2023; Neun et al., 2023). The initiative was created to support urban planners and policymakers in making informed decisions about transportation, infrastructure development, and urban planning (Bezerra et al., 2019; Aryandoust et al., 2019; Roy et al., 2020). While the literature on Uber's data provision is sparse, several articles suggest that Uber reveals valuable information about commute times, job accessibility measurements, traffic patterns, and congestion spots, which are valuable to urban planners (Gerte et al., 2018; Sun et al., 2020; Aryandoust et al., 2019; Roy et al., 2020; Perlman & Roy, 2021).This data helps local governments improve transportation planning, enforce regulations, and make evidence-based decisions about urban development initiatives for creating smart and sustainable cities (Cohen, 2018; Hawken et al., 2020).

Local Governments' Engagement with (Uber) Data

Uber's data is being used by local governments to improve urban planning and policymaking. This includes understanding urban mobility and analyzing driver behavior to tailor regulations (Cohen, 2018). Data-sharing partnerships are essential to illuminate traffic patterns, congestion spots, and regulation compliance (Hawken et al., 2020). The data is used to create parking density maps, assess socio-demographic influences, and inform Uber's service connectivity in different regions (Aryandoust et al., 2019; Perlman & Roy, 2021). Understanding the spatiotemporal distribution of Uber trips can improve public transportation options, reduce traffic congestion, and identify areas for economic growth and development (Roy et al., 2020; Sun et al., 2020). Additionally, data optimization of road networks, parking pricing, and overall transportation efficiency is being considered (Aryandoust et al., 2019).

The relationship between Uber and local governments is complex, with concerns about data privacy, security, balancing business innovation and worker protection, and conflicts of interest between Uber and city authorities (Benli-Trichet & Kübler, 2022). Uber's disruptive innovation strategies can challenge traditional governance, but its data can provide valuable insights into city transportation



dynamics. This presents both challenges and opportunities for urban planning and policymaking from local governments' perspectives (Cohen, 2018; Hawken et al., 2020). However, the risk of limiting citizens in urban decision-making through smart engagement is acknowledged, emphasizing the need to balance private interests with community needs (Bednarska-Olejniczak et al., 2019). The use of data for the public good is still in its early stages, and the optimal way to optimize data collaboratives is yet to be discovered (Verhulst, 2021).

Conclusion

The rise of gig economy platforms has led to the collection of (trip) data. However, the release of this data is often restricted due to concerns about privacy and ownership (Cohen, 2018; Burns & Andrucki, 2021). Despite this, urban planners and policymakers are increasingly requesting trip data to inform their plans, aligning to develop smart and sustainable cities (Hawken et al., 2020; Geropanta et al., 2021). This data can be used to make evidence-based decisions in urban planning and policymaking, providing insights into efficient, smart, and sustainable cities. However, challenges like data privacy and public-private partnerships limit its application (Susha et al., 2017; Hudović Kljuno & Krivošić Dizdarević, 2021). To effectively use (trip) data, a balance must be struck between innovation, privacy, and public interest (Hawken et al., 2020; Rasche et al., 2021), where data demand is met with supply (Martin et al., 2018; S. G. Verhulst et al., 2016).

Literature gaps

The literature indicates that urban planning and policymaking are increasingly dependent on data, and while data sharing and public-private partnerships are frequently mentioned, little in-depth research has been done on the challenges, opportunities, and outcomes of these partnerships as well as the specific supply and demand of ride-hailing and shared mobility trip data in the context of smart/sustainable urban development. Especially, the combined perspective and effort of private companies, governments, academia, and citizens in smart urban development may lead to effective outcomes.



15. APPENDIX E – (CROSS-CASE ANALYSIS FOUNDATION)

Table 7 Shared mobility comparison matrix

<u>Categori</u> es:	Twowheelers	Car-sharing
Supply:		
Data type	 9. Road quality (IR1) 10. Commercially sensitive data (IR2) 11. Mis-match data (IR2) 	
Data content	 Depends on agreements (IR1) Everything except privacy and business data (IR1) Average trip distance (IR2) Local road intensity (IR2) Number of trips (IR2) Start- and end location (IR2) Trip duration (IR2) 	 19. Annual averages (IR3) 20. Vehicle averages (IR3) 21. How are the cars being used (IR4) 22. How many cars does the provider have (IR4) 23. How many hours per day does the car drive (IR4) 24. How many people use the car (IR4) 25. Location of cars (IR4) 26. What distance does the car ride (IR4)
Data level	 27. Aggregated data (IR1) 28. Only start and end locations instead of the entire trip is shared (IR1) 29. Raw API (IR1) 30. Data level varies per municipality (IR2) 	 31. As much as possible (IR3) 32. The company does not look at specific vehicles (IR3) 33. Data gets mostly generated outside the municipal borders (IR3) 34. Number of customers per municipality (IR3) 35. Data can easily be retraceable in small markets (IR4) 36. The company has more data than it is willing to share (IR4)
Data sharing mode	 37. API (IR1) 38. Dashboard platform (IR1) 39. Depends on communication and agreement (IR1) 40. License to operate (IR1) 41. Mobility provider analyses (IR1) 42. As part of the license to operate (IR2) 43. Proactive data sharing (IR2) 44. Same as Uber Movement, but not public (IR2) 45. Standard KPI (IR2) 	 46. Proactive (IR3) 47. Voluntarily (IR3) 48. The company has its own data dashboard (IR4) 49. Free data sharing (IR4) 50. Obligated data demand as a basis and other data demand through constructive collaboration (IR4) 51. Voluntary data sharing instead of obligation is way easier and better (IR4) 52. Voluntary data sharing is fine (IR4)



Supply & demand matching		
Facilitati on	 53. Subcontract a data collector (IR1) 54. Mobility provider has data sharing with a third party to enhance and facilitate transportation offerings (IR1) 	 55. CDS-M as technical approaach of programma Natuurlijk Deelmobiliteit (IR4) 56. Company does not use an external dashboard (IR4) 57. CROW and Vianova as data dashboard (IR4) 58. Data dashboard for municipalities to access aggragate data (IR4) 59. Data dashboards only possible when there are many providers (IR4) 60. Natuurlijk Deelmobiliteit as technical solution and legal basis (IR4) 61. Programma Natuurlijk Deelmobiliteit fair data sharing (IR4)
Degree of access	 62. Depends on legal agreements (IR1) 63. Aimed at specific municipalities (IR2) 64. Data sharing in active municipality (IR2) 65. Data sharing in passive municipality (IR2) 	 66. After municipal demand (IR3) 67. Shared mobility provider may not even share live data with municipalities (IR3) 68. Awaiting judge's decision on data sharing (IR4) 69. Companies must react on municipal data-sharing obligations (IR4) 70. One of my main concerns is how we handle the data requests from the municipality (IR4)

Table 8 Municipality comparison matrix

<u>Categori</u> <u>es:</u>	Municipality Amsterdam	Municipality Rotterdam	Municipality The Hague	Municipality Eindhoven
Supply & demand matching :				
Facilitati on	- Communicatio n between part mobility provider and municipality is	- Regional platform for knowledge sharing (Gemeente	- Open Urban Platform (OUP) (Luk, 2022)	



	essential to understand each other's perspective	Rotterdam, 2021) - Facilitating data exchange to	
	 (MR7) Data sharing leads to better dialogue between provider and city (MR7) Dital Wallets may strengthen TOMP API (MR7) Technical connection between supply and demand is through an API (MR7) The municipality uses third 	 improve the efficiency and effectiveness of city processes and services (Bagheri, 2024) Knowledge institutes help municipalities innovate and use data (Maltha et al., 2021) Appstore (MR5) Outsourcing tasks (MR6) 	
	parties to read data (MR7)	Detectorally	
Degree of access	 Should be independent from third party facilitators (Gemeente Amsterdam, 2023) Open data (Royal HaskoningDH V, 2017) Lack of external sources (Gemeente Amsterdam, 2021b) 	 Data should be open to everyone (Meerman, 2023) Big difference in sharing different data types (MR6) Depends on use purpose (MR6) No access to raw data (MR5) External stakeholders (MR5; MR6 Open data (MR5) Real-time (MR6) 	 Many initiatives (Elstgeest, 2022) Municipalities poses many data



	-	Ownership of				would be
		data is an				shared (MR1)
		endless				
		discussion.				
		(MR7)				
Demand:						
Policy	-	Inclusive	-	Data for	-	Road safety
problem		society		accessibility		(Gemeente
_		(Gemeente		(Gemeente		Eindhoven,
		Amsterdam,		Den Haag,		2023)
		2021b)		2020a)	-	Decision-
	-	Better	-	Evidence-		making and
		decision		based		regulation
		making		decision		(Meijer, 2020)
		(Gemeente		making	-	Public tasks
		Amsterdam,		(Luk, 2022)		(Meijer, 2020)
		2023)	-	Data for	-	Insight in
	-	For		gaining		modal split
		identifying		insights into		(Meijer, 2020)
		people's		urban	-	Impact
		behavior and		mobility		assessment
		profile (Royal		(Gemeente		(Gemeente
		HaskoningDH		Den Haag,		Eindnoven,
		V, 2017)		2020b)		2023)
	-	Improving	-	Data for	-	All policy
		quality of life		infrastructur		(MD1)
		(Gemeente		e status		(MICI)
		Amsterdam,		(Gemeente	-	shared
		2023)		Den Haag,		mobility use
	-	For		2020b)		(MR1)
		performing	-	Data for	_	(WICI) Sustainability
		public tasks		public		(MR1)
		(Gemeente		objectives		(MICI)
		Amsterdam,		(Luk, 2022)		
		2021b)	-	Data to		
	-	Determining		create value		
		parking zones		(Gemeente		
		(MR7)		Den Haag,		
	-	Managing		2020a)		
		physical	-	Data to help		
		public space		citizens		
		(MK/)		(Luk, 2022)		
	-	Using data to	-	10 INIORM		
		reduce		oitizers		
		nuisance.		citizens		
		(MR7)		(Gemeente		



			Den Haag, 2020b) - To make smart apps (Gemeente Den Haag, 2020b) - Urban growth and	
			 Gemeente Den Haag, 2020b) Urban space (Gemeente Den Haag, 2020b) To allow 	
			mobility management (MR2) - To collaborate with others (MR4) To facilitate	
			 Forfacturate shared mobility (MR3) To show mobility hub locations (MR3) 	
			- To steer urban development (MR3; MR4)	
Use purpose	 Data basis (Royal HaskoningDH V, 2017) Public value (Gemeente Amsterdam, 2021b) Data sharing can also lead 	 Don't invest in data acquisition if you don't have a goal (Maltha et al., 2021) Issues as the purpose, ownership, and governance of 	 Data sharing for availability and capacity insights (Gemeente Den Haag, 2020b) End goal is essential in 	 No personal data for area development if you don't have a specific purpose (MR1) No use for personal data (MR1



	to unexpected insights or	the platform; development		data demand (Luk, 2022)	-	To identify shared
	objectives.	costs and	-	Only use data for what		mobility use
	- Use case and	data sharing and		you want to	-	To understand
	data query	privacy for		achieve		city dynamics
	reasoning is	Open Urban		(Luk, 2022)		(MR1)
	often still	(Bagheri 2024)	-	Specific		
	(MR7)	- More than just		and clear use		
		mobility		cases are		
		purposes (MR5)		sporadic		
		- To create a		(MR4) To identify		
		product (MR6)	-	urban area		
		1 7		utilization		
				(MR2)		
			-	To increase		
				(MR4)		
Expected	- In the future		_	Connection	-	Data allows
outcome	vehicle speeds			between		for
	should also be			current		opportunities
	adjustable			practices and		(E) (E) stgeest, 2022)
	based on zones (MR7)			data demand $(I_{\rm uk}, 2022)$		
	zones. (witty)		-	Mismatch		
				between data		
				and permits		
				(MR3)		Automatizatio
Continuit	- Based on		-	Closed	-	n (MR1)
У	conditions			Den Haag.	-	Eindhoven
	agreements			2024)		municipality
	(MR7)		-	Open		has ongoing
				(Gemeente		(MR1)
				Den Haag, 2024)	-	Periodically
			-	Shared		(MR1)
				(Gemeente		
				Den Haag,		
			_	2024) Balancing		
Data acquisitic	- Comming to	- Appropriate		data	-	Municipal
n n	(Gemeente	(Bagheri, 2024)		aquisition		(Elstgeest,
	`			strategy with shared		2022)



challenge		Amsterdam,	-	Citizen		mobility	-	Around data
s		2023)		engagement		facilitation		there are not
	-	Gathering data		(Bagheri, 2024)		(MR3)		so many
		is expensive	-	Collective	-	Many		obstacles,
		(Gemeente		ambitions		municipal		everything is
		Amsterdam,		(Bagheri, 2024)		departments		running pretty
		2021b)	-	Cost of data		(MR3)		well (MR1)
	-	Private		platforms	-	No data from	-	Need for a
		companies are		(Bagheri, 2024)		ride-hailing		legal basis
		reluctant in	-	Ethical use		companies		(MR1)
		data sharing		(Maltha et al.,		(MR3)	-	Need for a
		(Royal		2021; Meerman,	-	Not that		specific goal
		HaskoningDH		2023)		many		(MR1)
		V, 2017)	-	Identifying		difficluties		. ,
	-	Security and		supply (Maltha		(MR3)		
		privacy issues		et al., 2021)	-	Organization		
		(Royal	-	Lack in capacity		ally difficult		
		HaskoningDH		and capabilities		(MR4)		
		V, 2017)		(Maltha et al.,	-	Takes effort		
	-	Agreements		2021; Bagheri,		(MR3)		
		(MR7)		n.d.)				
	-	Differences	-	No technical				
		between USA		barriers (de				
		and EU		Lange, 2020)				
		(MR7)	-	Privacy				
	-	Municipalities		(Meerman,				
		lack		2023)				
		knowledge	-	Transparancy				
		and capacity		(Meerman,				
		(MR7)		2023)				
	-	Municipality	-	Trust (Bagheri,				
		lacks data		2024)				
		(MR7)	-	Costs (MR5)				
	-	Technical	-	Data				
		API's (MR7)		dependency				
				(MR5)				
			-	Data sharing				
				depends on				
				conditions				
				(MR5)				
			-	Data sharing				
				takes effort				
				(MR6)				
			-	Getting to a				
				central stake				
				(MR6)				



Data utilizatio n challenge s	Ethicality (Gemeente Amsterdam, 2021b) Technical capacity (Gemeente Amsterdam, 2021b) Data analysis to policy making (MR7) Lack of municipal technical capacity (MR7) Often changes in market development (MR7) Too much data (MR7)	 Ethical use (Meerman, 2023) Hard to recognize data utilization possibilities (Maltha et al., 2021) Lack in capacity and capabilities (Maltha et al., 2021) Long term vision (de Lange, 2020) Only analyzing data is not sufficient (de Lange, 2020) Only analyzing data is not sufficient (de Lange, 2020) Privacy (Meerman, 2023) Transparency (Meerman, 2023) Transparency (Meerman, 2023) Applications are independent from data (MR5) Data readability (MR5)	Data integration and combination (MR4; MR2) Data standardizati on (MR3)	Municipal capacities (Elstgeest, 2022)
		(MR6)		



16. APPENDIX F – (CONSENT FORM AND INFORMATION)

Informed consent form (interview)

In this study, I want to learn about the role of Mobility as a Service/New Mobility Service data in urban planning and policymaking. Participation in this interview is voluntary and you can quit the interview at any time without giving a reason and without penalty. Your answers to the questions will be used by me specifically. I will process your personal data confidentially and in accordance with data protection legislation (the General Data Protection Regulation and Personal Data Act). Please respond to the questions honestly and feel free to say or write anything you like.

[In case of anonymous handling: Everything you say or write will be confidential, and anonymous. This means that I do not ask for your name, and no one will know which respondent said what.]

I confirm that:

- I am satisfied with the received information about the research;
- I have no further questions about the research at this moment;
- I had the opportunity to think carefully about participating in the study;
- I will give an honest answer to the questions asked.

I agree that:

- the data to be collected will be obtained and stored for scientific purposes;
- the collected, completely anonymous, research data can be shared and reused by scientists to answer other research questions;

I understand that:

• I have the right to see the research report afterwards.

Do you agree to participate? o Yes o No



Information Sheet (interview)

Introduction

You are invited to take part in this study on data sharing and collaboration for evidence-based decisionmaking in urban developments. The purpose of the study is to learn about the role of Mobility as a Service/New Mobility Service data in urban planning and policymaking, specifically in the context of smart and sustainable developments in Dutch municipalities.

The study is conducted by Floris van den Dool who is a student in the MSc program Sustainable Business and Innovation at the Department of Sustainable Development, Utrecht University. The study is supervised by Iryna Shusha.

Participation

Your participation in this interview is completely voluntary. You can quit at any time without providing any reason and without any penalty. Your contribution to the study is very valuable to me and I greatly appreciate the time taken to complete this interview. I estimate that it will take approximately 40-50 minutes to complete the interview. The questions will be read out to you by the interviewer. Some of the questions require little time to complete, while other questions might need more careful consideration. Please feel free to skip questions you do not feel comfortable answering. You can also ask the interviewer to clarify or explain questions you find unclear before providing an answer. Your answers will be noted by the interviewer in an answer template. The data you provide will be used for writing a Master thesis report and may be used for other scientific purposes such as a publication in a scientific journal or presentation at academic conferences.

Data protection

The interview is also audio-taped for transcription purposes. The audio recordings will be available to me, the Master's student, and academic supervisors. I will process your data confidentially and in accordance with data protection legislation (the General Data Protection Regulation and Personal Data Act).

[In case audio recordings will be deleted: Audio recordings will be deleted when data collection is finalized and all interviews have been transcribed.]

In case audio recordings will not be deleted: Audio recordings will only be stored on a secured and encrypted server of Utrecht University]

[In case of an anonymous interview: Everything you say in this interview will be confidential and completely anonymous. This means that I will not ask for your name, date of birth, or other personal information that can be traced to you by the University or a third party]. I will process your data confidentially and in accordance with data protection legislation (the General Data Protection Regulation and Personal Data Act)]



17. APPENDIX G – (DESCRIPTION ETHICAL ISSUES)

The acquisition of data, especially primary data, should be done carefully. The rights and privacy of interviewees must be assured in order to conduct a thorough and ethical research. For this reason, this research focuses on informed consent of data subjects, data confidentiality and anonymity, data security and storage, and transparency.

Informed consent

Before acquiring primary data (e.g., through an interview), the subject and purpose of the study is explained. The procedures for data processing and publication are explained, but the importance of acquiring information and the potential benefits of this data are also directly addressed. Through the prepared consent form (Appendix B), data subjects can indicate their understanding and acceptance of the rights and conditions. Data subjects are given the space and possibility to withdraw at any time and have the opportunity to ask for clarification.

Data confidentiality and anonymity

To ensure data confidentiality and anonymity, it is ensured that data is anonymized immediately and there is no possibility of tracing the data back to the interviewee if the respondent preferred to be anonymous. The data is encrypted and the identifying information is securely stored separately from the interview data. In addition, participants have the right to request that the initial interview data will be deleted immediately after data processing.

Data security and storage

The data is only directly accessible to the researcher and the first supervisor. The data is processed confidentially according to the General Data Protection Regulation and Personal Data Act. The data is stored solely on secured and encrypted servers of Utrecht University.

Transparency

Throughout the research, there will be open communication with participants about the research process and progress. Regular updates on data processing will be given and questions and other interests of participants will be included in the research process.



18. APPENDIX H – (INTERVIEW GUIDE MUNICIPALITIES)

Thank you very much for agreeing to participate in this interview. My name is Floris van den Dool, and I am conducting research for my master's thesis for the University of Utrecht's Sustainable Business and Innovation program. This research delves into public-private collaboration in urban planning and policymaking, with a specific focus on the role of Mobility as a Service (MaaS)/New Mobility Service (NMS) data. This includes companies and services that are reshaping urban mobility, such as ride-sharing platforms, electric scooter rentals, and ride-hailing platforms. I'm interested in how their data on usage patterns, service demand, operational challenges, etc. can support more informed urban planning and policy decisions.

Today, we'll discuss the supply, demand, and usage of data for smart urban development within the Dutch context, paying special attention to data held by the private sector. Given the transformative potential of MaaS and NMS in shaping our urban environments, it's crucial to understand how data from providers like Uber, Bolt, and local vehicle-sharing services can inform smarter city planning and policy decisions.

This conversation should take about 40-50 minutes, and with your permission, it will be recorded. I understand the sensitivity of information, especially when it comes to unpublished data or internal insights. Please let me know your preference for confidentiality regarding our discussion. Would you be comfortable with your organization or role being mentioned, or do you prefer a higher degree of anonymity? You will also have the opportunity to review a transcript of this interview to ensure accuracy and comfort with the shared information.

Before we begin, do you have any questions?

Introduction

1. "Can you share your relevant work experience concerning the research topic, especially your role in partnerships with mobility providers and access to their data?

Public Problems

- 2. "How does MaaS/New Mobility Services fit into the vision of smart and sustainable city development in your municipality?"
- 3. "What are the current public (policy) problems within your municipality where you see a potential for NMS data to provide solutions?"

Data Supply, Types, Sources, and Collaboration

- 4. "How is this data shared with your municipality by New Mobility Service providers?"
 - a. "Can you give any specific examples of initiatives/collaborations/practices?"
 - i. "How successful are they and what have been the challenges to get access to the data?"
 - ii. "Was/ is data continuously demanded/ supplied?"
- 5. "What specific types of MaaS/NMS (trip) data are you interested in?"
 - a. "Can you give examples of the data content, such as geographic information, user behaviors, or service demand patterns?"

Demand and Utilization of Data



- 6. "What does the municipality desire to accomplish with Maas/ NMS (trip) data and how is it integrated into urban planning and policy-making?"
- 7. "Can you provide examples of how this data has influenced policy decisions or urban development projects in your municipality?"
 - a. "How and with which data provider was this collaboration formed?"

Challenges in Data Access and Utilization

- 8. "In your experience, what barriers exist in terms of data collaboration and sharing between municipalities and private companies?"
- **9.** "What challenges have you faced in accessing and utilizing data from private MaaS/NMS companies?"

Improving Data Collaboration:

- 10. "From your perspective, how can data-sharing initiatives between new mobility service providers and local governments be improved to better support smart city development?"
 - a. What policies/frameworks/tools are needed for that?"

Ending/Conclusion

These were all the questions regarding the research.

- 11. "Is there anything else you think is important regarding the supply, demand, and/or usage of NMS (trip) data in evidence-based decision-making for (smart and sustainable) urban development?"
- 12. "Do you have any questions for me or any aspects of the study you'd like to discuss further?"

This concludes the interview. Once again, thank you very much for your time and insights. Your contribution is invaluable to this research. Please remember that everything discussed will be treated confidentially, and I'm more than happy to share a transcript of our conversation with you. If you have any further thoughts or questions in the future, please feel free to contact me.


19. APPENDIX I – (INTERVIEW GUIDE RESEARCHERS/EXPERTS)

Thank you very much for agreeing to participate in this interview. My name is Floris van den Dool, and I am conducting research for my master's thesis for the University of Utrecht's Sustainable Business and Innovation program. This research delves into the role of Mobility as a Service (MaaS)/New Mobility Service (NMS) data in urban planning and policymaking. This includes companies and services that are reshaping urban mobility, such as ride-sharing platforms, electric scooter rentals, and ride-hailing platforms. I'm interested in how their data on usage patterns, service demand, operational challenges, etc. can support more informed urban planning and policy decisions.

Today, we'll discuss the supply, demand, and usage of data for smart urban development within Dutch municipalities, paying special attention to data held by the private sector. Given the transformative potential of MaaS and NMS in shaping our urban environments, it's crucial to understand how data from providers like Uber, Bolt, and local vehicle-sharing services can inform smarter city planning and policy decisions.

This conversation should take about 50-60 minutes, and with your permission, it will be recorded. I understand the sensitivity of information, especially when it comes to unpublished data or internal insights. Please let me know your preference for confidentiality regarding our discussion. Would you be comfortable with your organization or role being mentioned, or do you prefer a higher degree of anonymity? You will also have the opportunity to review a transcript of this interview to ensure accuracy and comfort with the shared information.

Before we begin, do you have any questions?

Introduction

1. "Given your background and experience, what questions should local governments be asking given the growth of New Mobility Service providers in Dutch cities?"

Data Collection and Sharing

- 2. "In your view, how should local governments go about the fact that NMS providers collect a lot of data that can be valuable for policy making?"
- 3. "In your view, what is the state of play in terms of data access and data sharing initiatives between government and NMS providers in Dutch cities?"
 - a. "Do you know of any concrete examples?"

Data Demand and Utilization

- 4. "From your point of view, what is the added value of evidence-based decision-making within Dutch municipalities?"
 - a. "What type of data is much needed and how should/can this data be shared and accessed?"
- 5. "Reflecting on your experience, to what extent are local governments making good use of the data available from NMS providers?"



a. "How could this be improved?"

Public Good vs Private Interests

6. "What are the overarching challenges in balancing public good and private interests in data sharing for urban development?

Challenges

- 7. "What are key challenges you see in data sharing and evidence-based decision making?"
- 8. "From your experience, can you discuss any (potential) gaps between theoretical and practical implementations of (NMS) data usage/?"

Data Collaboration Stimulation and Innovation

- 9. "How could data sharing and collaboration between NMS and municipalities be improved?"
 - a. "Can you discuss any models or frameworks that (could) facilitate effective data sharing and collaboration among these stakeholders?"
- 10. "How can data-driven approaches within municipalities be optimized to enhance urban development?"

Future Perspectives

- 11. "Looking ahead, can you envision any innovative approaches or potential breakthroughs that could significantly impact the future of data-driven urban development?
 - a. How might these innovations reshape the interaction between municipalities and data providers?"

Ending/Conclusion

- 12. "Is there anything else you believe is crucial for understanding the supply, demand, and utilization of trip data in evidence-based decision-making for developing smart and sustainable cities?"
- 13. "Do you have any questions or comments that you would like to add?"

This concludes the interview. Once again, thank you very much for your time and insights. Your contribution is incredibly important to my research. I'll ensure the information you've shared is used responsibly and with respect to your anonymity. If you have any further questions or wish to discuss any aspect of the study further, please don't hesitate to contact me.



20. APPENDIX J – (INTERVIEW GUIDE INDUSTRY REPRESENTATIVES)

Thank you very much for agreeing to participate in this interview. My name is Floris van den Dool, and I am conducting research for my master's thesis for the University of Utrecht's Sustainable Business and Innovation program. This research delves into public-private collaboration in urban planning and policymaking, with a specific focus on the role of Mobility as a Service (MaaS)/New Mobility Service (NMS) data. This includes companies and services that are reshaping urban mobility, such as ride-sharing platforms, electric scooter rentals, and ride-hailing platforms. I'm interested in how their data on usage patterns, service demand, operational challenges, etc. can support more informed urban planning and policy decisions.

Today, given the transformative potential of MaaS and NMS in shaping our urban environments, we'll discuss the perspectives of NMS companies on data sharing, collaboration, and their impact on urban development.

This conversation should take about 30-40 minutes, and with your permission, it will be recorded. I understand the sensitivity of information, especially when it comes to unpublished data or internal insights. Please let me know your preference for confidentiality regarding our discussion. Would you be comfortable with your organization or role being mentioned, or do you prefer a higher degree of anonymity? You will also have the opportunity to review a transcript of this interview to ensure accuracy and comfort with the shared information.

Before we begin, do you have any questions?

Introduction

1. "Could you briefly describe your work position in NMS, especially concerning data sharing and urban planning?"

Impact on Urban Planning

"From your perspective, how do NMS companies like yours impact Dutch city's mobility? "

 What role does data play in this impact?"

Data Sharing and Utilization in Practice

- 3. "Can you describe any existing collaborations involving data sharing between your company and urban planners or government bodies?
- 4. "What are the key challenges your company faces when sharing data with public parties?"
 - a. "How are these challenges being addressed, and what improvements do you envision could facilitate better collaboration?"
 - b. How does business interest perform against public good?

Data Demand

- 5. "How is data demand from municipalities identified, and how is data shared with these demanding parties?
 - a. Does your company share data proactively, or is it driven by specific requests?"

Collaboration, demand, and Integration



- 6. "What type of data is your company currently sharing or willing to share with governmental bodies, and what considerations play a role in what you might be willing to share?"
- 7. What data are municipalities demanding from your company?
 - a. Does your company share data proactively, or is it driven by specific requests?
- 8. "Could you elaborate on what this data entails?"
 - a. "Has this data been processed or analyzed before sharing?
 - i. If so, how?"

Value of Data Sharing

- 9. "How do you perceive the value of these data for both your company and governmental bodies?"
 - a. "Are there specific benefits or outcomes you aim to achieve through data sharing?"
 - b. Why are specific data types being shared?

Ending

- 10. "Is there anything else you think is particularly important regarding the supply, demand, and/or usage of trip data in evidence-based decision-making for urban development?"
- 11. "Do you have any questions for me, or is there anything you'd like to add that we haven't covered?"

This concludes the interview. Once again, thank you very much for your time and valuable insights. Your contribution is essential to understanding the complexities of data sharing between NMS companies and municipalities in the context of urban planning and development. Please remember that everything discussed will be treated confidentially, and I'm more than happy to share a transcript of our conversation with you. If you have any further thoughts or questions in the future, please feel free to contact me.